POST-CONSTRUCTION BIRD AND BAT FATALITY MONITORING FOR ONSHORE WIND ENERGY FACILITIES IN EMERGING MARKET COUNTRIES

Good Practice Handbook and Decision Support Tool









About IEC

IFC — a member of the World Bank Group — is the largest global development institution focused on the private sector in emerging markets. We work in more than 100 countries, using our capital, expertise, and influence to create markets and opportunities in developing countries. In fiscal year 2022, IFC committed a record \$32.8 billion to private companies and financial institutions in developing countries, leveraging the power of the private sector to end extreme poverty and boost shared prosperity as economies grapple with the impacts of global compounding crises. For more information, visit www.ifc.org.

About EBRD

The EBRD is investing in changing people's lives from central Europe to Central Asia, the Western Balkans and the southern and eastern Mediterranean region. Since 1991 we have invested over €180 billion in more than 6,600 projects. With an emphasis on working with the private sector, we invest in projects, engage in policy dialogue and provide technical advice which fosters innovation and builds modern economies that are competitive, well-governed, green, inclusive, resilient and integrated. The Bank is owned by 71 countries as well as the EU and the EIB. This work has been supported by the EBRD Shareholder Special Fund.

About KfW

KfW is one of the world's leading promotional banks. With its decades of experience, it works on behalf of the federal government and the federal states of Germany to improve economic, social and environmental living conditions at home and abroad. KfW Development Bank is responsible for part of KfW's international business, in which the two Group subsidiaries KfW IPEX-Bank and DEG are also involved: While KfW IPEX-Bank is active in international project and export finance, DEG finances and advises private companies investing in developing countries and emerging market economies.

© 2023 International Finance Corporation. All rights reserved.

2121 Pennsylvania Avenue, NW Washington, DC 20433 USA Internet: www.ifc.org

© European Bank for Reconstruction and Development

5 Bank St, London UK E14 4BG www.ebrd.com

© KfW Group

Palmengartenstrasse 5-9 60325 Frankfurt am Main

This work is a product of the staff of the International Finance Corporation (IFC), Kreditanstalt für Wiederaufbau (KfW) and European Bank for Reconstruction and Development (EBRD).

The material in this work is copyrighted. Copying and/or transmitting portions or all of this work without permission may be a violation of applicable law. All rights reserved. No part of this publication may be reproduced or transmitted in any form or by any means, including photocopying and recording, without the written permission of the copyright holder. Such written permission must also be obtained before any part of this publication is stored in a retrieval system of any nature.

Dissemination of this work is encouraged and we will normally grant permission to reproduce portions of the work promptly, and when the reproduction is for educational and noncommercial purposes, without a fee, subject to such attributions and notices as we may reasonably require.

All other queries on rights and licenses, including subsidiary rights, should be addressed to IFC's Corporate Relations Department, 2121 Pennsylvania Avenue, NW, Washington, DC 20433 USA; EBRD Editorial & Creative Department, 5 Bank St, London UK E14 4BG; KfW Development Bank. Communication. KfW Group. Palmengartenstrasse 5-9. 60325 Frankfurt am Main.

The International Finance Corporation is an international organization established by Articles of Agreement among its member countries, and a member of the World Bank Group. All names, logos, and trademarks are the property of IFC, and you may not use any of such materials for any purpose without the express written consent of IFC. Additionally, "International Finance Corporation" and "IFC" are registered trademarks of IFC and are protected under international law.

Terms and names used in this report to refer to geographical or other territories, political and economic groupings and units, do not constitute and should not be construed as constituting an express or implied position, endorsement, acceptance or expression of opinion by the European Bank for Reconstruction and Development or International Finance Corporation or KfW or its members concerning the status of any country, territory, grouping and unit, or delimitation of its borders, or sovereignty.

Disclaimer

While believed reliable, the IFC, EBRD and KfW does not guarantee the accuracy, reliability, or completeness of the content included in this work, or for the conclusions or judgments described herein, and accepts no responsibility or liability for any omissions or errors (including, without limitation, typographical errors and technical errors) in the content whatsoever or for reliance thereon.

Any views expressed in this technical guidance do not necessarily reflect the views of IFC, EBRD and KfW. Nothing in this technical guidance can be construed as a view on or interpretation of IFC's Sustainability Framework.

All information is provided "as is," with no guarantee of completeness, accuracy, timeliness, or of the results obtained from the use of this information, and without warranty of any kind, express or implied, including, but not limited to, warranties of performance, merchantability, and fitness for a particular purpose.

IFC, EBRD and KfW expressly disclaims any responsibility or liability for damages of any kind, including special, indirect, incidental, consequential, or compensatory damages, arising from or relating to the use of or failure to use or reliance on any information, methods, processes, recommendations, conclusions, or judgments contained in this technical guidance, and IFC,EBRD and KfW makes no representations about the alignment or conformity of this technical guidance with the international, national, or subnational legal requirements of any jurisdiction or any industry standards.

In publishing and making this technical guidance available, IFC, EBRD and KfW is providing a general guide for informational purposes and not for rendering professional or other services for, or on behalf of, any person or entity, and is not agreeing to perform any duty owed by any person or entity to another. Professional advice of qualified and experienced persons should be sought before entering (or refraining from entering) into any project activity or following any guidance set out in this technical guidance. Additionally, this technical guidance is not intended to constitute legal, securities, or investment advice, an opinion regarding the appropriateness of any investment, or a solicitation of any type.

Nothing herein shall constitute or be construed or considered to be a limitation upon or waiver of the privileges and immunities of the IFC all of which are specifically reserved.

Cover photo, top left: European Honey Buzzard (*Pernis apivorus*). Photo: Getty Images Cover photo, bottom right: Natterer's Bat (*Myotis nattereri*). Photo: Getty Images

TABLE OF CONTENTS

Symbols Used in the Handbook	vi
Glossary of Acronyms and Terms	vii
Foreword	xiii
Preface	xiv
Acknowledgments	xv

1. Understanding Post-Construction Fatality Monitoring21.1What is PCFM?21.2Why Focus on PCFM?31.3This Good Practice Handbook and Its Decision Support Tool41.4Who Should Use This Handbook?51.5Applying Relevant Lenders' Environmental and Social Standards5

2. Overview82.1 Outline82.2 Generalized First Year Design102.3 The DST and Compatibility with the Generalized Estimator of Mortality112.4 Consideration of WEF-Associated OHLs11

3. PCFM: Concepts and Principles 14 3.1 Core Concepts 14 **3.2** Cross-Cutting Concepts 16 **3.2.1** PCFM Target Species 16 3.2.2 Monitoring Objectives 17 3.2.3 Sources of Bias in PCFM Study Design 18 3.2.4 Detection Probability 19 3.3 PCFM Components—Turbine Searches 21 3.3.1 Study Duration 21 3.3.2 Turbine Sample 22 **3.3.2.1** How Many Turbines to Sample? 22 3.3.2.2 Which Turbines to Search? 22 3.3.3 Search Plot Shape and Size 23 **3.3.3.1** Why are Plot Shape and Plot Size Important? 23 3.3.3.2 Road and Pad Plots 24

3-3-4	. Transect Width	25
	3.3.4.1 What Determines Transect Width?	25
3-3-5	Scanning Searches: An Alternative Approach to Transect-Based Searches for Large Birds	26
3.3.6	Search Interval	27
	3.3.6.1 What is the Search Interval?	27
	3.3.6.2 Factors Influencing Search Interval Selection and Carcass Size	27
3-3-7	Bias Correction	28
	3-3-7-1 Unsearched and Unsearchable Area	28
	3.3.7.2 Searcher Efficiency	31
	3-3-7-3 Carcass Persistence	34
3.4 PCFM	A Components—WEF-Associated OHL Searches	35
3.4.1	Risks from OHLs	35
3.4.2	Comparing PCFM Designs for Turbines and OHLs	36
3.5 Chan	ice Finds	36
3.5.1	Chance Find Fatalities Found Within Search Plots	38
3.5.2	Chance Find Fatalities Found Outside Search Plots	38

4. PCF	M Des	signs an	d Using the DST	40
4.1	Turbir	ne PCFM I	Designs	40
	4.1.1	Using th	e DST	40
	4.1.2	First Yea	r Design	43
		4.1.2.1	DST Input — First Year Design	43
		4.1.2.2	DST Output — First Year Design	44
	4.1.3	Subsequ	ent Years Design	49
		4.1.3.1	DST Input — Subsequent Years Design	49
		4.1.3.2	DST Output — Subsequent Years Design	52
		4.1.3.3	The Design Explorer	53
4.2	WEF-A	Associate	d OHL Designs—Recommended Practices	54
	4.2.1	OHL San	nple	54
		4.2.1.1	Collector System within a WEF	54
		4.2.1.2	Interconnection Line	55
		4.2.1.3	OHLs Not Associated with WEFs	56
	4.2.2	OHL Sea	rch Plot Shape and Size	56
	4.2.3	OHLTrar	nsect Width	56
	4.2.4	OHL Sea	rch Interval	57
	4.2.5	OHL Bias	s Correction	57
		4.2.5.1	Unsearched and Unsearchable Area	57
		4.2.5.2	Searcher Efficiency Trials	58
		4.2.5.3	Carcass Persistence Trials	58
4.3	, Data (Collectior	n Forms	59

5. Data	a Ana	lysis	62
5.1	Fatali	ty Rate Estimation	62
	5.1.1	Precision	62
	5.1.2	Accuracy	63
	5.1.3	Relationship Between Precision, Accuracy, and Study Design	63
	5.1.4	Fatality Estimation Tools	64
		5.1.4.1 The GenEst Fatality Estimator	64
		5.1.4.2 Evidence of Absence	66
	5.1.5	Reporting of Data Analysis	67
		5.1.5.1 Reporting — PCFM Design Parameters	67
		5.1.5.2 Reporting — Success of PCFM Implementation	67
		5.1.5.3 Reporting — Raw Fatality Data	67
		5.1.5.4 Reporting — Bias Corrected Fatality Estimates Using GenEst	68
5.2	Analy	sis of Spatio-temporal Fatality Patterns	69

6. Adaptive Management	72
6.1 Adaptive Management Review	72
6.1.1 Review Process	72
6.1.2 Review Frequency	75
6.2 Adaptive Management Thresholds and Triggers	76
6.2.1 Determining Fatality Thresholds in an Emerging Market Context	76
6.2.2 Potential Biological Removal	77
6.2.2.1 Minimum Population Size (Nmin)	77
6.2.2.2 Maximum Annual Recruitment Rate (Rmax)	78
6.2.2.3 Recovery Factor (F)	78
6.2.2.4 From PBR Value to Threshold Through Expert Consultation	78
6.2.3 Other Risk Events	79
6.3 Management Actions	81
6.4 Roles, Responsibilities, Reporting, and Disclosure	84
6.4.1 Roles and Responsibilities	84
6.4.2 Reporting and Disclosure	85
6.4.2.1 Internal Reporting	85
6.4.2.2 External Reporting and Informing Strategic Planning	86

References	88
Appendices	96

LIST OF FIGURES

Figure 1.1	PCFM and the adaptive management process	4
Figure 3.1	WEF PCFM design components	15
Figure 3.2	General representation of detection probability as the product of searcher efficiency, carcass persistence probability, sampling fraction, and area correction	19
Figure 3.3	Examples of search plot types (light green polygons): (A) circular plot, (B) square plot, (C) road and pad plot, (D) partial circular plot because of flooding and crop land	23
Figure 3.4	Examples of transect spacing (not to scale): (A) 6 m spacing for a study intended to capture data on bats and small birds and (B) 20 m spacing for a raptor-focused PCFM study	25
Figure 3.5	Schematic examples of typical carcass persistence probability for four size classes of carcasses relative to search interval	28
Figure B3.4.1	Carcass distribution within the fall zone around a turbine overlayed with rings used to calculate DWP	30
Figure 3.6	Schematic examples of carcass density distributions for four carcass size classes and proportion of areas searched on hypothetical 80 m radius full plots or road and pad plots	31
Figure 4.1	DST User Interface tabs	41
Figure 5.1	Relationship between precision and accuracy in fatality rate estimation	63
Figure 5.2	Simple fatality estimation equation	64
Figure 6.1	Summary of the four key stages of the adaptive management process	75

LIST OF TABLES

Table 2.1	Resources to help practitioners design and implement PCFM programs effectively	10
Table 3.1	Comparison of turbine and OHL PCFM design components	37
Table 4.1	DST Project Set-up tab: Questions 1-3	43
Table 4.2	DST First Year tab: Questions 4-6	44
Table 4.3	DST Subsequent Years tab: Questions 1-14	50
Table 4.4	DST Design Explorer: Input table	53
Table 5.1	Common fatality estimators and their key assumptions	65

LIST OF BOXES

Box 3.1	Crippling bias	18
Box 3.2	Detection probability and meeting lender requirements	20
Box 3.3	Fatality searches using dogs	26
Box 3.4	Density Weighted Proportion and the DWP Tool	30
Box 3.5	Obtaining carcasses for bias correction trials: challenges and alternative approaches	32
Box 4.1	Optimizing Recommended Practices for Electrocution Risk	58
Box 5.1	Fatality Estimation for OHLs Using GenEst	65
Box 6.1	Minimizing collision and electrocution risk through on-site mitigation	82

APPENDICES

Appendix A:	PCFM Decision Support Tool	96
Appendix B:	Implementing PCFM Fieldwork	97
Appendix C:	Generalized Estimator of Mortality — Worked Example	119
Appendix D:	Density Weighted Proportion (DWP) Calculation Tool and Searcher Efficiency Trial Placement Tool	145
Appendix E:	PCFM Fatality Record Form	146
Appendix F:	Search Summary Form	148
Appendix G:	Searcher Efficiency (SE) Trial Form	150
Appendix H:	Carcass Persistence (CP) Trial Form	152
Appendix I:	Carcass Freezer Tags	154
Appendix J:	Freezer Contents Log	155
Appendix K:	Essential Fieldwork Equipment	156

SYMBOLS USED IN THE HANDBOOK

GENERAL



GOOD PRACTICE HANDBOOK ADDITIONAL RESOURCES



KEY CONCEPT



OVERHEAD POWER LINE RELATED INFORMATION

PCFM COMPONENTS



STUDY DURATION



TURBINE SAMPLE



SEARCH PLOT SHAPE AND SIZE



TRANSECT WIDTH



SEARCH INTERVAL



UNSEARCHED AND UNSEARCHABLE AREA



SEARCHER EFFICIENCY TRIALS



CARCASS PERSISTENCE TRIALS

GLOSSARY OF ACRONYMS AND TERMS

a	Mathematical symbol for Area Correction
ACCURACY	The degree to which an estimate matches the truth.
AICC	Akaike Information Criterion corrected for finite sample size
AREA CORRECTION	An adjustment to a carcass count to correct for the proportion of carcasses that may have fallen where searchers did not or could not search.
ARTIFICIAL DECOY	A fake version of a bird or bat, for example, plastic or fiberglass hunting pigeon or duck or a rubber toy bat that serves as a surrogate for searcher efficiency trials.
BAYESIAN MODEL	A statistical model that relies on the branch of statistics informed by Bayesian theory. Bayesian models are alternatives to the more familiar frequentist models.
BLADE FEATHERING	A method for slowing or stopping turbine blades to minimize collision risk (mainly used as collision mitigation for bats) achieved by adjusting the angle of the rotor blades so that wind resistance is minimized.
BIAS	With respect to PCFM, the extent to which a fatality estimate differs from true fatality. Carcass counts tend to be underestimates of true fatality, principally because of such things as imperfect searcher efficiency and carcass removal by scavengers and when portions of the carcass "fall zone" are unsearched.
BIAS CORRECTION	Mathematical adjustments to carcass counts that a statistical fatality estimator performs to produce an unbiased estimate of fatality. Some bias correction factors are calculated based on observed or known data. Others are calculated based on especially for that purpose.
BIAS CORRECTION FIELD TRIAL	An experiment in which a single trial carcass or decoy is used to test searcher efficiency, or a trial carcass is used to measure carcass persistence. Results from multiple trials are used to estimate searcher efficiency and carcass persistence at the WEF. These estimates are required to calculate unbiased estimates of fatality using the GenEst fatality estimator.
ВМР	Biodiversity Management Plan
CARCASS DENSITY	Relative frequency of carcasses beneath a turbine or OHL. Density tends to be higher closer to the turbine or nearer the centerline of the OHL than further away.
CARCASS DENSITY DISTRIBUTION	A statistical model that describes the carcass fall zone in terms of relative number of carcasses that occur at different distances and directions from turbines.
CARCASS PERSISTENCE	Average probability that scavengers (or other processes) will not remove a carcass that arrives at the WEF between two searches.
CI	Confidence Interval
CLEARING SEARCH	The first carcass search visit conducted by the search team, the purpose of which is to clear search plots of carcasses that were present before the study began so that a fatality estimate can be obtained for the study period only. Carcasses found during a clearing search will not be included in the fatality estimate.

COMMAND LINE INTERFACE	A way to access computing software (e.g., R statistical computing platform that GenEst uses) by typing computer language commands directly into the software. Compare with graphical user interface.
COLLECTOR SYSTEM	Medium-voltage power lines connecting turbines to the onsite substation which are typically underground cables unless geotechnical constraints or permitting conditions require parts of the system to be above-ground.
CONFIDENCE INTERVAL	A statistically estimated range of values within which one is X% confident that the truth occurs. X is the confidence level; the CI becomes wider as X increases. 90% is a common confidence level to use when reporting fatality estimates.
CRIPPLING BIAS	A process that causes carcass counts to be less than the true number of fatalities when injured animals travel beyond the bounds of a search plot before dying.
CURTAILMENT	Method that prevents turbine blades from rotating at low wind speeds: a principal collision mitigation measure for bats.
DENSITY WEIGHTED PROPORTION	An estimate of the fraction of carcasses that occur within the searched areas beneath turbines.
DESIGN TRADEOFFS	The concept that PCFM design in the context of limited resources requires decisions that balance effort and outcomes. For example, very narrow transect widths will result in high searcher efficiency but require a great deal of labor to execute, which may mean that fewer turbines are searched.
DETECTION PROBABILITY	The probability that a fatality that occurs at the turbine or OHL and is deposited in a search plot will be found by searchers.
DETECTION PROBABILITY INDEX	In the DST, an approximate prediction of the fraction of carcasses at the WEF that searchers will find. Includes detection probability (g) and density weighted proportion and is intended to be used to compare alternative PCFM designs rather than predict absolute detection probabilities.
DEVELOPER	The wind energy company that owns the WEF. It is not intended to reference early-stage project development only but is used generically to refer to the entity (company) ultimately responsible for the wind power project, including during the entirety of operations or any part of it.
DPI	Detection Probability Index
DST	Decision Support Tool
DWP	Density Weighted Proportion
EBRD	European Bank for Reconstruction and Development
EMERGING MARKET COUNTRIES	For the purposes of this Handbook, economies considered to be low income, lower middle income, and upper middle-income, per World Bank country classifications.
ENVIRONMENTAL MANAGER	Also referred to as the environmental and social manager; health, safety, and environmental manager; health safety environmental quality manager or other titles, this individual is responsible for overseeing implementation of project environmental and social management commitments, including for biodiversity.
ΕΟΑ	Evidence of Absence Estimator
ESIA	Environmental and Social Impact Assessment

f	Mathematical symbol for Sampling Fraction
FALL ZONE	The area below the wind turbine or OHL in which a carcass will land.
FATALITY	Death at the individual level.
FIRST YEAR DESIGN	A generalized PCFM design for implementation in the first year(s) of WEF operation that accounts for all size classes of birds and bats and could be implemented in all types of environments.
g	Mathematical symbol for Detection Probability
GENERALIZED ESTIMATOR OF MORTALITY	The statistical models and software recommended in this Handbook, developed specifically to analyze PCFM data.
GENEST	Generalized Estimator of Mortality
GIIP	Good International Industry Practice
GLOBAL SOUTH	A term used to broadly refer to the regions of Latin America, Africa, Oceania, and parts of Asia.
GPS	Global Positioning System
GRAPHICAL USER INTERFACE	A "point and click" way to interact with computer software such as the GenEst fatality estimator. Compare with command line interface.
IFC	International Finance Corporation
IUCN	International Union for Conservation of Nature
k	Mathematical symbol for the factor by which searcher efficiency decreases with each successive search
KFW	The German development bank, Kreditanstalt Für Wiederaufbau
LE	Lead Ecologist
LEAD ECOLOGIST	A regionally experienced, knowledgeable field ecologist with a bachelor's or higher degree in ecology or related science from an accredited college or university with adequate postgraduate ecological experience in the wind-wildlife sector.
LENDERS	Collective term for international finance institutions and other development and commercial financiers.
М	Mathematical symbol for estimated total number of fatalities
MEASURABLE ADVERSE IMPACTS	Negative impacts on biodiversity that can be measured.
MITIGATION HIERARCHY	Actions taken to anticipate and avoid or, when avoidance is not possible, minimize and, when residual impacts remain, offset for risks and impacts on the environment. The mitigation hierarchy is a central tenet to IFC, EBRD, and KfW environmental and social standards.
MORTALITY	Death at the population level.
n	Mathematical symbol for turbine sample size
Ν	Mathematical symbol for total number of turbines
OFF-PLOT DISCOVERIES	Bird and bat carcasses found outside the search plot.

OHL	Overhead power line
P	Mathematical symbol for the probability of detection on the first search — See Searcher Efficiency
PBR	Potential Biological Removal
PCFM	Post-Construction Fatality Monitoring. In this Handbook refers to a systematic search program conducted along walked transects around turbines and along OHLs and associated bias correction trials.
PRECISION	Ability to specify an estimate within a narrow range of values, usually expressed as a CI. For example, an estimate of 4 with a 90% CI of (3, 5) is more precise than an estimate of 4 with a 90% CI of (2, 6).
PRIORITY BIODIVERSITY VALUES	Biodiversity values used to implement a lender's biodiversity standards (e.g., IFC Performance Standard 6). These may include species that are a) listed as Vulnerable, Endangered, or Critically Endangered on the IUCN Red List or that are nationally or regionally listed; b) restricted-range, migratory, or congregatory species; or c) Least Concern on the IUCN Red List but may be especially prone to collision with turbines or WEF-associated OHLs. Priority biodiversity values may also include species of importance to stakeholders for other reasons, such as those that have cultural value, and that might be at risk from cumulative impacts.
PVA	Population Viability Analysis
r	Mathematical symbol for the probability that a carcass arriving during a specified interval persists to the end of the interval — See Carcass Persistence
RAP	Road and Pad Plot
ROAD AND PAD PLOT	Area around a turbine that is maintained free of vegetation during the operational life of a wind farm for the purpose of vehicular access (e.g., turbine pad, crane pad, access roads).
SAC	Stakeholder Advisory Committee
SCADA	Supervisory Control And Data Acquisition. Computer system that controls turbines at a WEF.
SAMPLE	A representative group selected from a larger population. For PCFM, searchers will search a sample of the total number of turbines at the WEF.
SAMPLING FRACTION	The proportion of wind turbines at a WEF or the proportion of segments at a WEF-associated OHL that are searched.
SEARCH AREA CORRECTION	A correction to the counts of fatalities to account for areas under turbines or OHLs that could not be searched but within which carcasses might fall. See also density weighted proportion.
SEARCH INTERVAL	The length of time between searches.
SEARCH PLOT	The area in which carcass searching occurs.
SEARCH TEAM	The team of searchers who will search for carcasses at turbines or WEF-associated OHLs. Also referred to as the 'field team'.
SEARCH VISIT	A single search of all sample turbines.
SEARCHER	The person conducting fatality searches and participating in bias correction trials for a PCFM program.

SEARCHER EFFICIENCY	The probability that a searcher will detect a carcass that is present within the search plot during the search.
SHUTDOWN ON-DEMAND	Bird collision mitigation measure where one or more turbines are temporarily stopped from rotating. Shutdown may be observer or machine initiated in response an increased likelihood of bird collisions at a WEF.
SIZE CLASS	An experimental group used in bias correction field trials. Searcher efficiency and carcass persistence are likely to differ according to size of the fatality. Trials to test these two sources of bias are conducted using different size groups of carcass or decoy; typically, bat, small bird, medium bird, large bird.
SUBSEQUENT YEARS DESIGN	A PCFM design informed by first year(s) PCFM results and optimized to monitor fatalities of PCFM target species.
SURROGATE	In bias correction field trials, a carcass or decoy used as a substitute when there are insufficient WEF fatalities available to use in these trials. For searcher efficiency bias trials artificial decoys may be used.
TRIAL ADMINISTRATOR	In searcher efficiency bias correction field trials, an individual (often the LE) who places trial carcasses or decoys in the search plot before a search day begins, and checks whether they are still present if searchers do not find them.
VISIBILITY CLASS	Categories describing the extent and height of vegetation in search plots. Used to assess how difficult it is for searchers to detect carcasses (e.g., Easy — bare ground 90% or greater, all ground cover sparse and 15 cm or less in height; Very Difficult - little or no bare ground, more than 25% of ground cover over 30 cm in height). Visibility classes are mapped for each search plot and used to answer questions in the DST relating to transect width and inform the DPI.
WBG	World Bank Group
WEF	Wind Energy Facility. Comprising turbines infrastructure and WEF-associated OHLs, including any above-ground sections of the medium-voltage collector system which connects the turbines to the onsite substation, and the high-voltage interconnection line which evacuates power from the onsite substation to the grid.
x	Mathematical symbol for the count of fatalities detected through monitoring



FOREWORD

Scaling up the deployment of renewable energy projects is vital to achieve the 2030 targets of the Paris Agreement on climate change. The recent World Energy Transitions Outlook found that current annual investment in renewable energy technology must more than quadruple if we are to keep global temperature rise to 1.5°C over pre-industrial levels. Yet, if poorly planned, sited or designed, the additional infrastructure needed to meet these targets will have considerable adverse impacts on wildlife, including migratory bird and bat species.

Birds and bats are especially susceptible to adverse impacts from wind farms, with high risk of collision with infrastructure and displacement from favored habitats. Identifying the key areas and migratory pathways for these species, coupled with siting and mitigation measures to avoid such impacts, is essential for preventing negative impacts.

Post-construction fatality monitoring (PCFM) of such infrastructure is a critical component for ensuring that negative impacts to birds and bats are understood and minimized. Data from such monitoring efforts will provide valuable information to further reduce fatalities across species, sites, landscapes, and regions.

As the only global convention focused on the conservation of migratory species and their habitats, the Convention on the Conservation of Migratory Species of Wild Animals (CMS) addresses these and other issues, working with national governments, scientific experts, international organizations and stakeholders.

The CMS Energy Task Force, established in 2015 in accordance with CMS Resolution 11.27 (Rev.COP13), has become one of the world's leading multi-stakeholder initiatives aimed at reconciling energy developments and wildlife. The CMS Energy Task Force provides a platform where governments, multilateral environmental agreements, private companies, investors, academia, and non-governmental organizations work together to promote and develop guidance for the sustainable deployment of renewable energy. *The Post-construction Bird and Bat Fatality Monitoring for OnshoreWind Energy Projects in Emerging Market Countries Good Practice Handbook* and *Decision Support Tool* is a result of such cooperation.

This Handbook is founded on robust science and provides a user-friendly Decision Support Tool, which will inform adaptive management measures throughout the life of a project. I would like to express our gratitude to the International Finance Corporation, European Bank for Reconstruction and Development, *Kreditanstalt für Wiederaufbau* who have undertaken the development of this valuable work, and I am keen to see it adopted by all relevant actors.



Any Frankel

Amy Fraenkel Executive Secretary Convention on the Conservation of Migratory Species of Wild Animals (CMS)



PREFACE

A transition to clean energy is at the heart of the climate change mitigation objectives of the Paris Agreement. Energy consumption is a key driver of global greenhouse gas emissions, contributing three-quarters of worldwide emissions. Global action is urgently needed to decarbonize our energy supply. To do so, expanding the role of renewable energy globally will be essential for shifting to a low-carbon economy and ensuring access to sustainable, affordable clean energy for millions of people who are currently unconnected.

Wind energy has a role to play in filling the energy access gap. It is the second-fastest-growing renewable energy source globally, after hydropower, with an annual growth of over 10 percent in 2021, yet it is still a vastly underused resource. If we harness an increased share of that potential, wind energy will help achieve affordable, reliable, sustainable, and modern energy for all (SDG7) needed to tackle poverty reduction and climate change.

Against the backdrop of such enormous resource potential, the global wind industry is on the cusp of unparalleled expansion, and it is imperative that biodiversity be protected while such growth is achieved. Onshore wind energy facilities (WEFs) have potentially significant impacts on wildlife, particularly birds and bats. In the face of an unprecedented rate of global biodiversity loss and to further the commitment of multilateral development banks to support investments that are compatible with nature, we look to approaches and tools that will help ensure that the shift to a decarbonized future integrates biodiversity considerations. The only way to understand the collision impacts of WEFs on birds and bats is to design and implement post-construction fatality monitoring (PCFM).

Three development finance institutions — International Finance Corporation (IFC), European Bank for Reconstruction and Development (EBRD), *Kreditanstalt für Wiederaufbau* (KfW) — have developed this Handbook and its Decision Support Tool to promote a consistent, systematic approach to designing and implementing PCFM programs. The Handbook enables the private sector, governments, conservation organizations, and other experts in emerging market countries to better understand the actual impacts of wind energy on wildlife and to take concrete steps to mitigate them. We hope it will set a new global benchmark for biodiversity data collection and impact mitigation for the sector.



Henrik Linders Managing Director, Environment and Sustainability Department European Bank for Reconstruction and Development (EBRD)



Andrea Kopf

Director, Non Financial Risks and Sustainability Kreditanstalt für Wiederaufbau (KfW)



Tania Kaddeche Director, ESG Sustainability, Advice and Solutions International Finance Corporate (IFC)

ACKNOWLEDGMENTS

The PCFM Handbook and Decision Support Tool have been developed by an international team of specialists guided by the experiences and advice from ecologists and wind wildlife practitioners from many parts of the world. This expertise has been essential and invaluable in realizing our core aim of providing a resource that develops and consolidates PCFM good practice for use in emerging market countries and elsewhere.

This initiative was led by Lori Anna Conzo (Global Biodiversity Lead, IFC), Daniel Skambracks (Senior Environmental Adviser, KfW) and Robert Adamczyk (Sector Lead Power and Heavy Industry, EBRD) who contracted the following consultancies to co-develop this work:



Western EcoSystems Technology (WEST)

A US-based consulting company that provides environmental and statistical consulting services and contract research nationally and internationally for industry, government, and private organizations.



Natural Power

An international consultancy and service provider working to create a world powered by renewable energy.

The core technical and drafting team consisted of Lori Anna Conzo and Simon Hulka from IFC, Paul Rabie and Kate MacEwan from WEST, and David Tidhar and Zoe Howell from Natural Power. In collaboration with other members of the team, Paul Rabie was the main developer of the Decision Support Tool and helped ensure its compatibility with the Generalized Estimator tool, and Simon Hulka was the main developer of the Density Weighted Proportion tool. IFC, KfW and EBRD are extremely grateful to WEST and Natural Power for their trusted technical guidance, their patience and tenacity, which have been a tremendous asset throughout the evolution of this work.

We are grateful for the guidance provided by our Stakeholder Advisory Committee, which was created at the inception of this work, and consisted of Tris Allison (BirdLife International), Jon Aronson (Camissa Sustainability Consulting), Alvaro Camiña Cardenal (ACRENASL consulting) and Cris Hein (National Renewable Energy Laboratory). In addition to the Stakeholder Advisory Committee, key contributions were provided by our Expert Review Committee, which was selected to incorporate global lessons learned on the topic of wind wildlife science and Post-construction Fatality Monitoring. They include Pablo Brandolin (*Universidad Nacional de Río Cuarto -* Argentina), Jan Blew (BioConsult SH Germany), Sherif Baha El Din (Nature Conservation Egypt),

Laith El Moghrabi (Fieldfare Ecology - Jordan), Branko Karapandža (Fauna C&M consultancy; Wildlife Conservation Society Mustela - Serbia), Sujit Narwade (Bombay Natural History Society - India) and Christian Voigt (Leibniz Institute for Zoo and Wildlife Research - Germany).

The lead organizers of the Convention of Migratory Species Energy Task Force also deserve special recognition — Rhiannon Niven (BirdLife International) and Iván Ramírez (Convention of Migratory Species) were instrumental in continuing to promote this Handbook throughout its development. We also thank those members of the Energy Task Force who provided comments on this document during the external review period, and we thank Ashton Berry and Harvey Rich who were formerly leading the Energy Task Force in the initial stages of development of this Handbook and its conception.

We also recognize the technical review and significant contributions by Caleb Gordon, Alessandro Casartelli and Les Hatton from IFC, the notable contribution from Gillian Vallejo from Natural Power on the power lines section of this Handbook and the review and inputs from Jurgen Von Borries and Jennifer Hruza from IFC. Lastly, we thank Rashanikka Hayley Fowler from IFC for her support throughout the development of this work and during the editorial and publication process.

UNDERSTANDING POST-CONSTRUCTION FATALITY MONITORING

1. UNDERSTANDING POST-CONSTRUCTION FATALITY MONITORING

Impacts on wildlife caused by onshore wind energy facilities (WEFs) have been well documented for decades and include collision, habitat modification, displacement, and barrier effects (Allison et al. 2019; Drewitt and Langston 2006; Gove et al. 2013; Katzner et al. 2019; Perrow 2017; Schuster, Bulling, and Köppel 2015). Of these, collisions of birds and bats with turbines is the most universal, resulting in fatalities and requiring the most attention (Arnett et al. 2016; Marques et al. 2014; Thaxter et al. 2017). Overhead power lines (OHLs) associated with WEFs¹ exacerbate risks to birds through collision and electrocution (Jenkins, Smallie, and Diamond 2010; Lehman, Kennedy and Savidge 2017; Martín Martín et al. 2021 Prinsen et al. 2011). OHL collision risk to bats is poorly understood but studies indicate that most bat species are unlikely to regularly collide with these types of structures (e.g., Mogdans, Ostwald, and Schnitzler 1988; Vanderelst, Holderied, and Peremans 2015). The risk of OHL electrocution to larger bat species has been demonstrated (Chouhan and Shrivastava 2019; Rajeshkumar, Raghunathan, and Venkataraman 2013; Tella et al. 2020).

As the wind energy sector expands globally, the potential for site-specific and cumulative impacts on birds and bats will continue to increase. In developed markets, the risks and impacts from WEFs and the efficacy of associated mitigation measures are generally well understood, and regulations for biodiversity protection are often in place. In many emerging market countries,² the industry is typically at an earlier stage of development, and regulatory processes may not require adequate biodiversity safeguards at project sites. In these situations, the potential for adverse impacts on susceptible wildlife populations to go unchecked is significant. This concern is amplified as species data in emerging market countries are typically scarcer than in developed markets, and the extent of threatened biodiversity is comparatively higher.^{3,4}

The impacts associated with WEFs are highly unpredictable, even when robust pre-construction baseline survey results are available (Ferrer et al. 2012; Hein, Gruver, and Arnett 2013; Solick et al. 2020). Although rigorous pre-construction baseline studies are necessary to characterize risks, post-construction fatality monitoring (PCFM) is the only way to understand the actual collision impacts of WEFs on birds and bats. Implementing a robust PCFM program during the operational phase of a WEF is therefore critical for effective management and mitigation of biodiversity impacts.

1.1 What is PCFM?

PCFM generally comprises:

- Searches conducted during the operations phase of a project to search for evidence of bird and bat fatalities at a WEF
- Field trials to estimate the number of fatalities missed during searches either because they were removed before the next carcass search (e.g., by scavenging animals) or because they were overlooked by searchers
- > Analyses to quantify bird and bat fatality rates at the WEF using fatality estimation software

¹ OHLs associated with WEFs are referred to as WEF-associated OHLs in this Handbook.

² For the purposes of this Handbook, "emerging market" countries are those considered to be low-income, lower-middle income, and upper-middle-income economies, according to the World Bank Country Classifications.

³ For example, of the 100 globally threatened bird and bat species predicted to be at the highest risk of collision with wind turbines, 82 and 88 percent, respectively, are in the Global South, based on data in Thaxter et al. (2017).

⁴ See also, Schmeller et al. (2017) which highlights the relative scarcity of data from biodiversity-rich regions compared to North America and Europe, and Frick, Kingston and Flanders (2020), which highlights this disparity for bat populations specifically.

Despite the focus on fieldwork and analysis, PCFM is not meant to be an exercise in data collection for its own sake; it is intended to inform implementation of a Biodiversity Management Plan (BMP) during the operations phase of a WEF, namely the effectiveness of mitigation, as prescribed in the relevant Environmental and Social Impact Assessment (ESIA) or by government consenting or licensing authorities. During the operations phase of a WEF, PCFM data serve as the backbone for adaptive management (see Section 6), decision making, and demonstration of compliance with environmental requirements of governments or financiers (see Section 1.5).

A well-designed PCFM program will document the range of species occurring as fatalities, the scale of mortality, and reveal spatial and temporal fatality patterns all of which will allow the WEF developer⁵ to effectively manage bird and bat collision and electrocution risk.

1.2 Why Focus on PCFM?

A unique aspect of biodiversity management at a WEF is that the focus of monitoring and mitigation is during the *operations phase*, rather than the *construction phase*, and the impacts are primarily in the *air space*. This combination of factors could make the screening of biodiversity risk for this sector difficult for financial institutions with large portfolios. For the wind energy sector, the condition of the terrestrial environment, no matter how modified, is not an appropriate indicator of risk. PCFM results provide the only quantitative basis for measuring the actual impacts of WEFs on birds and bats, and although practical advice is available on avoiding and minimizing impacts following the mitigation hierarchy⁶ (e.g., Bennun et al. 2021; Rodrigues et al. 2015), there remains a need for a globally informed practitioners handbook to guide the process for monitoring and evaluating the scale of those impacts in a credible, robust manner during the operations phase of a project.

PCFM practice varies considerably between regions, especially in emerging market countries, where guidance may be limited or non-existent. Although PCFM is essential to understanding the impacts of operational WEFs, if not conducted following good international industry practice (GIIP), it could lead to impacts being underestimated or overestimated. As wind energy development is expanding rapidly, implications for entire landscapes are considerable. Although wind energy is regarded as green energy, impacts can be significant (e.g., Frick 2017), and without a GIIP-aligned



Black Harrier (Circus maurus). Photo: Chris Van Rooyen

⁵ The term "developer" is used to refer to the wind energy company that owns the WEF. It is not only intended to reference early-stage project development but is used generically to refer to the entity (company) ultimately responsible for the wind power project, including during the entirety of operations or any part of them.

⁶ In alignment with the International Finance Corporation's Performance Standard 1, the mitigation hierarchy is defined as actions taken to anticipate and avoid risks and impacts on the environment or, when avoidance is not possible, minimize and, when residual impacts remain, offset these risks and impacts. The mitigation hierarchy is a central tenet to the environmental and social standards followed by the International Finance Corporation, the European Bank of Reconstruction and Development and *Kreditanstalt für Wiederaufbau*.

PCFM program, the impacts of the sector on bird and bat populations, including at landscape and cumulative scales, cannot be properly understood or effectively managed. Furthermore, once a WEF is operational, impacts may persist throughout its approximately 20-year operational life. Potentially this represents substantive biodiversity and reputational risks for the wind energy developer and its financiers if decisive measures are not taken to understand and curb the impact. Figure 1.1 illustrates the role of PCFM within the adaptive management process





1.3 This Good Practice Handbook and Its Decision Support Tool

This Good Practice Handbook (hereafter referred to as the "Handbook") provides guidance on the design of PCFM following GIIP, underpinned by the scientific method and based on decades of evolution of field and analytical methodologies. The accompanying Decision Support Tool (DST, presented in Appendix A) functions in tandem with the Handbook and is designed to assist in

selecting an appropriate PCFM study design, based on WEF characteristics. The primary objective of the Handbook and the DST is to provide practical guidance for the design and implementation of GIIP-aligned PCFM methodology at WEFs in emerging market countries. A secondary objective is to promote global standardization in methodologies for monitoring bird and bat fatalities at WEFs so that fatality rates can be compared across sites, landscapes, countries, and regions, promoting more uniform, more effective management of site-specific and cumulative impacts. Although this Handbook is principally designed to account for collision risks, the methods were developed also to assess impacts of electrocution on birds and bats at WEF-associated OHLs.

1.4 Who Should Use This Handbook?

Four main user groups are expected to benefit from the Handbook and the DST.

- > Wind energy developers and the ecologists working with them are the primary targeted end users. The Handbook and DST will enable them to characterize the impacts of the WEF, inform the type and scale of mitigation, help safeguard bird and bat populations within and around the WEF, and conform to environmental standards of international financiers.
- International development financiers, commercial financiers, institutional investors, and asset managers are encouraged to promote use of the Handbook and the DST among the wind energy developers in their portfolio to ensure that green energy investments are not having undue or unsustainable impacts on biodiversity.
- > **Governments** are encouraged to use this material as a resource to inform the permitting and licensing of WEFs to regulate the impacts of WEFs on biodiversity in a sustainable, sector-specific manner.
- > Conservation organizations and academic institutions may use this guidance to inform development of databases to track landscape-scale impacts of WEFs on birds and bats and to understand regional trends.

This Handbook uses the terms "reader" and "user" to refer to all of these types of stakeholders. The Handbook also frequently references "practitioner" and "Lead Ecologist (LE),"⁷ which refers to the individual(s) working for the wind energy developer who will put this guidance into practice. LE and other terms are defined in the Glossary of Acronyms and Terms.

1.5 Applying Relevant Lenders' Environmental and Social Standards

This Handbook accompanies the environmental and social standards, guidance documents, and guidelines of the International Finance Corporation (IFC), the European Bank for Reconstruction and Development (EBRD), and the German development bank *Kreditanstalt für Wiederaufbau* (KfW). (These and other development and commercial financiers are collectively referred to as "lenders.") Relevant standards and guidelines include:

 IFC Performance Standard 6 on Biodiversity Conservation and Sustainable Management of Living Natural Resources and its accompanying Guidance Note 6 (IFC 2012, 2019)

⁷ For further information about this role, see Appendix B.

- EBRD Performance Requirement 6 on Biodiversity Conservation and Sustainable Management of Living Natural Resources and its accompanying Guidance Note (EBRD 2019, 2022)
- > As applicable to Kreditanstalt für Wiederaufbau's public sector lending, the World Bank's Environmental and Social Standard 6 on Biodiversity Conservation and Sustainable Management of Living Natural Resources and its accompanying Guidance Note (World Bank 2017, 2018)
- > The World Bank Group Environmental, Health, and Safety Guidelines for Wind Energy (WBG 2015)
- > The World Bank Group Environmental, Health, and Safety Guidelines on Electricity Transmission and Distribution (WBG 2007)

PCFM results are relevant to the application of; a) Natural Habitat requirements of IFC Performance Standard 6 and World Bank Environmental and Social Standard 6 on *Biodiversity Conservation and Sustainable Management of Living Natural Resources*; b) Priority Biodiversity Features requirements of EBRD's Performance Requirement 6; and c) Critical Habitat requirements of all three⁸. They are particularly applicable to the no net loss^{9,10,11} principle relevant to Natural Habitats and Priority Biodiversity Features requirements and the no measurable adverse impacts^{12,13} goal associated with Critical Habitat requirements. PCFM, together with an operations-phase Adaptive Management Plan, represent a specific management tool to respond to these lender standards when applied to WEFs.



Pipistrelle Bat (Pipistrellus pipistrellus). Photo: Getty Images

⁸ It should be noted that many bird and bat populations use the air space above degraded or converted habitats. In these instances, no net loss and no measurable impacts requirements may be applied to the species using the air space itself rather than the habitat below.

[•] As defined in footnote 9 of IFC's PS6 "No net loss is defined as the point at which project-related impacts on biodiversity are balanced by measures taken to avoid and minimize the project's impacts, to undertake on-site restoration and finally to offset significant residual impacts, if any, on an appropriate geographic scale (e.g., local, landscape-level, national, regional)."

¹⁰ As defined in footnote 8 of the World Bank's ESS6 "No net loss is defined as the point at which project-related biodiversity losses are balanced by gains resulting from measures taken to avoid and minimize these impacts, to undertake on-site restoration and finally to offset significant residual impacts, if any, on an appropriate geographic scale."

¹¹ As defined in footnote 77 of EBRD's PRG "No net loss is defined as the point at which project-related biodiversity losses are balanced by gains resulting from measures taken to avoid and minimize these impacts, to undertake on-site restoration and finally to offset significant residual impacts, if any, on an appropriate geographic scale."

¹² As defined in paragraph GN86 of IFC's Guidance Note 6 "...[no measurable adverse impacts] means that project-related direct and indirect impacts will not jeopardize the long-term persistence of the biodiversity value(s) for which the critical habitat was designated, considering the range of mitigation measures implemented by the client throughout the life of the project and in alignment with the mitigation hierarchy." It should be noted that this is referred to as "measurable net reduction" in the WB's ESS6 [see paragraph 24(c)].

¹³ As defined in footnote 79 of EBRD's PRG^{*}Measurable adverse impacts mean the project's direct and indirect impacts will jeopardize the persistence within the study area of any biodiversity value that triggers a critical habitat designation."

OVERVIEW

2. OVERVIEW

This section provides a brief overview of the Handbook and an introduction to the DST.

2. Overview

2.1 Outline

This Handbook is developed for those who are new to the topic of PCFM, as well as for more experienced practitioners. The content is organized into the following six sections.



- 🔷 Defines PCFM and its importance for understanding impacts at a WEF
- O Introduces the Handbook and DST
- 🔷 Explains who is to benefit from the Handbook and DST
- O Explains the relevance of PCFM to Lenders' environment and social standards

OVERVIEW

- O Provides an overview of the key content in each of the sections of the Handbook
- O Provides details of the resources that are provided to help users implement PCFM
- O Introduces the Generalized "First Year" Design
- Outlines the compatability of the DST with the GenEst fatality estimation software
- O Introduces PCFM guidelines for WEF-related OHLs

PCFM: CONCEPTS AND PRINCIPLES

- Outlines the "must know" core concepts and underlying principles needed to develop a PCFM design
- Outlines core concepts and underlying principles needed to develop a PCFM design
- O Explains how monitoring objectives may influence PCFM design trade-offs
- O Focuses on PCFM for turbines and highlights how concepts apply to WEF-associated OHLs

8

PCFM DESIGNS AND USING THE DST

O Provides guidance on developing PCFM designs for turbines and WEF-associated OHLs

O Provides guidance on using the DST to develop a generalized First Year design

O Provides guidance on using the DST to develop a "Subsequent Years" PCFM design

O Provides guidance on developing WEF-associated OHL PCFM designs

DATA ANALYSIS

O Focuses on the analysis of PCFM data

Provides a brief overview of available fatality rate estimation tools

O Recommends approaches for reporting PCFM data

ADAPTIVE MANAGEMENT

Describes how PCFM results are used to adaptively manage impacts

- Provides insights for developing an adaptive management plan
- Addresses adaptive management thresholds
- Outlines roles and responsibilities of an adaptive management program
- O Provides an outline for reporting and disclosure

To support the Handbook, several resources are provided to help users design and implement PCFM programs effectively and efficiently. These are found in the appendices at the end of the document and summarized in Table 2.1.

Table 2.1 Resources to help practitioners design and implement PCFM programs effectively



2.2 Generalized First Year Design

Turbine collision susceptibility is unknown for many species of birds and bats in emerging market countries, and impacts are often site-specific and not always accurately predicted from preconstruction baseline studies. For this reason, this Handbook and the DST provide a generalized PCFM design for implementation in the first year(s) of WEF operation that accounts for all size classes of birds and bats and could be implemented in any environment (see Section 4.1.2). Although PCFM monitoring objectives may have been preliminarily established in the ESIA or as part of national permitting and licensing conditions, implementing the First Year PCFM design will enable the developer to determine accurate fatality estimates for a broad range of species size classes so that PCFM target species (see Section 3.2.1) can be identified, and an optimized PCFM program can be designed and implemented for subsequent years. Identifying an accurate set of PCFM target species is key to informing the WEF's monitoring objectives, which may change once fatality results become known (see Section 3.2.1 for further information on this topic). Although the study is referred to as the First Year design, it may need to be implemented over multiple years if the first year of monitoring does not accurately identify a set of PCFM target species (see Section 4.1.3). For more information on study duration, which is a separate concept, see Section 3.3.1.

2.3 The DST and Compatibility with the Generalized Estimator of Mortality

The DST (see Appendix A) is a spreadsheet-based support tool accompanying this Handbook that enables the user to begin designing a PCFM program. As described in detail in Section 4.1.1 of this Handbook, the DST asks the user a series of site-specific questions relevant to monitoring and estimating fatalities at the WEF (and based on the answers, outputs a recommended PCFM design). The DST is equipped with separate tabs (worksheets) for determining the First Year design (see Section 4.1.2) and the 'Subsequent Years' design, which is an optimized design focused on PCFM target species derived from the output of the first year(s) results (see Section 4.1.3).

The DST and this Handbook facilitate collection of PCFM data that are compatible with the Generalized Estimator of Mortality (GenEst) modeling tool¹⁴ (Dalthorp et al. 2018), which is currently the best method for estimating bird and bat mortality at WEFs (Rabie et al. 2021). GenEst is a free, publicly available, user-friendly software that enables PCFM results from various projects and landscapes to be standardized (see Section 5.1.4.1 and Appendix C for further information).

The DST and GenEst are complementary and together support the wider objective of the Handbook; to promote a standard global PCFM method. A key benefit of improving standardization between WEFs is that it creates the opportunity to develop regional PCFM databases that will allow a better understanding of the cumulative effects that a rapidly expanding global wind energy industry may have on bird and bat populations.

2.4 Consideration of WEF-Associated OHLs

Although this Handbook focuses on PCFM at turbines, WEF-associated OHLs are also covered. This includes any above-ground sections of the medium-voltage collector system that connect the turbines to the onsite substation¹⁵ and the high-voltage interconnection line that transfers power from the onsite substation to the grid. Practices recommended in this Handbook may be applicable to other OHLs, regardless of their association with wind energy generation, if implemented through the guidance of a qualified expert and with respect to any existing countryor region-specific guidelines.

¹⁴ See https://www.usgs.gov/centers/forest-and-rangeland-ecosystem-science-center/science/a-generalized-estimator-estimating

¹⁵ Although the collector system is usually buried (see also Section 4.2.1.1 of this Handbook), it is included in the definition of OHLs because there may be some exposed overhead lines due to geotechnical constraints or permitting conditions. The collector system is sometimes referred to as "cabling."

As mentioned in Section 1, fatalities at WEF-associated OHLs can account for a substantial proportion of total fatalities at a project (Brenninkmeijer and Klop 2017) and, unlike turbines, OHLs may present collision and electrocution risks to birds and bats. An overview of these risks is provided in Section 3.4.1 of this Handbook, and recommended PCFM practices are introduced in Section 3.4 and presented in 4.2. There is no DST for WEF-associated OHLs, because the key components necessary to run the logic in the tool (e.g., carcass fall zone at OHLs) have not been sufficiently researched. Nonetheless, the guidance in this Handbook provides a reasonable approach for designing a PCFM study.



White Storks (Ciconia ciconia). Egypt. Photo: Emad Abd Allah

PCFM: CONCEPTS AND PRINCIPLES

3. PCFM: CONCEPTS AND PRINCIPLES

This section provides the conceptual background and building blocks for designing a PCFM study. The focus is on turbine searches, with differences in the design for WEF-associated OHLs presented in Section 3.4. One of the aims of this section is to explain how decision making throughout the design process may affect the precision¹⁶ of PCFM results.

3. PCFM: Concepts and Principles

3.1 CORE CONCEPTS

As highlighted in Section 1 of this Handbook, PCFM principally involves:

- > Designing, planning, and conducting searches for bird and bat fatalities
- > Conducting bias correction field trials
- > Calculating fatality rate estimates from the data collected during the previous two activities

The first two activities are covered in this section, and fatality rate estimation is described in Section 5.

Designing and planning a PCFM search program will require the practitioner to be familiar with the following design components:



STUDY DURATION is the number of years that PCFM should be conducted.



TURBINE SAMPLE is the number or proportion of turbines searched for bird and bat fatalities.



SEARCH PLOT SHAPE AND SIZE is the shape and size of the search area at each individual turbine. The search plot may be circular or square and centered on the turbine, or it may consist only of the gravel pad around a turbine, adjacent crane pad, and the road(s) leading up to it.



TRANSECT WIDTH is the spacing between transects walked by searchers within the search plots at each turbine when searching for fatalities.



SEARCH INTERVAL is the time between consecutive fatality searches (sometimes referred to as "search frequency").

¹⁶ Precision refers to the repeatability of a statistical estimate and is usually indicated with a confidence interval (CI). See Section 5 of this Handbook.

Additionally, to derive unbiased fatality rate estimates, the results from a PCFM search program must be corrected to account for the main biases that arise when conducting PCFM fieldwork. To do this the practitioner will need to understand the following bias correction design components:



UNSEARCHED AND UNSEARCHABLE AREA refers to the areas within the fatality fall zone at a sample turbine that are unsearched for some reason, such as being beyond the search plot boundary, or within the search plot but unsafe to search (e.g., steep terrain) or having such complex ground cover that detection would be nearly impossible (e.g., forested areas).



SEARCHER EFFICIENCY TRIALS are field trials used to measure the probability that a searcher will detect a carcass on the ground at the time of a search.



CARCASS PERSISTENCE TRIALS are field trials used to measure the typical amount of time bird and bat carcasses persist on the ground before being removed by scavengers or becoming undetectable because of environmental factors (e.g., flooding, decomposition).

Figure 3.1 illustrates the PCFM design components for fatality searches and bias correction.



Figure 3.1 WEF PCFM design components

3.2 CROSS-CUTTING CONCEPTS

Four main cross-cutting concepts underpin all aspects of PCFM design:

- > PCFM target species
- > Monitoring objectives
- > Sources of bias
- > Detection probability

An understanding of these cross-cutting concepts will help the LE to evaluate the PCFM components (see Section 3.1) and build a robust, appropriate design.

3.2.1 PCFM Target Species

Lender biodiversity standards are often implemented with respect to a defined set of priority biodiversity values.^{17,18} Depending on the context, these priority values may be determined as part of an ESIA, BMP, or other biodiversity-related assessment (e.g., Critical Habitat Assessment, Biodiversity Monitoring and Evaluation Plan, Adaptive Management Plan). At a WEF, priority biodiversity values may include:

- > Species listed as Vulnerable, Endangered or Critically Endangered on the International Union for Conservation of Nature (IUCN) Red List¹⁹ or are nationally or regionally listed
- > Restricted-range, migratory, or congregatory species
- > Non-threatened species on the IUCN Red List but especially prone to collision with turbines²⁰

Priority biodiversity values may also include species of importance to stakeholders for other reasons, such as those that have cultural value, and species that might be at risk from cumulative impacts.

This Handbook uses the term *PCFM target species* which are bird or bat priority biodiversity values that occur in the WEF area of influence and that are being impacted, or have the potential to be impacted, through collision with turbines or collision or electrocution at WEF-associated OHLs. It is likely that PCFM target species will be a subset of the priority biodiversity values.

Although PCFM target species may be preliminarily defined in the ESIA or BMP, not all will be known before implementation of PCFM. Accurately identifying PCFM target species is key to defining the monitoring objectives for a WEF. It is also essential to *optimizing* the PCFM design over the study period and as part of adaptive management (see Section 6).



Cape Vulture (Gyps coprotheres). South Africa. Photo: Chris Van Rooyen

¹⁷ For example, see paragraphs GN12 and GN13 of IFC's Guidance Note 6.

¹⁸ This concept is related to but separate from the concept of critical habitat species, which are those of highest biodiversity value. See Performance Standard 6 (IFC 2012), Environmental and Social Standard 6 (World Bank 2017), and Performance Requirement 6 (EBRD 2019) and related guidance notes.

¹⁹ See https://www.iucnredlist.org/.

²⁰ For more information on collision susceptible bats see Arnett et al. 2016 and for collision susceptible birds see Thaxter et al. 2017

3.2.2 Monitoring Objectives

The main objective of any PCFM study is to characterize wildlife collision impacts to inform potential mitigation and adaptive management during the operational phase of the project. Other objectives may include:

- Determining the number of fatalities with respect to defined fatality thresholds²¹ in the project's Biodiversity or Adaptive Management Plan or with respect to national or regional thresholds
- > Determining temporal and spatial patterns of impact to inform and optimize mitigation strategies
- > Meeting project-specific lender or regulatory policy requirements or achieving other management or scientific goals
- Contributing to an understanding of the cumulative or landscape-scale effects on PCFM target species and informing wind energy sector spatial planning

This Handbook assumes that the objective of characterizing wildlife collision impacts will initially be achieved by implementing and reviewing the results of the First Year PCFM design (see Section 4.1.2). However, some PCFM target species may not appear as fatalities in PCFM monitoring in the first year(s) (e.g., because of their rarity in the landscape, inter-annual variation) and yet might remain a focus of the project's monitoring objectives. Determining which species to retain as PCFM target species, and therefore as monitoring objectives, will consider both stakeholder values and the efficient use of resources. The monitoring objectives should be revisited after implementing the First Year PCFM design and routinely throughout the life of the project (see Section 6). PCFM study duration is covered in Section 3.3.1.



Rüppell's Vultures (Gyps rueppelli). Ethiopia. Photo: Alvaro Camiña

²¹ See Section 6.2 for further information on thresholds.

3.2.3 Sources of Bias in PCFM Study Design

Generating scientifically robust estimates of fatality rates means accounting for and addressing sources of bias that are inherent to carcass counts from PCFM searches. The term "bias" with respect to PCFM means that counts of carcasses are lower than the true number of fatalities. Bias in PCFM studies is generated in three principal ways:

- > Searchers do not find all carcasses (searcher efficiency bias).
- > Not all carcasses persist long enough to be found (carcass persistence bias).
- > Not all carcasses fall within searched areas (unsearched and unsearchable area bias).

Understanding the principles that underpin each PCFM component (Section 3.3) will help minimize these sources of bias when designing a PCFM program.

Bias may also be generated as a result of "crippling bias". See Box 3.1 for more information.

Box 3.1 Crippling bias

Crippling bias occurs when birds and bats are not detected during searches because, despite colliding or being electrocuted, they are able to move away from the search plot before dying (e.g., Murphy et al. 2016; Rioux, Savard and Gerick 2013; Travers 2021). As with the other types of bias described in this Handbook, crippling bias results in fatality estimates that are below the true number of fatalities. This bias is however impractical to quantify at the project scale, because it requires real-time monitoring of non-fatal collisions and electrocutions, which are likely to be rare and unpredictable. There is a gap in research on the extent to which crippling bias occurs as a result of collisions with turbines while for OHLs the option of using values from dedicated crippling bias research is not considered reliable because of the large variance in reported values (APLIC 2012; Birdlife International 2015). Recognizing these problems, approaches for assessing crippling bias are not included in the Handbook or the DST. The most appropriate approach when reporting fatality estimates is to highlight that crippling bias is not accounted for and that estimated mortality may therefore be lower than actual mortality. When there is evidence that injured birds or bats have moved outside the search plot, increasing the search plot distance from the turbine or OHL is a viable option for reducing the effects of this type of bias.
3.2.4 Detection Probability

Imperfect ability to detect carcasses can be accounted for through the detection probability, which is defined as the probability or likelihood that a fatality that is caused by the WEF will be found by searchers. Detection probability is also a convenient measure to use when comparing PCFM designs.

In a perfect world, in which searchers detected 100 percent of fatalities caused by a WEF, the detection probability would be 1, and no bias corrections would be necessary. In the real world, detection probability is always less than 1 due to the combined effects of factors that can prevent searchers from detecting fatalities. However, the primary benefit of the bias correction methods discussed in this Handbook is that, as long as biases can be characterized and the detection probability can be estimated, it is possible to estimate the "true" or unbiased bird and bat fatality rates generated by a WEF.

As shown in Figure 3.2 detection probability incorporates:

- > The effectiveness of searchers in detecting carcasses [searcher efficiency (p)]
- > The probability a carcass is not removed by scavengers or other factors [carcass persistence (r)]
- > The proportion of turbines sampled [sampling fraction (f=n/N)]
- > The proportion of fatalities that occur within searched areas [area correction (DWP)]

The result will be a number between 0 and 1. For example, a PCFM design with a detection probability of 0.1 means that on average, 10% of carcasses that are killed by WEF infrastructure will be found by searchers, whereas a design with a probability of 0.45 means that searchers will find an average of 45% of WEF-caused carcasses.





Box 3.2 Detection probability and meeting lender requirements

At each WEF, geographical, environmental and ecological conditions will differ. These differences will result in varying levels of bias and affect the detection probabilities within PCFM programs. For example, a WEF in a sandy desert will generally have a higher searcher efficiency, less unsearched area, and possibly lower scavenger pressure, resulting in higher detection probability, compared with a WEF in a tropical forest, where these biases are likely to be higher. These differences between WEFs and the limitations that they impose mean that it is not feasible to set a single detection probability that each project should attain to meet lender standards.

In general, lenders will expect more monitoring effort at WEFs where there are PCFM target species with higher risk profiles, and greater monitoring effort should lead to higher detection probability. For example, a higher risk profile may include collision susceptible species that are a) listed as Vulnerable, Endangered, or Critically Endangered on the IUCN Red List; b) restricted-range, migratory, or congregatory species; or c) non-threatened species on the IUCN Red List but for which there are demonstrated high numbers of fatalities, also considering cumulative impacts. In these cases, the developer should expect to expend more monitoring effort (e.g., sampling more turbines, decreasing search interval) to increase detection probability. This determination should be made on a project-by-project basis in consultation with the developers and financiers.

To develop a PCFM design that is likely to meet lender requirements, the developer should have a clear understanding of the PCFM components and principles outlined in this Handbook and use the DST to develop an initial recommended PCFM design and, where applicable,^a use the Design Explorer tool in the DST (see Section 4.1.3.3) to test alternative designs. The developer should also use the detection probability index results in the DST to evaluate and compare the recommended and alternative design and consult with lenders to agree on the PCFM design that will best meet lender requirements.

a The DST outputs a generalized First Year design. The Design Explorer tool is developed to be used when developing a Subsequent Years PCFM design (see Section 4.1.3)



Differing ground cover search conditions. Left: Algarve region, Portugal. Right: Tafila region, Jordan. Photos: Alvaro Camiña (left), Simon Hulka (right)

3.3 PCFM COMPONENTS—TURBINE SEARCHES

In this section, each PCFM component for turbine searches is described in detail, and the key underlying principles supporting design decisions are presented and highlighted in blue boxes. By the end of this section, the reader should understand:

- > The individual contribution of each component to detection probability
- > The basic principles important to making decisions about each component
- > How the components interact with each other

Each component is connected to activities in the field. Appendix B provides guidance on how to collect high-quality field data, which will directly influence the accuracy and precision of fatality estimates.

3.3.1 Study Duration

Key Concept

Longer study duration increases confidence that the species impacted by the WEF are documented and reduces the chances of missing rare events such as collision fatalities of scarce species with high biodiversity value.

With respect to study duration (number of years of monitoring), a minimum of three years after commercial operation of the WEF has begun, covering all relevant seasons^{22, 23, 24} is recommended. Study duration should be extended when there are PCFM target species with higher risk profiles or the effectiveness of adaptive management warrants longer-term monitoring. Longer studies have advantages, including establishing potential spatial and temporal trends in fatality data, and certain PCFM target species may only be detected after multiple years of PCFM. Ultimately, the study duration should be determined by the level of risk and impact to PCFM target species.



²² Seasons are defined based on climate (e.g., summer, autumn, winter, spring; rainy (monsoon), dry) and according to ecological and/or biological factors (e.g., migration periods, breeding and non-breeding seasons of the PCFM target species).

²³ At some WEFs, turbines may begin operating before reinstatement of the ground surface around turbines is completed. During this period, the WEF may implement a protocol that requires fatalities and injured animals to be documented as "chance finds," following the guidance in Section 3.5.

²⁴ For very large WEFs, commercial operations may be staged, with groups of turbines scheduled to become operational sequentially over an extended period. In these cases, PCFM should start at each turbine group when it becomes operational and not be delayed until the WEF is fully operational.

3.3.2 Turbine Sample



Key Concept



Sampling more turbines leads to higher detection probabilities and more precise fatality estimates.

How many turbines are searched (what proportion of the WEF is searched) is a basic and important decision for any PCFM study design. To maximize detection probability, the preference is to search all turbines at a WEF. If that is not feasible, a representative sample should be selected. The DST will recommend the number of turbines, although the decision will ultimately depend on the risk to PCFM target species.

When increased effort is necessary to maximize detection probability during a particular season, it may be appropriate to search more turbines during that season. The number of turbines searched should be the same throughout a season, unless, for example, one or more of the turbine search plots become unsearchable (e.g., the land around the turbine becomes flooded, maintenance activities prohibit searching). The turbines searched should only be changed at the beginning of a new season. For example, if constraints on a PCFM design mean that only a fraction of turbines can be searched for very large WEFs, then an option could be to search a different sample of turbines in the next season. Although searching a different set of turbines in each season will not increase the sampling fraction, it will reduce the possibility that fatality hotspots are missed.²⁵

3-3-2-2 Which Turbines to Search?

Key Concept

To obtain unbiased fatality rate estimates, the selection of sample turbines should be random.

When only a sample of turbines is selected for a PCFM design (i.e., fewer than 100% of turbines are searched), they should be selected *randomly* so that unbiased fatality rate estimates for the WEF can be obtained. Using non-random factors to quide selection of search turbines, such as characterizing

²⁵ Altering the sample completely can also lead to questions about whether differences in seasonal or annual fatality patterns may be related to the turbines sampled. Some practitioners prefer to maintain a consistent portion of the turbine sample over seasons and years. For example, of 40 turbines included in the first PCFM season, 10 are maintained and 30 new turbines are selected for the second PCFM season.

certain spatial fatality patterns or experimental aims (e.g., characterizing the effectiveness of operational phase mitigation), requires advanced statistical techniques to produce unbiased fatality rate estimates and are outside the scope of this Handbook.

3.3.3 Search Plot Shape and Size

3.3.3.1 Why are Plot Shape and Plot Size Important?

Key Concept

Bigger plots contain more of the carcasses that fall below turbines, resulting in higher detection probability and more precise fatality estimates, but they require a greater level of effort per search.

Plots can vary in shape but are typically circular or square²⁶ with the base of the turbine tower located in the center (see Figure 3.3). Circular plot size is described in terms of its maximum radius, and square plot size by the length of the sides. Optimum selection of search plot size is generally defined in terms of the search plot radius in relation to the maximum installed blade tip height of the turbines, which is equivalent to the hub height plus the rotor radius. This is because bigger turbines will generally cause bird and bat carcasses to fall over wider, more dispersed areas (Choi, Wittig, and Kluever 2020). Regardless of size or shape, plots will require an area correction to account for carcasses that may fall beyond the edge of the plot. For more information on accounting for the unsearched and unsearchable area, see Section 3.3.7.1.

Figure 3.3 Examples of search plot types (light green polygons): (A) circular plot, (B) square plot, (C) road and pad plot, (D) partial circular plot because of flooding and crop land

С

²⁶ They could also be a rectangle, for example, if there are limitations in the terrain.

B

D





Α







3-3-3-2 Road and Pad Plots

A road and pad (RAP) plot is a search plot restricted to ground surfaces or substrates that will be maintained free of vegetation during the operations phase of the WEF out to a fixed, pre-defined radius. Such cleared areas generally include the gravel area around the turbine base, the area maintained clear for crane access (crane pad), and portions of turbine access roads (Figure 3.3 C).

In many environments, such as croplands, forests, shrublands, and moist or wet climate zones, RAP search plots may be the only option because areas outside of the RAP areas are likely to be unsearchable,²⁷ but in environments such as steppes, deserts, and other areas that are more readily searchable, full search plots are recommended.

RAP plots have the benefit of being rapid and relatively easy to search and generally lead to higher searcher efficiency than the areas off the RAP. However, there is a tradeoff, because the proportion of the carcass fall area around a turbine that is covered by RAP surfaces is smaller than a full plot. Consequently, only a small proportion of the potential carcasses available for searchers to find will land within a RAP plot. This will result in low detection probability and fatality estimates with wide confidence intervals, resulting in a level of precision that may limit the ability to identify PCFM target species. For RAP designs, the fatality estimate is therefore very sensitive to the unsearched area correction.



Road and pad plots. Photos: Inkululeko Wildlife Services



²⁷ This is because vegetation growth impedes detection unless it is managed by mowing or other means, which is generally not recommended practice in emerging market countries given the potential impacts on local communities (e.g., croplands) and on biodiversity.

3.3.4 Transect Width

3.3.4.1 What Determines Transect Width?

Key Concept

Narrower transects will increase detection probability but will increase effort. Optimum transect width should be determined based on the size class of the PCFM target species.

To standardize search effort and methodology, searches should be conducted by walking along regularly spaced transects that collectively provide comprehensive coverage of the search plot. The transect spacing directly influences the proportion of carcasses that searchers find, particularly for smaller fatalities (notably bats), and this can affect the determination of PCFM target species. Optimum spacing of transects generally ranges from 4 to 20 meters and should be determined based on the size class of the PCFM target species and density of vegetation within the search plot, noting that smaller PCFM target species and vegetation and other ground conditions that hamper visual detection will generally require more closely spaced transects (Figure 3.4). Transects can be oriented in a variety of ways but are usually parallel and face a consistent direction (e.g., north-south) in all search plots. In RAP search designs, access roads within search plots may require only one or two search transects (along the road edge), whereas crane pads will require several regularly spaced, parallel transects, similar to larger search plots. For information on conducting searches with dogs see Box 3.3 below.

Figure 3.4 Examples of transect spacing (not to scale): (A) 6 m spacing for a study intended to capture data on bats and small birds and (B) 20 m spacing for a raptor-focused PCFM study

Note: Dotted lines indicate the walking path of a searcher. Transects should be no further than half the transect width from the edge of a plot.

25







Box 3.3 Fatality searches using dogs

In some more developed markets, where trained ecologists conduct all PCFM fieldwork, dogs have been used effectively to search for fatalities. Searches with dogs are particularly effective when PCFM target species are bats or small birds and where the search plot has dense vegetation or is otherwise difficult to search.



PCFM using dogs is a specialized technique that requires extensive dog and handler training, dog search protocols to ensure consistency of searches, resources to ensure animal welfare, and ongoing trials to assess dog search efficiency. This PCFM method should only be considered if trained ecologists conduct fieldwork and use dogs professionally trained for this purpose. At WEFs where both human and dog searches are conducted, each plot should be designated for search exclusively by a dog or a human search team. WEF environmental managers and LEs considering this approach are strongly encouraged to read published studies to fully understand the benefits, challenges, and limitations of using dogs in wildlife studies (e.g., Cablk and Heaton 2006; Wasser et. al. 2004) and, specifically, PCFM using dogs (e.g., Bennet 2015; Domínguez del Valle, Cervantes Peralta, and Jaquero Arjona 2020; Mathews et al 2013; Paula et al 2011).

3.3.5 Scanning Searches: An Alternative Approach to Transect-Based Searches for Large Birds

Where PCFM target species are large birds, and ground surface visibility allows searchers to see fatalities at long range, a systematic scanning search method, conducted from fixed locations around each turbine, is an alternative that can be used when it is impractical or unfeasible to conduct transect-based surveys.²⁸ Although scanning and transect search designs include the same PCFM components, aspects of the scanning search design and the analysis are more complex, and only advanced PCFM practitioners with assistance from a statistician should use this method. See Appendix B for further information.

²⁸ Transect-based and scanning searches should not be used together for a single turbine because it makes the analysis more complicated, although it would be acceptable to have transect-based searches at some turbines and scanning searches at other turbines in the same WEF. If combining scanning and transect-based searches at the same turbine is unavoidable, a statistician should be consulted.

3.3.6.1 What is the Search Interval?

Key Concept

Shorter search intervals result in fewer opportunities for carcasses to decompose or be removed by scavengers before searchers can find them, increasing detection probability and fatality estimate precision.

The search interval is the length of time between searches and is sometimes referred to as "search frequency." Every PCFM program should set a consistent search interval for each season.²⁹ The search interval directly influences the proportion of fatalities that searchers find and, in turn, the precision of the fatality rate estimate. There is an important relationship between the search interval and the average number of days that fatalities will remain on the ground before scavengers remove them, referred to as "carcass persistence" (see Section 3.3.7.3). With shorter search intervals, carcasses will be fresher when found, which will help with identification and reduce opportunities for carcass removal by scavengers or other means. As search interval decreases, the probability of a carcass the overall detection probability.

3.3.6.2 Factors Influencing Search Interval Selection and Carcass Size

Key Concept

Smaller carcasses tend to be removed more quickly than larger carcasses and therefore require a shorter (more frequent) search interval.



Carcasses are generally classified into four size classes: a) bats³⁰, b) small birds (<30 cm body length), c) medium birds (30-55 cm body length), and d) large birds (>55 cm body length). Smaller carcasses, especially bats, tend to have shorter persistence times than large birds, such as large raptors (see also Section 3.3.7.3 and Figure 3.5), which means that scavengers are more likely to remove them between searches. Other factors also affect the amount of time the carcass will persist. For example, carcasses sometimes disappear faster in wetter and warmer climates than in drier and colder climates. Carcass persistence times are highly variable (and could vary seasonally), and it is unlikely that they will be known in advance of the PCFM study.



²⁹ Minor deviations from this search interval are inevitable, but statistical fatality estimators, including GenEst, can easily handle them.

³⁰ Most bats are classified as small (<30 cm body length with a wingspan of <100 cm), although some of the larger fruit bats, known as flying foxes, in Asia Pacific and off the east coast of Africa have a body length greater than 30 cm and a wingspan greater than 100 cm. There is little PCFM information on these larger bat species, but for practical purposes, they may be classified in the same size class as medium birds.





Source: Confidential pooled data. Note: In all cases, persistence probability increases as search intervals decrease.

3.3.7 Bias Correction

Bias correction refers to mathematical adjustments to carcass counts that a statistical fatality estimator performs to produce an unbiased estimate of fatality. Some bias correction factors (e.g., unsearched area correction; Section 3.3.7.1) are calculated based on observed or known data. Others (e.g., searcher efficiency and carcass persistence corrections; Section 3.3.7.2 and 3.3.7.3, respectively) are calculated based on experimental data gathered especially for that purpose.

3.3.7.1 Unsearched and Unsearchable Area

What are unsearched and unsearchable areas?

Birds and bats fall at varying distances from the turbine base after colliding with a turbine blade; this is the "fall zone."³¹ Within the fall zone, fatalities that land within the designated PCFM search plot and are not removed by scavengers, are available to be detected by searchers. The others are

³⁹ The size of the fall zone depends on turbine height, blade length, wind regime, and carcass size and varies from study to study but can easily have a radius that is larger than the maximum turbine blade tip height.

not detectable by searchers because they are in areas that are not searched or cannot be searched. For example:

- > If the search plot does not cover the entire fall zone, fatalities that land **beyond the search plot** will not be detected (the unsearched area).
- > Some areas within the search plot may be **inaccessible to searchers** because of, for example, land access restrictions, dense vegetation, or hazardous terrain (the unsearchable area).

To produce unbiased fatality estimates, **it is important to account for carcasses that land in these unsearched and unsearchable areas of the fall zone**, especially if such areas cover a large proportion of the fall zone (e.g., in RAP search plots). Unsearched and unsearchable areas are accounted for by calculating the density weighted proportion (DWP; Huso and Dalthorp 2014) of the fall zone that is searched. The DWP allows fatality estimates to be adjusted to account for carcasses falling in unsearched and unsearchable areas. See below and Box 3.4 for further description of DWP.

Accounting for unsearched and unsearchable areas

The total size of the fall zone can be estimated based on turbine size and carcass fall patterns that have been reported in the scientific literature. In principle, it would seem simple to apply an area correction factor equivalent to the proportion of the area of the fall zone that the search plot covers and is searched, but this simple method is not valid, because it does not account for well-documented carcass fall patterns. Carcass fall density is not uniform within the fall zone; it is generally concentrated toward the base of the tower (Figure 3.6). Furthermore, the spatial patterning of carcass fall is typically different for different types of carcasses. For example, bat carcasses often fall close to the tower,³² whereas birds tend to fall in a more dispersed pattern (Figure 3.6). A number of statistical methodologies have been developed to account for this type of variation in carcass fall density within searched and unsearched areas (e.g., Dalthorp et al. 2020; 2022; Huso and Dalthorp 2014; Maurer et al. 2020; Studyvin et al. 2020), although most of these require proficiency with relatively advanced statistical methods and statistical software such as R (R Development Core Team 2018).

Users with the resources to explore such methods are encouraged to do so. To support users who may not have access to these statistical tools, the DWP tool in Appendix D of this Handbook presents a relatively simple method for estimating DWP, which is also outlined in Box. 3.4.



Unsearchable areas within search plots. Photos: Inkululeko Wildlife Services

³² Note that their fall pattern may be highly influenced by the local wind regime.

Box 3.4 Density Weighted Proportion and the DWP Tool

It is usually not feasible to search all areas beneath wind turbines where carcasses may fall; the carcass fall zone is large and may include areas that have impenetrable vegetation or other hazards that make it impractical to search. Adjusting fatality estimates to account for the proportion of carcasses that fall beyond the boundaries of plots (unsearched areas) or in areas that are impractical to search (unsearchable areas) is essential to production of an unbiased fatality estimate. This estimate is called the density weighted proportion of area searched to emphasize that the proportion of area searched is not the same as the proportion of carcasses in the search area. Wind wildlife ecologists in emerging market countries frequently mention the lack of a simple to understand method for estimating the DWP as a principal barrier to estimating fatality rates at WEFs. This is why the DWP tool provided in Appendix D has been developed.

Figure B3.4.1 shows the locations of simulated carcasses beneath a wind turbine. The green and blue areas represent unsearchable land. They are exactly the same size, but the green area includes about three times as much of the carcass fall density as the blue area. By accounting for unsearched area in rings (red circles) and considering carcass counts in rings, the DWP tool can estimate the area correction more accurately than if it simply used the proportion of area searched.

Figure B3.4.1 Carcass distribution within the fall zone around a turbine overlayed with rings used to calculate DWP



The DWP tool asks users to provide:

- > The proportion of area searched within each 5 m ring extending to the limit of the searched area around each turbine at the WEF
- > The carcass count (according to carcass size class) within each 5 m ring around all turbines at the facility

Then, rather than estimate the simple proportion of area searched, the tool uses the input data to estimate the proportion of carcasses occurring in each 5 m ring around the turbine and, together with the information about proportion of area searched, the tool estimates the DWP.

Figure 3.6 Schematic examples of carcass density distributions for four carcass size classes and proportion of areas searched on hypothetical 80 m radius full plots or road and pad plots



Source: Confidential pooled data. Note: These density distributions are indicative of general carcass fall patterns; the distances of carcasses from the turbine base at other facilities may vary.

3-3-7-2 Searcher Efficiency

Searcher efficiency and why to measure it

Searcher efficiency is the probability that a searcher will detect a carcass on the ground at the time of a carcass search. Despite best efforts, searchers may not detect carcasses within search plots for a variety of reasons: size, cryptic coloration, carcasses that are no longer intact or have been reduced to feather piles, vegetation or other ground conditions that hamper visual detection. Thus, every PCFM study will need to conduct searcher efficiency trials to measure the proportion of carcasses missed by searchers and then apply the resulting searcher efficiency correction factor in the statistical estimates of total facility-generated fatality rates.

Measuring searcher efficiency for different carcass size classes

Carcasses of smaller animals are harder to detect than those of larger animals. Carcass coloration may also influence detectability, which is a possible explanation for why bat carcasses (typically brown) are generally less detectable than small bird carcasses, which are more variable in coloration. For these reasons, searcher efficiency is typically measured and characterized separately for each of four carcass size classes (bats, small birds, medium birds and large birds), with searcher efficiency trial carcasses in each size class reflecting as closely as possible the characteristics of the PCFM target species at the site. See Box 3.5 on obtaining carcasses for use in bias correction trials.

Box 3.5 Obtaining carcasses for bias correction trials: challenges and alternative approaches



Ideally, bias correction trials should use bird and bat collision fatalities found at the WEF that are kept fresh and stored in an on-site chest freezer at the project site, but at many WEFs, the number of bird and bat carcasses found is well below the sample size needed to run valid bias correction trials. In such cases, suitable surrogates and other approaches are needed.

Searcher efficiency and carcass persistence trials have different constraints and criteria governing the suitability of surrogates. For searcher efficiency trials, it is most important that surrogates visually resemble actual collision fatalities in the eyes of (human) searchers. They should therefore be of similar size, shape, and color as actual fatalities but smell is unimportant. For this reason, non-animal surrogates can be suitable for searcher efficiency trials.

For carcass persistence trials, in addition to being of similar size, shape, and color, surrogates should smell like the actual fatalities recorded so that the same scavengers detect them over similar timescales. Surrogate carcasses such as domesticated poultry and game birds are more palatable to carcass scavengers (and hence removed more quickly) than carcasses of the large birds that they could be used to surrogate for, specifically raptors. It is likely that using these types of surrogates for raptors will inflate fatality rate estimates for this group and therefore should be avoided (but see, e.g., Hallingstad et al. 2023). This is particularly important because raptors will often be PCFM target species at a WEF. Importantly, the LE must ensure that all carcasses acquired for bias correction trials regardless of their source must be legally obtained.

Surrogate carcasses that may be acceptable for searcher efficiency and carcass persistence trials and have been used in PCFM studies for this purpose:

- > Farm-raised ducks and pigeons as surrogates for non-raptor bird fatalities
- > Dark-colored adult mice as surrogates for bats
- > Fresh road-killed birds and bats
- Freshly killed and frozen birds and bats that non-governmental organizations or government officials have confiscated from illegal possession (through hunting or trafficking)
- Freshly killed and frozen bird and bat carcasses available from lethal control undertaken at airports or elsewhere

Surrogate carcasses that should not be used in searcher efficiency or carcass persistence trials:

- > Animals suspected of being poisoned
- > Domestic poultry
- Game birds as raptor surrogates (see DeVault et al. 2017; Urquhart, Hulka, and Duffy 2015) unless it can be demonstrated that persistence rates can be reliably extrapolated (see, e.g., Hallingstad et al. 2023)
- > Any bird or bat acquired by killing individuals from wild populations

When insufficient fatalities or suitable surrogates are available, alternate approaches for implementing bias correction trials may be needed. Examples include the following:

- For searcher efficiency trials, a variety of artificial decoys may be suitable, provided that they are of similar size and color to the carcasses likely to be found at the WEF. These include hunting bird decoys and toy bats. It is generally acceptable to re-use artificial decoys in multiple searcher efficiency trials if they remain reasonably intact.
- For carcass persistence trials, it may be possible to conduct joint trials with neighboring projects if located within the same habitat with a similar composition of scavengers. Data could be pooled for use at multiple projects. For certain species, especially raptors that persist for long periods, some organizations or regions are developing standardized carcass persistence rates that can be used in fatality estimation calculations (Hallingstad et al. 2023; Wilson, Hulka, and Bennun 2022).

When and where to measure searcher efficiency

Searcher efficiency should be measured throughout the study period and over the entire spatial extent of the WEF to capture possible variation over time and space, for example, as a result of seasonal vegetation growth. Searcher efficiency trials work best when they are spaced regularly throughout the season and widely across sampled areas so that estimated searcher efficiency does not depend on conditions during a single day or within a limited portion of the WEF. For practical information on conducting searcher efficiency trials, see Appendix B and then use the Searcher Efficiency Trial Placement Tool³³ in Appendix D to select random placement of carcasses.

Using visibility classes in searcher efficiency models

Searcher efficiency is sometimes measured separately for each visibility class present in the search plots (e.g., Easy — bare ground 90% or greater, all ground cover sparse and 15 cm or less in height; Very Difficult — little or no bare ground, more than 25% of ground cover over 30 cm in height), and doing so improves the accuracy of fatality estimates, albeit at the cost of increased field effort to map visibility classes and conduct searcher efficiency bias trials in each type of visibility class. Although this Handbook does not specifically suggest including visibility class in the GenEst analysis of searcher efficiency models to reduce fieldwork effort and the number of trial carcasses or decoys required, readers should be aware that it is a practice in some regions, and an option if desired.

³³ This tool generates a random distance and compass direction from a turbine for each searcher efficiency trial carcass. At the start of the trial, carcasses are placed in the locations generated by the tool.



Turbine collision fatalities of different carcass size classes. Left: Big free-tailed bat (*Nyctinomops macrotis*). Photo: Confidential source. Center: Eurasian Sparrowhawk (*Accipiter nisus*). Photo: Alvaro Camiña. Right: Booted Eagle (*Hieraaetus pennatus*) Photo: Photo: Alvaro Camiña

3-3-7-3 Carcass Persistence

What is carcass persistence and why measure it?

If a fatality occurs and the carcass is removed or disappears from the search plot before the next search visit, ³⁴ searchers cannot detect it. If the proportion of carcasses that disappear (before searches are conducted) is not estimated and factored into total fatality rate estimates, fatality rates will be biased downward (underestimated). Carcasses may be removed or disappear from search plots for a variety of reasons, including scavenging, weathering, decomposition, and human-related activities (e.g., agricultural practices). "Carcass persistence" refers to how long a carcass remains or "persists" on the ground before one of these factors removes it. Because of the recognized importance of this bias factor, PCFM studies should incorporate specialized carcass persistence field trials to quantify carcass persistence rates at the site and then apply the resulting carcass persistence correction factors in the statistical estimation of total facility-generated fatality rates.

Why is carcass persistence important for determining search interval?

As explained in Section 3.3.6, carcass persistence is fundamentally linked to search interval. Because carcasses disappear over time, more frequent searches increase the chances that a collision-generated carcass will be present at the time of the next search.

When and where to measure carcass persistence

Similar to searcher efficiency trials, carcass persistence should be measured throughout the study period and over the entire spatial extent of the WEF to capture possible variation in carcass persistence over time and space, for example as a result of seasonal variation in scavenger activity. Carcass placement should be limited to a maximum of two trial carcasses at any one turbine at any one time to avoid undue attraction of scavengers. For additional information on obtaining carcasses for use in bias correction trials, see Box 3.5. For technical information on conducting carcass persistence trials, see Appendix B.

³⁴ In this Handbook, a "search visit" is a single search of all sample turbines.





Highly decomposed large bird carcass. Martial Eagle (Polemaetus bellicosus). Photo: Inkululeko Wildlife Services



3.4 PCFM COMPONENTS—WEF-ASSOCIATED OHL SEARCHES

WEFs typically require OHLs to transport energy that turbines generate to substations and electrical grids. This infrastructure presents a potential collision risk to birds and an electrocution risk to birds and bats and requires a dedicated PCFM design to quantify WEF-associated OHL fatalities and to assess the overall impact of a WEF. Collision and electrocution risk at OHLs is increasingly relevant to emerging market countries with high numbers of potentially susceptible species and a rapidly expanding infrastructure (Bernardino et al. 2018; Martín Martín et al. 2022; Shaw et al. 2021).

This section outlines risks to birds and bats associated with this infrastructure and highlights similarities and differences in the underlying principles determining turbine and OHL PCFM designs.

3.4.1 Risks from OHLs

The conductor wires that form part of WEF-associated OHLs present a potentially high collision risk to birds as they are difficult to detect and occupy low-altitude airspace where flight activity is often highest. OHLs may extend through considerable proportions of habitat used by collision susceptible species. Heavy-bodied birds with limited maneuverability such as bustards, cranes, waterbirds, and large raptors are particularly susceptible to collision with OHLs. Potential impacts are especially high when OHLs are constructed in areas used by species at risk of multiple collisions because they typically fly in flocks, and in areas where collision susceptible species congregate, such as wetlands, migratory corridors (especially bottleneck areas), colonial breeding, and roost sites (Bevanger 1998; De La Zerda and Rosselli 2002; Prinsen et al. 2011; WWI 2018).

Birds and large bats are electrocuted on OHLs, particularly medium-voltage distribution lines, where they come into simultaneous contact with energized and grounded components (e.g., Biasotto et al. 2022 Dixon et al. 2015; Tella et al. 2020). Any above-ground collector lines are likely to present this risk, particularly for raptors, including vultures, owls, falcons, and eagles). These birds are attracted to pylons and distribution line poles because they provide nest locations and perches for resting or to use during hunting. For bats, the principal known threat is to flying foxes (*Pteropus spp.*), which are known to hang from OHLs and are electrocuted when their wings touch the wires (Chouhan and Shrivastava 2019; Rajeshkumar, Raghunathan, and Venkataraman 2013).



American Kestrel (Falco sparverius). Brazil. Photo: Alvaro Camiña

3.4.2 Comparing PCFM Designs for Turbines and OHLs

A PCFM program conducted along a WEF-associated OHL has the same components as a turbine search (e.g., plot size and shape, transect width, search interval, bias correction trials) and the same overarching objective of characterizing wildlife impacts to inform potential mitigation and adaptive management during the operational phase of the project. There are two key differences:

- Some of the underlying principles supporting design decisions differ because OHLs are static linear structures rather than a collection of separate structures rotating from fixed locations.
- > PCFM for OHLs will reveal fatalities caused not only by collision, but also by electrocution.

Table 3.1 lists the components used to design PCFM programs for turbines and highlights some of the key differences from designs for OHLs. The topic of above-ground collector lines and electrocution risk is further discussed in Section 4.2.1.1.

3.5 CHANCE FINDS

During a PCFM program, search teams and other personnel (e.g., operations, maintenance personnel, WEF contractors) may discover bird and bat fatalities that are not part of the systematic

Table 3.1 Comparison of turbine and OHL PCFM design components

	PCFM CO	MPONENT	TURBINE DESIGN	OHL DESIGN	RELATED INFORMATION	
FATALITY SEARCHES		STUDY DURATION	Turbine and OHL designs based effects on PCFM target species.	See Section 3.3.1.		
	TURBINE SAMPLE		Sample turbines randomly selected	The OHL sample is not random: sample section location and sampling rate depend on expected level of risk to susceptible species within landscape along OHL.	See Section 4.2.1 for more details about defining OHL sample.	
	K K V V	SEARCH PLOT SHAPE AND SIZE	SEARCH PLOT SHAPE AND SIZESquare or circular plot centered on turbine informed by fatality fall zone and determined by adequate detection probability value.Corridor on either side of OHL center line defined according to length of each OHL sample search plot and extent of fall zone scaled according to maximum height of OHL.		See Section 3.3.7.1 for underlying principles relating to fall zone. See Section 4.2.2 for more details about defining OHL search plot size.	
	۰] د	TRANSECT WIDTH	Turbine and OHL transect widt probability for species and size of the species and species	See Section 3.3.4 for underlying principles involved in selecting transect width. See Section 4.2.3 for more details about defining OHL transect width.		
		SEARCH INTERVAL	Turbine and OHL search interva persistence time for species and	al based on estimated carcass d size class(es) of interest.	See Section 3.3.6. for underlying principles involved in selecting search intervals. See Section 4.2.4 for more details about defining OHL search interval.	
BIAS CORRECTION		UNSEARCHED AND UNSEARCHABLE AREA	See Section 3.3.7.1 for underlying principles involved in determining unsearched and unsearchable areas and why they are important in calculating fatality estimates. See Section 4.2.5.1 for more details about defining OHL unsearched areas.			
		SEARCHER EFFICIENCY TRIALS	OHL searcher efficiency trials b. principles as for PCFM for turbin	See Section 3.3.7.2 for underlying principles relevant to designing searcher efficiency trials. See Section 4.2.5.2 for more details about designing OHL searcher efficiency trials.		
	J.	CARCASS PERSISTENCE TRIALS	See Section 3.3.7.3 for underlying principles relevant to designing carcass persistence trials. See Section 4.2.5.3 for more details about designing OHL carcass persistence trials.			

PCFM plot searches for turbines or WEF-associated OHLs. Fatalities may be found within plots but between search visits or in areas outside defined search plots (referred to as off-plot discoveries).

3.5.1 Chance Find Fatalities Found Within Search Plots

Collision fatalities discovered within search plots *that are not found* as part of systematic PCFM searches are typically *included* in the sample of fatalities used to derive total, bias-corrected fatality rate estimates for the WEF. The assumption is that these fatalities would have been discovered within that search plot during the next search. It is important that these fatalities be included in the sample used to derive fatality estimates only if they resulted from collision with turbines or collision or electrocution at WEF-associated OHLs during the study period. If the fatality resulted from another source (e.g., poisoning) or occurred outside of the study period, it should be recorded and reported but not included in the fatality rate estimate sample.³⁵

3.5.2 Chance Find Fatalities Found Outside Search Plots

Fatalities discovered at any time outside of search plots (off-plot discoveries) are not included in the sample of fatalities to derive total bias-corrected fatality rate estimates for the WEF. Although off-plot carcass discoveries do not influence estimation of bias-corrected total fatality rates, they provide valuable information regarding the species composition of fatalities at the WEF and are important for understanding impacts especially with respect to thresholds for PCFM target species. (for more information on threshold-setting see Section 6.2).

All chance find fatalities at a WEF or associated OHL (including non-collision, non-electrocution fatalities) should be recorded on a fatality record form (Appendix E), and the carcass should be removed and stored in the same way as PCFM carcasses so that it is not found a second time. There is a section on the fatality record form to indicate whether the fatality was a within-plot or off-plot chance find and the likely cause of death (turbine collision, WEF-associated OHL collision or electrocution, other).



Red Kite (Milvus milvus). Photo: Alvaro Camiña

³⁵ At times, accurately determining cause of death or estimating time of death is difficult in the field. Consulting with a specialist to perform a necropsy may be necessary, particularly for globally threatened and PCFM target species.

PCFM DESIGNS AND USING THE DST

4. PCFM DESIGNS AND USING THE DST

Section 4 builds on the core concepts and principles of PCFM design described in Section 3 and provides guidance to develop real-world PCFM designs for WEFs in emerging market countries. For turbine searches, the DST (Appendix A) is key to that process, and this section outlines in detail how to use it. Standardized PCFM designs for OHLs are still evolving, so the DST does not attempt to automate this process, but recommended design approaches based on GIIP are included for WEF-associated OHLs in Section 4.2.

4.1 TURBINE PCFM DESIGNS

This Handbook and its DST recommend two PCFM design approaches for turbine searches:

- > A generalized **First Year** design for use during the first year(s) of operations.
- > An optimized **Subsequent Years** design informed by the first year(s) of PCFM data and the project's monitoring objectives.

Section 4.1.2 explains the purpose of the First Year design, provides advice on using the DST to generate a Year 1 design and then explains the DST First Year design output. Section 4.1.3 follows the same format and provides equivalent information for the Subsequent Years design.

Importantly, it should be noted that **all fatalities found during PCFM surveys are recorded regardless of whether a First Year or a Subsequent Years design is being implemented and regardless of the PCFM target species.** For example, a Subsequent Years design that is optimized for large birds with a relatively wide transect width and long search intervals will not be optimized for bats or smaller birds, but carcasses in these size classes will be found during searches and should always be recorded.

4.1.1 Using the DST

The DST is operated using a single Microsoft Excel spreadsheet. The reader is encouraged to open the DST (Appendix A) when reading this section to trial its content. Within the spreadsheet, the user will find 11 worksheet tabs. The first tab is labeled "Instructions," and users are expected to interact with the DST through the four tabs labelled "User interface..." (Figure 4.1). The remaining seven tabs run DST logic and calculations. Users cannot interact with these, but they are visible to allow users to understand the tool's mechanics.

The content and purpose of the 'User Interface...' tabs are summarized below.

Figure 4.1 DST User Interface tabs

×	AutoSa	ave On	F						₽ Sei	arch				Ø	-		×
File	Но	me Insert	Draw	Page Layout	Formulas	Data	Review	View	Automate	Help				Con	nments	🖻 Sha	ire 👻
B20		V I X V	fx														~
-	А								В								
1																	
3																	
4	•	Instructions	User	nterface-Proje	ct Setup	Jser Inte	erface-Yea	ar 1 🛛 Us	ser Interface-	Subsequent Year	s User Interfac	e-Desigr	n Explore	er (): (-	•
Ready	SX AC	cessibility: Investi	gate								Display Settings	Ħ		<u> </u>	-	+	100%

ctions User Interface-Project Setup User Interface-Yee

User Interface — Project Setup

Asks three questions about WEF specification relevant to both the First Year and to the Subsequent Years design.

etup User Interface-Year 1 User Interface-Subsequent Ye

User Interface — First Year

Asks three questions about practical constraints related to the PCFM during the first year. These are used to adjust the First Year design so that it is applicable to the specific WEF. This tab generates a generalized design for the first year(s) customized to specifications and practical constraints at the WEF.

ar 1 User Interface-Subsequent Years User Interface-De

User Interface — Subsequent Years

Asks 14 questions about the First Year design PCFM results, project monitoring objectives and practical constraints on PCFM to build a profile of the WEF. Based on answers to these questions, the tab generates a recommended design optimized for PCFM target species.

ent Years User Interface-Design Explorer

User Interface — Design Explorer

Allows the user to input two alternative PCFM designs and compare these with the recommended design that the Subsequent Years tab generates. The Design Explorer is provided to allow flexibility to test and evaluate the validity of alternatives to the recommended Subsequent Years PCFM design. This flexibility may be needed if the project has specific constraints, and the Design Explorer will help ensure that these constraints result in a PCFM design that can fulfill the principal objective of providing robust, reasonably precise fatality rate estimates for PCFM target species.



Lead Ecologist explaining PCFM. Photo: Caleb Gordon

4.1.2 First Year Design

At the start of the first year of monitoring, no prior knowledge of fatality rates at the WEF is assumed. The generalized First Year PCFM design provides information about any species and size classes occurring as fatalities at turbines. The objective is to adequately characterize wildlife collision impacts for the first year(s) and determine PCFM target species so that optimized, targeted monitoring can progress in subsequent years. Although this generalized First Year design is recommended, the LE may need to consider additional site-specific habitat and ecological factors, as well as priority biodiversity values of particular importance to stakeholders when making the final First Year design decisions. All deviations from the DST First Year design should be justified and documented. Lastly, although intended for one year, the generalized PCFM design might need to be used for multiple years, as explained in Section 4.1.3.

4.1.2.1 DST Input — First Year Design

For ease of use, most PCFM components for the generalized First Year design are predetermined, and user input is limited to some basic information about the WEF. These predetermined values have been carefully selected so that, collectively, the results will achieve optimal detection probability for a generalized First Year design given the characteristics of the WEF. To generate a recommended PCFM design for the first year(s) for a specific WEF, the LE needs to input the following details into the DST.

User Interface-Project Setup

Open the **User Interface** — **Project Setup tab** and answer questions 1 to 3 about the WEF (Table 4.1).

Table 4.1	DST Project Set-up tab: Questions 1-
-----------	--------------------------------------

QUESTION	USER INPUT
1. How many turbines are at the facility (1-999)?	WEF-specific answer
2. What is the hub height of the turbine (m)?	WEF-specific answer
3. What is the blade length of the turbines (m)?	WEF-specific answer

User Interface-Year 1

Open the **User Interface** — **First Year tab** and answer questions 4 to 6 about the area beyond the RAP (Table 4.2).

Q	JESTION	USER INPUT			
4.	Is it legal to access areas not on the RAP		Yes/No		
	Legality of access extending to maximum radius of full plot searches should be clarified with WEF management.				
5.	Is it safe to access areas not on the RAP ?	2.	Yes/No		
	Hazards such as ruggedness of terrain and risks from animals will determine whether full plot searches are possible. Health and safety risks and protocols should be discussed with WEF management.				
6.	. Is vegetation beyond the RAP conducive to searching?		> Yes (no more than 75% of the surface		
	The LE should (qualitatively) assess for the WEF overall how difficult it is to walk transects off the RAP. This should be assessed during the pre-PCFM visit to each turbine (see Appendix B 'Tasks to be completed by the Lead Ecologist before the start of PCFM fieldwork').	>	has vegetation exceeding 30 cm in height) No (at least 75% of the surface has vegetation exceeding 30 cm in height)		

4.1.2.2 DST Output — First Year Design

As mentioned in 4.1.2.1 most of the DST First Year design output values are predetermined and will be the same regardless of the project for which the design is being developed. For example, the recommended transect width will always be 6m, search interval will always be 7 days and every design will recommend searching all turbine RAPs. Other values such as number and size of full plots will differ between projects depending on the size of the WEF and the turbine specifications. The DST is used to generate only the project-specific values with respect to the First Year design.

Once answers to the six questions above are entered into the DST, the generalized First Year PCFM design is generated with respect to the project site on the right-hand side of the User Interface — First Year tab.

This provides:

- > Number of turbines to search with RAP and full plots
- > Full plot search radius
- > Transect width
- > Length of access roads to search when conducting RAP searches
- > Search interval
- > Number of searcher efficiency trial carcasses / decoys to use per size class per season
- > Number of carcass persistence trial carcasses to use per size class per season

No DPI value is available for the generalized First Year design because results from searcher efficiency and carcass persistence trials conducted during the first year are required to estimate DPI. The DPI is available only for the Subsequent Years design.

The First Year design component outputs are explained in detail below and also appear as output values in the DST First Year tab.



Design

- > RAP searches conducted at all turbines, plus
- > Full plot searches at a predetermined proportion of turbines, depending on the size of the WEF
 - ≤50 turbines search all turbines.
 - 51-100 turbines search 50 turbines OR 75% of turbines, whichever is greater.
 - >100 turbines search 75 turbines OR 50% of turbines, whichever is greater.

Justification

Determining the number of turbines to search is key to achieving an adequate detection probability for collision-susceptible species using the airspace in the WEF (see Section 3.3.2). The goal of the generalized First Year design is to achieve a search effort across the whole WEF that produces adequate detection probability for all size classes that have the potential to occur as fatalities.

Searching RAPs at all turbines is quicker and easier than full plot searches, provides fatality information from across the whole WEF regardless of its size, and reduces the likelihood of missing turbine fatality hotspots. However, a RAP-only design will cover only a small proportion of the fall zone at each turbine and so will be able to detect only a small proportion of fatalities potentially present, resulting in low detection probability. This will lead to imprecise fatality estimates, which can limit the ability to identify PCFM target species, a key objective of the First Year design. Searching only the RAP may also miss species that fall further from the turbine. To address these limitations, the First Year design includes full plot searches at a randomly selected, predetermined proportion of WEF turbines.

Possible design adjustments

In some cases, vegetation or other ground conditions will make it impossible to search outside of the RAPs (see Section 3.3.3.2), in which case, just RAPs will be used in the First Year design, with all turbines included in the sample.

Although random selection of turbines for full plot searches is ideal, there may be logistical constraints (e.g., land access) that preclude some turbines as candidates for full plots.



Search Plot Shape and Size

Design

RAP — conduct searches of turbine and crane pad area plus any access roads to a distance from turbine base equivalent to the maximum blade tip height.

Full plot — conduct searches to a radius equivalent to half the maximum turbine blade tip height.

Justification

In the First Year design, the maximum tip height of turbines at the WEF influences the size of full plots and the distance to which access roads are searched during RAP searches. This is because the extent of the fatality fall zone around a turbine is broadly related to the size of the turbine (see Section 3.3.3). Linking search plot size to maximum blade tip height in the design method means that search plots are consistently scaled to the fall zone size for any WEF turbine specification.

The First Year design full plot search area accounts for the increasing (human and financial) resources required to search large areas toward the limit of the fall zone and the lower proportion of fatalities found in these outlying areas. Based on a review of fall distance data (e.g., AWWI 2020a; 2020b), limiting full plot searches to a radius equivalent to half the maximum turbine blade tip height has the potential to detect the majority of bird and bat fatalities occurring at a turbine (see Section 3.3.7.1 and Figure 3.6 for more details on fall distances from turbines).

For RAP, searching access roads to a distance equivalent to the blade tip height provides important information about the overall fall zone of fatalities and allows the number of carcasses falling beyond the search area on the turbine and crane pads or within full plots to be estimated. This information is used to improve the validity of the DWP output, which is an essential input into the GenEst fatality rate analysis (Box 3.4 gives more information on DWP).

Possible design adjustments

The DST assumes that full plots will be circular. This was necessary for streamlining the DST, but users may elect or need to choose a square rather than a circular full plot in the field. In these cases, the estimates provided in the DST will be sufficiently accurate.



Transect width

Design

- > RAP searches 6 m
- > Full plot searches 6 m

Justification

Narrower transect widths result in higher detection probability and more precise fatality estimates. This is particularly relevant for smaller, more difficult to detect species, such as bats, and where environmental conditions (e.g., substrate, vegetation) result in moderate to low visibility within search plots (see Section 3.3.4.1). Before PCFM, not all PCFM target species may be evident, so the design must adequately capture fatality information from all fatality size classes to provide an overview of fatality risk at the site and determine PCFM target species. Deciding on a generalized design with a single transect width for all fatality size classes is challenging. If transects are too

wide, detection probability becomes low, especially for bats and small birds, and estimates may be too imprecise to identify PCFM target species. Conversely, if transects are too narrow, human and financial resources required to conduct the monitoring may be unfeasibly high (see Section 3.2.4 for more details on detection probability).

Possible design adjustments

A 6 m transect width is recommended, however, if visibility is high at the WEF site and its surroundings and there is existing PCFM information from the area indicating low risk to bats, a 10 m transect width is a reasonable alternative.



Search interval

Design

- > RAP searches every 7 days
- > Full plot searches every 7 days

Justification

Deciding how frequently to conduct a search is related to how long a fatality persists on the ground before it is no longer available for search teams to find it, with longer average persistence times for larger birds and raptors than for small birds and bats (see Section 3.3.6.1). The search interval for the First Year design is set at 7 days so that, when combined with other PCFM component values (e.g., plot size, transect width), the resulting fatality estimates for all size classes should be sufficiently precise to identify PCFM target species.



Bias correction: Searcher efficiency trials

Design

> 12 trials (carcasses or decoys) per season per size class

Justification

Correcting for the number of fatalities that searchers miss because they are difficult to detect due to their size, color, or condition or the ground conditions at the search plot is essential for calculating unbiased fatality rate estimates (see Section 3.3.7.2). The objective of the generalized First Year design is to provide fatality rate estimates for all species and size classes that could collide with turbines at the WEF so that PCFM target species can be better identified. Searcher efficiency trials should therefore be conducted for all size classes.

The First Year design recommends placing 12 trial carcasses or decoys per size class per season for searcher efficiency trials. For example, a PCFM program with four size classes and four seasons

would require 192 trials (12 carcasses x 4 size classes x 4 seasons). For two seasons (e.g., wet and dry), this would require 96 (12 x 4 x 2) trials. This should provide valid searcher efficiency values for each season and for each fatality size class at the WEF turbines (see Appendix B for practical advice on conducting searcher efficiency bias correction trials).



Bias correction: Carcass persistence trials

Design

> 12 trials (carcasses) per season per size class

Justification

Correcting for the number of fatalities not found during searches because scavengers remove them or they disappear because of weathering, decomposition, or human-related activities is essential for calculating unbiased fatality rate estimates (see Section 3.3.7.3). The goal of the generalized First Year design is to provide fatality rate estimates for all species and size classes that could collide with turbines at the WEF so that PCFM target species can be better identified. Carcass persistence trials are therefore recommended to be conducted for all size classes.

The First Year design recommends placing 12 trials per size class per season for carcass persistence experiments, which should provide valid carcass persistence values for each season and for each fatality size class at the WEF.

Possible design adjustments

Sourcing sufficient carcass persistence trial carcasses is often challenging because there may be few WEF fatalities, and suitable surrogates may be difficult to obtain. In these situations, it may be valid to conduct joint trials with neighboring projects or use available standardized persistence rates (see Box 3.5 for more information).



Bias Correction: Unsearched and unsearchable area

Although not an output of the DST, correcting for the number of fatalities not detected because the area was unsearched or unsearchable is essential for providing unbiased fatality estimates. To achieve this, in the First Year design, the LE will map all unsearched and unsearchable areas before fieldwork begins (see Appendix B *Tasks to be completed by the Lead Ecologist before the start of PCFM fieldwork* for detailed practical advice). This information will be used in conjunction with the location of fatalities within the searched area to calculate the DWP (see Section 3.3.7.1 for more information on the unsearched area, Box 3.4 for principles related to DWP, and Appendix D for the DWP tool used to calculate DWP). Unlike the First Year design, there is no prescribed Subsequent Years design. Rather, the DST is developed to inform decision making, taking the user through a series of questions about the First Year design PCFM results, project monitoring objectives, and practical constraints on PCFM at the WEF. The Design Explorer tool within the DST is provided to enable the initial Subsequent Years design to be customized to account for this site-specific knowledge (see Section 4.1.3.3). As with the First Year design, the LE should always guide Subsequent Years design decisions accounting for site-specific knowledge about the WEF and bird and bat ecology at the site.

After implementing the First Year design for the initial year of the PCFM program, the LE should consider whether the fatality data collected are adequate to determine the PCFM target species. If not, the First Year design may need to be repeated for one or more additional years before moving forward with the more tailored Subsequent Years design. Possible reasons for a WEF needing to repeat the generalized First Year design include the following:

Fatality risk is likely to be different between years, for example, if:

- > The WEF is near a migratory corridor, and migratory movements through the area are known to vary locally between years.
- > Weather or other environmental conditions during the first year are exceptional, resulting in atypical numbers of species or individuals at the WEF.
- > The WEF was not fully operational during the first year. This may be particularly important if non-operational turbines are in areas of potentially high risk to birds or bats.

The PCFM design was not implemented fully, for example, if:

- Unforeseen environmental constraints resulted in a reduced search schedule (e.g., exceptional weather conditions limiting access to search plots or making working conditions unsafe for searchers).
- There were practical problems associated with conducting the PCFM program (e.g., significant weaknesses in how the fieldwork was conducted or the data collected or problems acquiring carcasses for bias correction trials).

Individually or together, these types of situations may prompt the LE to conclude that the results have not adequately identified the PCFM target species and therefore require a repeat of the First Year design before a more tailored Subsequent Years design is implemented using the DST.

4.1.3.1 DST Input — Subsequent Years Design

The Subsequent Years design tab is used to optimize the PCFM design once the PCFM target species are better understood through implementation of the First Year design. When using the DST to develop a Subsequent Years PCFM design, the size class of the PCFM target species, ground visibility and recent carcass persistence data are especially important in determining design parameters.

The DST Subsequent Years design tab should be used to inform and update the PCFM design at least annually as conditions and PCFM target species may change during the life of the project or PCFM study duration period.

To generate a recommended Subsequent Years PCFM design the LE needs to input the following information into the DST:

User Interface-Project Setup

Open the **User Interface-Project Setup** tab and check that First Year design responses to questions 1-3 about the WEF (Table 4.1) are correctly entered.

User Interface-Subsequent Years

Open the **User Interface — Subsequent Years tab** and answer questions 1-14 to build a profile of the WEF informed by First Year design PCFM results, project monitoring objectives, and practical constraints on PCFM (Table 4.3).

Table 4.3 DST Subsequent Years tab: Questions 1-14

Q	JESTION	USER INPUT		
1.	How large are the turbine pads (average square meters)?	WEF-specific estimate		
2.	Do first year(s) fatality data indicate that the annual turbine fatality rate for <i>large birds</i> (over 55 cm) at the project is likely to be a concern ^a or do monitoring objectives for the project focus on <i>large birds</i> for some other reason?	Yes/No		
3.	Question 2 applied to medium birds (30 -55 cm) and larger fruit bats (>30cm)	Yes/No		
4.	Question 2 applied to small birds (<30 cm)	Yes/No		
5-	Question 2 applied to bats (<30 cm)	Yes/No		
	Fatality concerns for larger fruit bats (also known as flying foxes) should be addressed in Question 3 'medium birds' rather than Question 5)			
6.	Is it legal to access areas not on the RAP?	Yes/No		
	Legality of access extending to maximum radius of full plot searches will determine whether full plot searches are possible and should be clarified with WEF management.			

7-	Considering the easiest to search areas, how difficult will it be to search for carcasses? ^b To help determine an appropriate transect width and the DPI for a PCFM design, the DST needs to incorporate the extent to which vegetation cover affects detection of fatalities. Question 7 asks the user to describe the vegetation cover in the easiest to search areas (not including the surface of the RAP). This, combined with the response to question 8, which describes the vegetation cover in the most difficult to search areas provides an assessment of vegetation cover needed to determine transect width and inform the DPI. (The LE should use the mapped visibility class information collected during the visit to each turbine before the start of PCFM fieldwork to determine how difficult it is to search the least vegetated search areas (see section in Appendix B: Tasks to be completed by the Lead Ecologist before the start of PCFM fieldwork). Based on this assessment, one of four options should be selected from the drop-down list in the User Input column	 Very Difficult: Little or no bare ground; more than 25% of ground cover over 30 cm in height Difficult: Bare ground 25% or less; 25% or less of ground cover over 30 cm in height Moderate: Bare ground 25% or greater; all ground cover 15 cm or less in height and mostly sparse Easy: Bare ground 90% or greater; all ground cover sparse and 15 cm or less in height
8.	Considering the most difficult to search areas, how difficult will it be to search for carcasses? ^b To help determine an appropriate transect width and the DPI for a PCFM design, Question 8 asks the user to describe the vegetation cover in the most difficult to search areas (not including the surface of the RAP). This, combined with the response to question 7, provides the assessment of vegetation cover needed to determine transect width and inform the DPI. As with Question 7, the LE should use the mapped visibility class information collected during the visit to each turbine before the start of PCFM fieldwork to determine how difficult it is to search the most-vegetated search areas.	 Very Difficult: Little or no bare ground; more than 25% of ground cover over 30 cm in height Difficult: Bare ground 25% or less; 25% or less of ground cover over 30 cm in height Moderate: Bare ground 25% or greater; all ground cover 15 cm or less in height and mostly sparse Easy: Bare ground 90% or greater; all ground cover sparse and 15 cm or less in height
9.	How difficult is it to walk transects off the RAP? Rugged terrain will make searching more difficult and reduce searcher efficiency, which will be reflected in the DPI values in the DPI table below the Recommended Design table. The LE should (qualitatively) assess how difficult it is to walk transects beyond the RAP surface. This should be assessed during the pre-PCFM visit to each turbine (see section in Appendix B: Tasks to be completed by the Lead Ecologist before the start of PCFM fieldwork). Based on this assessment, one of four options should be selected from the drop-down list in the User Input column.	 Impossible: Terrain is such that searchers cannot safely traverse the area while looking for carcasses Difficult: Hilly to steep landscape with abundant rocks, hummocks, hollows, or vegetation that requires a surveyor's attention Moderate: Hilly landscape, or moderate abundance of rocks hummocks, hollows, or vegetation that require a surveyor's attention Easy: Flat to gently rolling landscape with little vegetation and few to no rocks, hummocks, hollows, or other obstacles on the ground
10	. Is it safe to search off the RAP? Hazards such as ruggedness of terrain and risks from animals will determine whether full plot searches are possible. Risks and WEF health and safety protocols should be discussed with WEF management.	Yes/No
11.	Based on carcass persistence data from the project area or from a comparable wind project in the vicinity, ^c what is the likely median removal time (days) of a large bird? The LE will refer to the median carcass persistence time for large birds calculated from the WEF's first year(s) carcass persistence trials (or comparable results from the immediate vicinity) to estimate removal time.	 > less than 1 day > 1-3 days > 4-6 days > 7-9 days > 10-15 days > 16-20 days or unknown > 20-30 days > more than 30 days

12. Question 11 applied to medium birds (30-55 cm) and larger fruit bats (>30 cm, if they occur in the region)	 less than 1 day 1-3 days 4-6 days 7-9 days or unknown 10-15 days 16-20 days 20-30 days more than 30 days
13. Question 11 applied to small birds (<30 cm)	 > less than 1 day > 1-3 days > 4-6 days or unknown > 7-9 days > 10-15 days > 16-20 day > 20-30 days > more than 30 days
14. Question 11 applied to bats (<30 cm)	 less than 1 day 1-3 days or unknown 4-6 days 7-9 days 10-15 days 16-20 days 20-30 days more than 30 days

^a Determining if the fatality rate at the project is 'likely to be a concern' may be informed by, for example, project-specific thresholds and/or relevant thresholds defined by national and/or regional stakeholders

^{b.} Note that the DST uses this information to inform an assumption about average searcher efficiency that may be achieved at the WEF.

^c The vicinity is defined as an area (within the same habitat type) that has a similar fatality risk profile and scavenging populations to that of the project site.

DST Output — Subsequent Years Design 4.1.3.2

Once the LE has provided answers to the 14 questions in the User Interface — Subsequent Years tab, the DST automatically generates a recommended design and calculates a DPI value.

These outputs are located to the right of the question panel, and the LE can use them to make decisions about different study designs. The DPI is calculated like a detection probability (see Section 3.2.4 for how detection probability is calculated) and has been devised as a tool for comparing and ranking study designs. Although it is expected that the DPI for a series of study designs will sort roughly in the same order as actual detection probabilities, the output should not be taken as a prediction of actual detection probability.³⁶

The DPI can be used to compare alternative designs in the User Interface — Design Explorer tab of the DST (see Section 4.1.3.3).

³⁶ The detection probability that a study design achieves will depend on field conditions during the study that are impossible to predict in advance of the study, so DPI predictions from the DST must rely on many simplifying assumptions. For this reason, users should not be surprised if the estimated detection probability differs from the DPI.

4.1.3.3 The Design Explorer

The User Interface — Design Explorer tab within the DST allows the LE to compare a recommended study design generated in the User Interface — Subsequent Years tab with up to two alternative designs created in the Design Explorer tab. For example, the project may be concerned about occasional fatalities of a globally scarce species and wish to sample more turbines than the recommended design sample size. The Alternative Design 1 and 2 columns in the Design Explorer tab allow the LE to test the effect of different sample sizes on DPI values and then allows PCFM components for each design to be modified to arrive at a valid design that provides the best balance between monitoring objectives and available resources.

To use the Design Explorer, the LE would:

User Interface-Design Explorer

Open the **User Interface** — **Design Explorer tab.** (The Recommended Design output generated in the User Interface — Subsequent Years tab is replicated and displayed on the left side of the worksheet.) To generate alternative designs, enter the following PCFM design component values into the Alternative Design 1 and Alternative Design 2 columns (Table 4.4).

Table 4.4 DST Design Explorer: Input table

DESIGN COMPONENT	VALUES
1. Number of turbines to sample	Enter sample equal to or less than the number of turbines at the WEF
2. Plot type	Select RAP or circular full plot from dropdown list
3. Search radius (m)	Select a search radius greater than 5 m (the DST will impose an upper bound based on the blade tip height; usually it is approximately 200 m)
4. Transect width (m)	Enter a value between 1 and 20 m
5. Search interval (m)	Select value between 1 and 41 days from dropdown list

Once the values are entered for alternative designs 1 and 2, the Design Explorer automatically calculates an overall value and size class specific DPI values for each alternative design.

The overall and size class specific DPI values for the recommended (Subsequent Years) design and alternative designs 1 and 2 are presented side by side so that they can be easily compared.

Once a study design is decided on, the LE will need to consider how to record and organize the data that the PCFM fieldwork will generate so that it is compatible with use in GenEst. At this point reading the step-by-step worked example provided in Appendix C and using the freely available GenEst User Guide (Simonis et al. 2018) is recommended to understand how the GenEst fatality estimator handles data.

4.2 WEF-ASSOCIATED OHL DESIGNS-RECOMMENDED PRACTICES

Section 3.4 of this Handbook discussed the components and principles of PCFM design for WEFassociated OHLs; this section discusses recommended practices for designing a PCFM program for this infrastructure and builds on the OHL design components introduced in Table 3.1. Although no DST is available for WEF-associated OHLs, this section provides a reasonable approach for designing a PCFM study. The guidance provided here applies to all OHLs regardless of horizontal length, tower or pole and line height, and voltage.

To ensure that all relevant underlying principles related to components of the PCFM study design have been considered, reviewing Section 3.3 of this Handbook is recommended when developing WEF-associated OHL designs. Box 4.1 provides further guidance on PCFM when electrocution is a primary risk.

4.2.1 OHL Sample

In some cases, it will not be necessary or feasible to include the entire length of OHLs in the PCFM study, and a sampling approach will be necessary. Suggested sampling approaches for various classes of OHL are described in the sections below.

Although sampling can produce accurate estimates of fatalities, **estimates based on a sample describe only the areas that were considered for the sample**. For example, if 15 km of a 20 km OHL is in areas that are considered low risk for birds, and PCFM sampling was restricted to 5 km within higher risk areas, the resulting estimates could be used only to make inference about the 5 km that was considered for sampling. Attempting to make inference about the entire 20 km OHL would not be appropriate because none of the low-risk areas would be represented in the data. These limitations on the kind of inference that is possible when sampling should be considered when designing a study and included when reporting the results.

4.2.1.1 Collector System within a WEF

To align with the World Bank Group Environmental, Health, and Safety Guidelines for Wind Energy, the collector system power lines within a WEF should be buried because above-ground collector lines present high collision and electrocution risks. Where above-ground collector lines are unavoidable, such as for geotechnical or permitting reasons, they should be designed to minimize bird electrocution risk by adequately separating energized and grounded components and using bird-safe pylon designs (for guidance, see Martín Martín et al. 2021 and RPS 2021). They should also be designed to reduce collision risk by routing them to avoid areas of highest risk and deploying mitigation measures (e.g., fitting lines with bird diverters) during construction. PCFM should be carried out over the entirety of any above-ground collector lines or a representative portion of them. Box 4.1 provides additional advice for developing PCFM designs when electrocution is a principal risk.


4.2.1.2 Interconnection Line

WEF infrastructure includes a high-voltage interconnection line (a transmission line, usually overhead) that runs between the WEF and the grid. The developer may build and own this line (or part of it), but ownership is often transferred to the utility for operation and maintenance. Whether the developer or the utility owns it, PCFM is recommended. The sampling approach for this type of OHL may be different where these structures cross high-risk habitats with likely occurrences of PCFM target species *susceptible to collision or electrocution with OHLs* than in areas where this information is unknown. It also differs based on the length of the line, which could be extensive. The following scenarios are envisioned and should be implemented under the guidance of a statistician or ecologist with OHL sampling experience.

Scenario 1: When the interconnection distance between the WEF substation and the grid substation is short (<5 km), PCFM should be conducted along its whole length, with the search corridor divided into standard length sample search plots (e.g., 500 m).³⁷

Scenario 2: When the length of the interconnection line makes it unfeasible to survey its full length (>5 km), the following approaches should be considered.

- When OHLs cross known high-risk habitats (e.g., wetlands, water bodies, areas of known habitat for large bats) or managed areas (e.g., legally protected areas,³⁸ internationally recognized areas³⁹) with likely presence of PCFM target species that are susceptible to collision or electrocution with OHLs:
 - a) The section(s) of OHL that traverse these areas (high-risk OHL sections) should be delimited.
 - b) The delimited section(s) of the high-risk OHL sections should be divided into segments of equal length (1 km is recommended).
 - c) PCFM should be conducted along a standard proportion of each segment (these will be the sample search plots; see Section 4.2.2 for how to determine the width of sample search plots). A recommended benchmark is that sample search plots cover 50 percent of each segment (e.g., a 500 m sample search plot per 1-km segment).

The length of the segments and proportion within each segment to survey should be sufficient to provide a representative sample of fatalities occurring along the section of OHL. This proportion is to be determined on a project-by-project basis based on the length of the overall interconnection line and accessibility. The survey proportion in each segment should be consistent throughout.

2. When OHLs cross areas where collision or electrocution risk is *unknown* in terms of habitat type or presence of PCFM target species susceptible to collision or electrocution with OHLs:

³⁷ The term "sample search plot" refers to delimited sample areas where PCFM will be conducted along an OHL. The location and length of each plot are determined using the methods described in this section. The width of sample search plots is determined using the method described in Section 4.2.2.

³⁸ As defined in IFC Performance Standard 6, footnote 16 "This Performance Standard recognizes legally protected areas that meet the IUCN definition: 'A clearly defined geographical space, recognized, dedicated and managed, through legal or other effective means, to achieve the long-term conservation of nature with associated ecosystem services and cultural values.' For the purposes of this Performance Standard, this includes areas proposed by governments for such designation."

³⁹ As defined in IFC Performance Standard 6, footnote 17 "UNESCO Natural World Heritage Sites, UNESCO Man and the Biosphere Reserves, Key Biodiversity Areas, and wetlands designated under the Convention on Wetlands of International Importance (the Ramsar Convention)." Important Bird and Biodiversity Areas (IBAs) are often co-located with Key Biodiversity Areas (KBAs). If they are not, they should also be considered.

- a) The OHL should be divided into segments of equal length (5 km is recommended).
- b) PCFM should be conducted along a standard proportion of each segment (these will be the sample search plots; see Section 4.2.2 for how to determine the width of sample search plots). A recommended benchmark is that a sample search plot will cover 20 to 50 percent of each segment (e.g., a 1- to 2.5-km sample search plot per 5-km segment) to be determined on a project-by-project basis.

4.2.1.3 OHLs Not Associated with WEFs

Transmission and distribution lines not directly associated with a WEF may extend regionally over long distances across multiple landscapes and present varied risks to species susceptible to collision and electrocution. The approach to deciding the OHL sample should be determined on a projectby-project basis based on the size of the line, the extent of high-risk OHL sections, accessibility, and resources. A statistician should be consulted to ensure an appropriate sampling design.

4.2.2 OHL Search Plot Shape and Size

The WEF-associated OHL search area is assumed to be a corridor along the centerline of the powerline. The sample search plot perimeter is defined by the length of the plot within each OHL segment (see Section 4.2.1.2) and the estimated limit of the fatality fall zone which forms the outer edges of the corridor. The recommended practice below assumes that the fall zone width from the OHL increases with height:

- For OHLs less than 25 m high, the search plot should be four times as wide as the maximum height of the OHL (e.g., for a 10 m high OHL, the searcher would search up to a distance of 20 m on either side of the OHL centerline, and for a 20 m high OHL, the searcher would search up to a distance of 40 m on either side of the OHL centerline).
- **For OHLs more than 25 m high**, the outer edge of the search plot should be 50 m on either side of the OHL centerline.



WEF-associated OHL transects should ideally be parallel to the OHL centerline, 4° with as many transects as necessary to achieve acceptable searcher efficiency rates for the species and size class(es) of interest (see Section 3.3.7.2 for more details on the factors effecting searcher efficiency rates).





⁴⁰ A zig-zag search route has been referenced in the literature (e.g., Gómez-Catasús et al. 2020; Lazo et. al. 2016; Martín Martín et. al 2022), but a parallel route is recommended for consistency in terms of the unsearched area. The zig-zag pattern could create a gap in visibility in the middle of each triangle.

OHL sample search plots may require search teams to walk transects⁴¹ that extend over 1 km or more and end a substantial distance away from their starting point (see Appendix B for advice on efficient search strategies along OHLs).

4.2.4 OHL Search Interval

Large birds are likely to be most at risk from collision and electrocution at WEF-associated OHLs, and PCFM will usually be designed to estimate fatality rates for these species. Based on this, the following approach is recommended:

- Set the initial search interval to two weeks (14 days) unless there are data to justify a different search interval.⁴²
- After the start of a PCFM program, use carcass persistence results from each year of monitoring to check and, if necessary, adjust the search interval to optimize its effectiveness within the overall study design (see Section 3.3.6 for more details on the relationship between search interval and carcass persistence).

4.2.5 OHL Bias Correction

4.2.5.1 Unsearched and Unsearchable Area

Because of the potential length of WEF-associated OHLs, sample search plots may cross a wider range of habitats and ground cover than PCFM for turbines, and as a result, unsearchable areas may be more likely to change throughout the study year. In these cases, maps of unsearchable areas may need to be created seasonally.

To assess unsearched and unsearchable areas at a WEF-associated OHL:

- Map unsearchable areas at each sample search plot before the start of PCFM fieldwork and revise maps if unsearchable areas change (e.g., because of seasonal changes in ground cover).
- Use unsearched area information in conjunction with location of fatalities within the searched area to calculate the DWP⁴³ (see Section 3.3.7.1 for more information on the unsearched area, Box 3.4 for further details on DWP, and Appendix D for the DWP tool used to calculate OHL DWP).





^{4*} In certain circumstances, such as when the OHL line is long, visibility is high, or there are large carcasses with high carcass persistence, use of vehicles to conduct PCFM may be an option.

⁴² The initial search interval may be reduced if documented evidence of fatalities from an existing OHL located in a similar habitat and sufficiently close to the WEF-associated OHL indicates that smaller species are likely to be PCFM target species.

⁴³ If unsearched areas change seasonally, DWP and fatality estimates should be calculated separately for each season.



To provide searcher efficiency information suitable for fatality rate estimates at WEF-associated OHLs:

- Place 12 searcher efficiency trial carcasses or decoys per size class per season across the range of ground cover visibility classes within the search plots.⁴⁴
- Place no more than one trial carcass per 100 m at any given time (see Section 3.3.7.2 for more detail on the underlying principles and Appendix B for practical advice about conducting searcher efficiency trials).

4.2.5.3 Carcass Persistence Trials

To provide carcass persistence information suitable for fatality rate estimates at WEF-associated OHLs:

- > Place 12 persistence trial carcasses per size class per season across the range of ground cover visibility classes present within the search plots.
- Place no more than one trial carcass per 100 m at any given time (see Section 3.3.7.3 for more detail on the underlying principles and Appendix B for practical advice about conducting searcher carcass persistence trials).

Box 4.1 Optimizing Recommended Practices for Electrocution Risk



OHL electrocution occurs when a bird or bat makes simultaneous contact with two different electrified parts (conductors) or with an electrified part and a grounded part such as a crossarm. Electrocution typically occurs when large birds, especially raptors, perch or nest on pylons or poles of medium-voltage distribution lines, and carcasses are therefore found below these structures. (Biasotto et al. 2022; Dwyer, Harness, and Donohue 2014; Lehman, Kennedy, and Savidge 2007; Loss, Will, and Marra 2014). Large bats, such as fruit bats, are electrocuted as they hang from distribution line wires, and carcasses are therefore likely to be found below these (e.g., Tella et al. 2020). Electrocuted birds and bats may not always fall to the ground and may remain suspended from wires, at the top of poles, or within the pylon structure.

⁴⁴ At a WEF where there are insufficient trial carcasses to conduct dedicated carcass persistence trials at a WEF-associated OHL, the LE may use results from trials conducted to test carcass persistence at the WEF turbines.

When possible, electrocution risks should have been designed out of the project. When this is not the case (if there are existing above-ground collector lines associated with the WEF; see Section 4.2.1.1), and electrocution is identified as a potential risk, the overall OHL PCFM design should take account of the following:

- > OHL sample: The OHL PCFM design should ensure that sample search plots include areas beneath poles and pylons, as well as directly beneath OHL wires, to ensure that locations where electrocuted birds and bats may fall are searched.
- OHL searches: Searchers should check for bird and bat fatalities on the crossarms and around the top of pylons and poles and for bat fatalities hanging from wires. When found, these fatalities should be included in the sample used to derive corrected fatality estimates for the WEF-associated OHL. Use of binoculars is recommended to help identify fatalities.
- > Searcher efficiency and carcass persistence trials: As part of the WEF-associated OHL PCFM bias trials, at least one large bird and one large bat trial^a per season should be placed in areas where electrocuted fatalities may fall. For carcass persistence, trial carcasses should have equivalent persistence times to electrocuted fatalities.

^a Bias trials to test searcher efficiency and carcass persistence for electrocuted fatalities are conducted only for large birds and bats because smaller size classes are unlikely to be electrocuted.



4.3 DATA COLLECTION FORMS

To ensure robustness of field data collection, a series of easy-to-print data collection forms is included with this Handbook in Appendices E to J. The data collection forms are designed to be used for turbines and WEF-associated OHLs (see Appendix B for guidance on the use of these forms as part of PCFM fieldwork).

- Appendix E PCFM Fatality Record Form: Used to record details of each fatality found during systematic searches or as chance finds (see Appendix B for how to conduct fatality searches at WEFs and Section 3.5 for information on chance finds). Each fatality found requires a separate PCFM Fatality Record Form. In addition to fatalities, this form should be used to record injured birds and bats found alive.
- > Appendix F PCFM Search Summary Form: Each searcher should have a copy of this form and complete it as they work. At the end of each search visit⁴⁵ the LE should review all search summary forms and ensure that each fatality record (Appendix E) and searcher efficiency trial corresponds to information provided on the search summary form and that search summaries account for all turbines in the study plan.

⁴⁵ For example, if study design is to search 20 turbines every seven days, the first search of all 20 turbines in the first seven days will be search visit 1, the second search of these same 20 turbines in the following seven days will be search visit 2, etc. Similarly, with WEF-associated OHLs, if the study design is to search ten 5-km segments per month, the first search of plots in these 10 segments during the first month will be search visit 1, the second search of the same plots in these 10 segments in the second month will be search visit 2, etc.

- Appendix G PCFM Searcher Efficiency Trial Form: Used to record details on each searcher efficiency trial (see Section 3.3.7.2 for the theory behind searcher efficiency trials, Appendix B for how to conduct searcher efficiency field trials at WEFs, and Section 4.2.5.2 for details about designing OHL searcher efficiency trials). Each trial carcass, surrogate, and decoy requires a separate PCFM Searcher Efficiency Trial Form. The trial administrator will complete this form, not the searcher.
- Appendix H PCFM Carcass Persistence Trial Form: Used to record details of each carcass persistence trial (see Section 3.3.7.3 for the theory behind carcass persistence trials, Appendix B for how to conduct carcass persistence trials in the field at WEFs, and Section 4.2.5.3 for details about designing OHL carcass persistence trials). Each trial carcass requires a separate PCFM Carcass Persistence Trial Form. The person who places the trial and the person(s) checking the carcasses on the required check days will use the same form.
- > Appendix I PCFM Carcass Freezer Tag: Each carcass should have its own unique label or tag before being placed in the freezer so that it can be cross-referenced with the relevant PCFM Fatality Record Form (Appendix E).
- Appendix J PCFM Freezer Contents Log: Used to record the contents of the dedicated carcass freezer and check type and number of carcasses available when planning searcher efficiency and carcass persistence trials. Ideally, the log should be kept in a plastic sleeve or binder and attached to or located near the freezer. Every time a new carcass is placed in the freezer, the log should be filled in. The log should be filled in only once for each carcass (e.g., if a carcass is removed from the freezer for a searcher efficiency trial and returned if retrieved, it will not be entered into the log a second time, but if its condition has changed this should be noted on the log sheet).



Freezer stored fatalities. Photo: Inkululeko Wildlife Services

DATA ANALYSIS

5. DATA ANALYSIS

Much of the focus of this Handbook has been on the design and implementation of PCFM studies in the field, and many of the elements of the study design are a response to a fundamental problem in bird and bat fatality estimation at WEFs: that the number of fatalities discovered during searching *is only a proportion of the number of fatalities* that the WEF generates. Generating robust estimates of total bird and bat fatalities at a WEF goes beyond fieldwork; it is necessary to analyze those data using fatality estimation tools and present the technical results of the PCFM analysis so that others can understand how results were derived. Section 5 focuses on this and introduces the analysis of spatio-temporal fatality patterns, which also should be considered.

5.1 FATALITY RATE ESTIMATION

To evaluate results generated using fatality rate estimation appropriately, a good conceptual understanding of two core statistical principles—precision and accuracy—is necessary. Understanding these concepts will help the LE decide whether the PCFM design that generated the fatality estimates was sufficient to characterize wildlife collision impacts adequately and inform management. Understanding of precision and accuracy will also provide insights into how PCFM and associated bias correction experiments could be adjusted to improve the value of future estimates. In addition to describing these two concepts, this section provides background on GenEst, the current state-of-the-art estimator referred to in Section 2.3. Interested readers are referred to James and colleagues (2017) for an introduction to statistical concepts and practice and to the references in Table 5.1 for further information on fatality estimation.

5.1.1 Precision

Precision is the ability to specify an estimate within a narrow range of values.

If we were 90 percent confident that a WEF had a fatality rate of 13.1 to 19.9 bats per megawatt per year, it is likely that the management response would be quite different than with a fatality rate of 2.3 to 30.7 bats per megawatt per year.⁴⁶ In both cases, the best estimate of the fatality rate is 16.5 per megawatt per year, but in the first case, we are 90 percent confident that the true value is between 13 and 20 bats per megawatt per year. Whereas in the second case, the true value could be as high as 31 bats per megawatt per year.

These two hypothetical examples differ only in their precision. In the former case, there is higher precision and hence less uncertainty and better information on which to act. Precision of fatality rate estimates is usually expressed as a confidence interval (CI) around an estimate (e.g., the median). The most important features influencing the precision of fatality rate estimates at WEFs are sample size, detection probability, and natural variability in fatality rates over time and space. Increasing *the level of sampling by increasing plot size, reducing transect width or conducting more frequent searches* will increase the precision of fatality rates stimates is limited, but in general, more intensive sampling efforts will increase the level of precision compared with less intensive sampling efforts.

⁴⁶ See Section 5.1.5.4 for instructions on how to derive this metric.

5.1.2 Accuracy

Accuracy is the degree to which an estimate matches the truth.

The accuracy of a fatality estimate is a measure of how close it is to the truth, or the true fatality rate at the WEF. Continuing with the hypothetical example above, in which the bat fatality was estimated at 16.5 bats per megawatt per year, what if there was another analysis of the same data, using a different statistical estimator, in which the mean estimated value of total bat fatality was 25 bats per megawatt per year? Whichever of these estimates is closer to the true number of fatalities would be the more accurate estimate, regardless of the precision of either estimate, but how would you know which is more accurate? How would you know which is closer to the true fatality rate? You would not, but if you understand the factors that push the estimate away from this value you can take steps to "correct" the estimates from the influence of these factors and reach a more accurate estimate.

Sources of bias (see Section 3.2.3) are factors that can push an estimate away from the truth, decreasing accuracy. In PCFM studies, the best-known sources of bias, or at least those believed to exert the greatest influence on the accuracy of total fatality rate estimates, are carcasses that fall outside of search plots (see Section 3.3.7.1), searcher efficiency (through carcass detectability; see Section 3.3.7.2), and carcass removal (e.g., through scavenging; see Section 3.3.7.3). All three of these factors decrease raw carcass counts below the true value of project-generated fatality (thereby underestimating true fatality rates). These sources of bias are why it is important to use field and analytical methods associated with bias correction to estimate detection probability accurately so that estimates of total, project-generated fatality rates are not biased.

5.1.3 Relationship Between Precision, Accuracy, and Study Design

The concepts of precision and accuracy in fatality rate estimation in WEF PCFM studies can be understood by considering the relationship of the fatality rate estimates to the true fatality rate (Figure 5.1). Precision and accuracy are related to study design through detection probability (see Section 3.2.4). Higher detection probabilities lead to more precise estimates, and a correct estimate of the detection probability is essential to an accurate estimate of fatality.

Figure 5.1 Relationship between precision and accuracy in fatality rate estimation



Note: The orange dot in the center represents the "true" fatality rate. This is the "target" that one is trying to hit with the fatality rate estimates (black dots). The four panels show examples of estimates with different levels of precision and accuracy.

5.1.4 Fatality Estimation Tools

5.1.4.1 The GenEst Fatality Estimator

Methods for estimating total wildlife fatality rates at WEFs have evolved considerably since the first models were developed and applied for this purpose in 2004. Most fatality estimators have used data on searcher efficiency, carcass persistence, area searched, and sampling fraction (fraction of turbines searched at a WEF) to estimate fatality rates. A simple equation summarizes the generic features of the common fatality estimators (Figure 5.2).





Although most early fatality estimators were based on solid statistical principles, including correcting for biases, there were differences between them in terms of data required, assumptions, and calculation methods. Not only did the choice of which estimator to use confuse users, the estimators also all produced slightly different results. Most of these tools also required users to have statistical programming capability or the ability to use tools such as R (R Development Core Team 2018).

To address the inconsistencies between estimators, Dalthorp and colleagues (2018) developed GenEst as a universal, user-friendly general estimator that would allow for consistency and facilitate comparison across PCFM studies. GenEst is the easiest to use and most flexible fatality estimation modeling framework available and is considered to be the current state-of-the-art (see Rabie et al. 2021). This Handbook is designed to be compatible with GenEst while recognizing the contributions of previous estimators. As its name suggests, GenEst represents a **Gen**eralization of several of those previous **Est**imators and owes much of its theoretical underpinning to them. Table 5.1 presents the common fatality estimators and their key assumptions, including GenEst.

ESTIMATOR	KEY ASSUMPTIONS AND FEATURES	CITATION
SHOENFELD	 Searchers have more than one opportunity to detect a carcass. Searcher efficiency does not decrease as carcasses age. Carcass persistence rate is not a function of carcass age. Assumes a single, average search interval to estimate carcass persistence. 	Shoenfeld 2004
HUSO	Searchers have a single opportunity to find a carcass.Requires estimates of time since death.	Huso 2011
CARCASS (R PACKAGE)	> Searchers have more than one opportunity to detect a carcass.> Searcher efficiency decreases as carcasses age.	Korner-Nievergelt et al. 2015
GENEST	 Searchers have more than one opportunity to detect a carcass. Searcher efficiency decreases as carcasses age. Generalized extension of Huso and Shoenfeld estimators. More flexible carcass persistence modeling than earlier estimators because each parameter can be modeled as a function of covariates. 	Dalthorp et al. 2018

GenEst was designed with an accessible, web-based graphical user interface that, unlike many other estimators, does not require working knowledge of statistical programming languages. GenEst is recommended in most circumstances. Section 5.1.4.2 describes the Evidence of Absence estimator (Dalthorp and Huso 2015; Dalthorp, Huso, and Dail 2017), which is preferable to GenEst under some circumstances, such as for rare species for which the number of carcasses detected is very low (fewer than five) or even zero. Box 5.1 describes how GenEst could be used for fatality estimation for OHLs. Appendix C demonstrates the use of the GenEst graphical user interface and further explains how GenEst generates fatality estimates.

Box 5.1 Fatality Estimation for OHLs Using GenEst



OHL fatality rates can be estimated using GenEst and requires the same input data tables used for turbines: search schedule, carcass observations, searcher efficiency trial data, carcass persistence trial data, and a density weighted proportion (DWP) of area searched table (see Appendix C: Preparing and Inputting Data into GenEst — The Five Data Files).

Data from OHL PCFM designs for each of these inputs are compatible with the required data entry format for GenEst. One minor difference is that, in the GenEst Search Schedule table, the sample unit is the segment rather than the turbine (see Section 4.2.1.2). Where OHLs are short (<5 km), and entire length will be searched (see Section 4.2.1.2, Scenario 1) it may be appropriate to subdivide the search corridor into standard length search plots and enter these as sample units in the GenEst Search Schedule table. DWP can be calculated for OHLs using the OHL DWP worksheet in the DWP Tool (Appendix D).

Fatality estimates (fatalities per kilometer) should be reported separately for each segment type identified in Section 4.2.1.2 (the interconnection line).



Small bird fatality. Malachite Sunbird (Nectarinia famosa). Photo: Inkululeko Wildlife Services

5.1.4.2

Evidence of Absence

When the goal is to estimate fatalities of rare species, other methods may be preferable to GenEst. Finding the carcass of a rare species is a rare event, meaning that the carcass count of rare species is often zero, and the estimate returned by GenEst will be zero. Huso and colleagues (2015) described this as "absence of evidence" rather than "evidence of absence" of fatalities of the species.

GenEst relies on dividing the number of carcasses by the probability of detection (see Figure 5.2). When detection probability is low, finding zero carcasses may mean that no individuals of the species collided with project infrastructure or that the search was not rigorous enough to detect those individuals. GenEst (and the other common estimators) will always return an estimate of zero if no carcasses are found because o/(detection probability) = o regardless of detection probability.

The Evidence of Absence estimator (EoA; Huso et al. 2015) overcomes the limitations of other estimators for rare species. EoA is a Bayesian model designed for fatality estimation of rare species. It requires all the same information as other estimators (bias trials, area correction, sampling fraction, carcass count), but the fatality estimate produced can be greater than zero even when no carcasses are found. The estimate from EoA can be interpreted as an upper bound on the number of carcasses that could plausibly have occurred, considering the search effort and carcass count. In that sense, EoA can be useful when WEFs have strict thresholds for individual species.

EoA is not treated in any more detail in this Handbook, except to say that it is a special estimator which should be used only when there is a need to put an upper bound on the mortality of individual species, and carcass counts are low (fewer than five individuals). When carcass counts are higher, GenEst is the superior estimator. Readers are referred to Huso and colleagues (2015) and Dalthorp and colleagues (2017) for more information.

5.1.5 Reporting of Data Analysis

Just as PCFM studies are not complete until the field data are analyzed to produce fatality estimates, fatality estimates are not useful unless the data analysis is reported in adequate detail. This section discusses reporting of the details of the PCFM design, the extent to which the design was implemented, raw fatality numbers, fatality estimates, and other associated output from the GenEst analysis—the topics covered so far in this Handbook. Further reporting in the context of *adaptive management*, which includes analysis of data to inform operational mitigation, is included in Section 6.

5.1.5.1 Reporting — PCFM Design Parameters

At a minimum the following six pieces of information should be included when reporting details of the PCFM design for each plot type used (RAP, full plot, OHL plot) and each season included in the study:

- > Search dates
- > Sampling fraction
- > Search plot type and size
- > Transect width, search interval
- > Number of bias trial carcasses of each type

5.1.5.2 Reporting — Success of PCFM Implementation

After reporting PCFM design parameters, the extent to which the design was implemented should be summarized. This section of the report should use simple tables to compare the planned fatality search effort with the actual search effort and the number of scheduled bias correction trials with the actual number of trials conducted. Reasons for any significant departures from the intended study design should also be explained. For example, if the search interval was twice weekly, but some turbines were unsearched for a period due to turbine maintenance, that should be noted in the report methods.

5.1.5.3 Reporting — Raw Fatality Data

At a minimum the following information should be included when summarizing the raw fatality data:

- > Details for each bird and bat carcass found, including species, date found, turbine or OHL location, age, sex (if known), and cause of and estimated time since death recorded in a table with one row per carcass.
- > Number of fatalities included in the fatality estimate, tabulated according to season, plot type, carcass size class, and any other relevant factors (e.g., habitat type, turbine size).

> Number of fatalities not included in the fatality estimate (non-collision and off-plot chance finds — see Section 3.5), tabulated according to PCFM target and non-target species and type (off plot, non-collision).

This information should be reported separately for the turbine search and the WEF-associated OHL search. It is often desirable to include additional data in appendices to a report, including raw data from bias trials and fatality searches. Including raw data in a report allows it to be easily accessed and used with data from other studies, for example, for the analysis of cumulative effects.

5.1.5.4 Reporting — Bias Corrected Fatality Estimates Using GenEst

The GenEst software will generate the analysis results (see Appendix C for more information), which can be downloaded from the software as tables or figures. GenEst is flexible in how it reports fatality estimates; estimates can be split according to size class, season, or any number of other categories that are present in the input data (see Appendix C). Results to be reported in the main body of a report should allow a reader to evaluate the sources of bias (see Section 3.2.3 and Section 3.3.7) and the total number of fatalities and fatality rate at the WEF, including the associated OHLs. Results should be presented so that the reader can understand whether there are important seasonal trends in fatalities. All estimates in this section should include the point estimate (the median value from GenEst), as well as the 5th and 95th quantiles of the estimates, which together make up a 90 percent CI (see Appendix C). These results include the following:

- > Annual and 6-monthly (or seasonal) fatality estimates **for each carcass size class** for turbines and for the WEF-associated OHL
- > Annual and 6-monthly (or seasonal) fatality estimates **for each PCFM target species** for turbines and for the WEF-associated OHLs
- Annual fatality estimates per megawatt for each size class for the entire WEF (including the WEF-associated OHL)⁴⁷
- > Carcass persistence and searcher efficiency estimates from GenEst output tables

GenEst produces tables and figures that can be downloaded. Corrected Akaike's Information Criterion(AICc)⁴⁸ scores (Burnham and Anderson 2022; see Appendix C Figures C15 and C18) from the carcass persistence and searcher efficiency models used in the analysis should be provided in the report text, and the entire model selection table should be included in an appendix to the report. Additional tables and figures are often useful to graphically illustrate how bias trial results or fatality estimates differ according to season or other factors. To keep the reporting concise, these figures and other relevant additional output from GenEst may be best referenced in the main body of the report and placed in an appendix.

⁴⁷ The annual fatality estimate per megawatt can be easily obtained by dividing the fatality estimates and CIs from GenEst by the megawatt capacity of the WEF. For example, a 50-MW project with an annual WEF median fatality rate of 100 bats (95% CI 30-200) would have a fatality rate of two bats per megawatt (95% CI 0.6-4).

⁴⁸ AICc is a metric that is used to determine the most appropriate model to fit to the data. The process of using AICc to choose a model is called "model selection."



African Straw-coloured Fruit-bats (Eidolon helvum). Photo: Inkululeko Wildlife Services

5.2 ANALYSIS OF SPATIO-TEMPORAL FATALITY PATTERNS

A valuable use of PCFM data is to identify spatial⁴⁹ and temporal⁵⁰ fatality patterns to inform adaptive management and mitigation actions. This involves identifying turbines or sections of OHLs with a high proportion of fatalities or identifying periods during the year when fatality rates peak. Reliably identifying problem turbines and OHL sections or high-risk periods requires PCFM data from several years. This is because many factors can contribute to a turbine or OHL section having a disproportionately high number of fatalities (e.g., topography, weather, species ecology, food availability). Multiple factors may drive interannual variation, and distinguishing spatial or temporal trends when many variables are involved requires considerable data.

In practice, during the early operational phase, PCFM data may indicate that turbines and OHL sections have high fatalities, but it will not be possible to reliably assess if this is likely to recur. This can present a challenge for the WEF because it may need to implement an adaptive management response well before sufficient data are available to make optimal decisions. This is particularly the case when safeguarding PCFM target species as the project may be required to uphold zero or very low thresholds to meet lender or regulatory requirements (see Section 6.2). The developer must therefore take an ongoing, dynamic approach to understanding and using spatial and temporal PCFM data to inform adaptive management (see Section 6). To make the best use of PCFM data in this context, the following approaches are suggested:

Report spatial and temporal fatality patterns annually and 6-monthly (or seasonally) to allow emerging patterns to be identified as the project lifecycle progresses (e.g., compare fatality patterns in the second year of operations with first year patterns).

⁴⁹ Spatial patterns can be assessed by obtaining per-turbine fatality estimates from GenEst and plotting them onto a map of the WEF with map symbols proportional to the size of fatality estimates.

⁵⁰ Temporal patterns are most easily assessed by examining bar charts showing counts of carcasses according to date or GenEst estimates according to week.

- > Compare spatial and temporal fatality patterns with other source information that may indicate high levels of bird or bat activity, for example:
 - Acoustic bat activity data
 - Bird monitoring data
 - Proximity of topographic features in the landscape such as slopes and ridges that may be disproportionately used by or risky to large soaring bird species
 - Proximity of habitat features used by species recorded as fatalities, such as trees, river valleys, lakes
 - Proximity to breeding and roost site locations and regularly used flight routes between these locations
 - Proximity to areas of prey aggregations or carrion for birds of prey or vultures
 - Occurrence of specific environmental conditions during periods of high recorded fatality (e.g., wind and other weather conditions)
- Review available flight activity data and other relevant literature from the area near the WEF or associated OHLs to further understand the predictability of fatality patterns recorded.
- > Consult with a statistician to help determine the consistency of spatial and temporal patterns recorded.

As described in Section 6, adaptive management and mitigation measures should be implemented and reviewed over relevant, specified time periods (e.g., after each migration or breeding season) to evaluate their effectiveness.



Habitat features may influence spatio-temporal fatality patterns at a WEF. Photo: Inkululeko Wildlife Services

ADAPTIVE MANAGEMENT

6. ADAPTIVE MANAGEMENT

As described in Section 1 of this Handbook, even with good quality pre-construction baseline data, impacts of WEFs on birds and bats are unpredictable, which is why a well-designed, systematic PCFM program is essential. But how should PCFM data be used to safeguard these species?

Adaptive management is the process of informing and updating the approach to biodiversity management by incorporating the results of monitoring or integrating new findings. As described in IFC's Guidance Note 6, although adaptive management may be described as a practical approach to managing uncertainty, it is not a trial-and-error process but rather a structured learning-by-doing process informed by monitoring.⁵¹

PCFM underpins a WEF's adaptive management program and, therefore, decision making with respect to ongoing mitigation and management of biodiversity throughout the project life cycle.⁵² The adaptive management program should be included as part of the project's ESIA and/or its management plans.⁵³ Although adaptive management is executed during operations, the types of management actions likely to be required should be considered during the project planning and design phases to ensure that they are adequately resourced.

This section identifies the components of the adaptive management program and provides information on the adaptive management review process; fatality thresholds, and other adaptive management triggers; management actions; and guidance on roles, responsibilities, reporting, and disclosure.

6.1 ADAPTIVE MANAGEMENT REVIEW

The LE should identify a timetable (frequency) with which to compare PCFM results with the pre-defined project commitments in the ESIA or management plans or review them with respect to government licensing and permitting conditions. If project commitments were not defined during the pre-construction phase, they should be included as part of the adaptive management program (see Section 6.3). The LE should also outline in advance the objectives of the adaptive management review (see Section 6.1.1).

6.1.1 Review Process

The overarching purpose of the adaptive management review process is to ensure that the project monitoring objectives are efficiently fulfilled following an evidence-based approach. The review process is designed to optimize the PCFM study design to identify and detect PCFM target species throughout the study duration and seek efficient use of financial and human resources allocated to safeguarding birds and bats.

⁵⁹ Adaptive management is also a key aspect of an effective Environmental and Social Management System (ESMS), which draws on the elements of "plan, do, check, and act" in accordance with established business management processes. IFC's Guidance Note 1, which accompanies its Performance Standard 1, captures the objective of adaptive management by stating that "effective management programs have an adaptive approach," and "The client should develop and implement procedures to...adapt actions and mitigations as appropriate based on the environmental and social monitoring data. This iterative process promotes flexible decision making that takes into consideration uncertainties, recognizes the importance of variability of the social and natural systems, and can be adjusted as outcomes from management actions, mitigations and other events become better understood" (Guidance Note 1, para. GN7O).

⁵² See Section 1.2 for the definition of mitigation in the context of the mitigation hierarchy.

⁵³ See paragraph GN₅0 in IFC's Guidance Note 6 (2019) for information on developing a Biodiversity Management Plan (BMP). The WEFs Adaptive Management Plan could be a component of the BMP or a standalone document.

Although the details of the review process will be tailored to the project-specific context and may differ depending on the stage of the operational phase, the underlying structure of the process is similar for all projects, following a sequential approach centered on fulfilling four key objectives as shown below. The review should be implemented based on a predetermined schedule, as well as in response to specific fatality events (see Section 6.1.2). The LE should periodically assess whether fatality thresholds continue to be valid and may also need to consider whether the effectiveness of existing mitigation measures should be formally evaluated (5 and 6 below). These points are elaborated here and in the decision-making process summarized in Figure 6.1.

ASSESSING PCFM RESULTS

1. Determine the adequacy of the PCFM design:

- > Are the PCFM estimates sufficiently precise to assess thresholds?
- Does the study design need to be modified to better characterize the risk to size class(es) of higher risk species (e.g., threatened, restricted-range, migratory, congregatory)?
- > Should the intensity of monitoring be increased or decreased in a certain season?

Optimizing the design is always appropriate, but especially after the first year(s) of monitoring.

2. Determine the adequacy of the PCFM target species list:

Considering the project's monitoring objectives, which species should be retained, included or eliminated as PCFM target species?

After implementation of the First Year PCFM design, the PCFM target species should be better understood in the context of the project's monitoring objectives. Evaluating the PCFM target species list is an iterative process and should be reviewed seasonally throughout the study duration.

ASSESSING IMPACTS

3. Determine the implications of the PCFM fatality estimates:

- > Do fatality estimates exceed pre-defined thresholds (see Section 6.2)?
- > If yes, by how much?
- > Do exceeded thresholds put the project in breach of regulatory or lender standards?

Fatality estimates should be evaluated against thresholds (see Section 6.2). It is important that uncertainty around estimates is also evaluated. If CIs are too wide, a more intensive PCFM study may be required to better determine whether thresholds have been exceeded.

MANAGING IMPACTS

4. Determine the adaptive response required:

When thresholds are exceeded or new findings emerge that alter the risk profile of the project, what mitigation actions should be taken or adjusted (see Section 6.3 below)?

ADDITIONAL ACTIONS

5. Determine the accuracy of thresholds:

As described in Section 6.2, determining numeric thresholds can be challenging given data limitations. If new data on species occurrences become available or if categorization of the IUCN Red List changes, the thresholds should be re-evaluated. Routine reassessment of thresholds should be factored into the review process.

6. Determine the effectiveness of mitigation:

If a mitigation measure (e.g., use of curtailment to minimize impact) was executed over the previous reporting period, was it effective in reducing fatalities?

One way to evaluate the effectiveness of mitigation, notably with respect to curtailment for bats, is to use an experimental design. The mitigation treatment (e.g., curtailment) would be applied to a randomly selected subset of turbines, and another subset would be left untreated to serve as a control group. That way, the treatment and control groups can be directly compared with one another.⁵⁴ Consulting with a statistician is recommended to ensure that the mitigation experiment is likely to produce data that can be used to increase understanding of mitigation effectiveness.⁵⁵

⁵⁴ Note that it may not be enough to compare pre-treatment and post-treatment fatality rates because year-to-year variation may interfere with the ability to determine the effectiveness of a treatment.

⁵⁵ For example, many types of studies can be executed by assigning turbines to mitigation treatments and control groups at random, but in some cases, particular turbines may be known to have especially high impacts, and mitigation treatments must be applied at high-impact turbines. It is possible to design statistically rigorous experiments in these cases, but doing so requires statistical expertise, which may go beyond the training of the LE.

Figure 6.1 Summary of the four key stages of the adaptive management process



6.1.2 Review Frequency

Adaptive management reviews are recommended as follows:

- > Routinely, corresponding to the timeframe for generating PCFM fatality estimates
- > As needed, in response to an unexpected high-risk incident, such as an exceedance of regulatory or lender standards

It is recommended that fatality estimates and PCFM reports are produced every six months or at least annually. If the study generates enough data to produce fatality estimates seasonally, and the risk profile of the project warrants greater scrutiny, or if constraints require that the design is changed seasonally, for example as a result of changes in unsearched area, PCFM reports should be generated seasonally. The DST should be consulted on the same schedule as part of the routine review of fatality estimates (see Section 6.4.2.1).

6.2 ADAPTIVE MANAGEMENT THRESHOLDS AND TRIGGERS

Thresholds are defined as limits beyond which project-related impacts become a concern because further loss of individuals is considered a risk to the long-term viability of a population. Defining numeric thresholds is an important component of an adaptive management program to assess the significance of impacts and determine appropriate management strategies. Regarding project finance in emerging market countries, thresholds have been defined with respect to an annual monitoring period (annual fatality thresholds; see, e.g., IFC 2017). Methods for setting thresholds for multiple years and for the lifetime of the project are available and, although not commonly used at projects in the context of financing in emerging market countries, may be more common in the future.⁵⁶

That said, a principal challenge is how to derive biologically meaningful numeric thresholds when the species-specific data required to do so are limited or absent, *which is often the case in emerging market countries*, especially for bat populations. Despite this challenge, and the inevitable compromises and assumptions that must be made, fatality thresholds can provide a valuable tool for tracking and managing impacts and determining alignment with lender standards (see Section 1.5).

This section provides an overview of, and guidance on developing thresholds and other adaptive management triggers in the emerging market context.

6.2.1 Determining Fatality Thresholds in an Emerging Market Context

As the approaches to threshold-setting may evolve quickly, readers are encouraged to scrutinize and apply the latest methods, assuming data availability, scientific robustness, and applicability to the wind energy sector. The current recommended approach to threshold-setting is provided below.

When there are national or regional fatality thresholds, such as the South African Bat Fatality Threshold Guidelines, developers should adhere to them (MacEwan et al. 2020). When national or regional guidance has not been developed, techniques to estimate threshold-setting include population matrix modeling, population viability analysis (PVA), and potential biological removal (PBR).

For species for which details on a wide range of biological and demographic parameters are available, PVA should be considered as an appropriate approach for modeling future population trends and informing judgment about future population viability. In some cases, especially for high conservation status species, a PVA may have been published that could be evaluated for application as a fatality threshold value. Importantly, PVA estimates have to be established for source populations which could be different from the species populations associated with the WEF, such as in the case of some migratory species (Voigt et al. 2012). In many emerging market countries, however, the information needed to conduct a PVA or other type of species viability analysis is often not available. In these situations, when information on population dynamics is limited, PBR offers a simple, robust calculation (see Dillingham and Fletcher 2011; Niel and Lebreton 2005; Wade 1998)

⁵⁶ For an example of methods used to establish long term thresholds see Dalthorp and Huso (2015). The EoA software also has a multiple years component that provides tools for estimating cumulative mortality across multiple monitoring years.



Potential for cumulative impacts may increase where PCFM target species occur in focal areas for wind energy. Photo: Alvaro Camiña

and one potential method to arrive at the threshold in the context of PCFM adaptive management, if used cautiously and informed by expert consultation.

6.2.2 Potential Biological Removal

PBR is a method for estimating the number of human derived mortalities that a wildlife population can sustain each year while allowing it to reach or maintain its optimum sustainable level. (see Wade 1998). PBR is calculated using a) the maximum annual recruitment rate (Rmax), b) a conservative estimate of the population size (Nmin), and c) a recovery factor (F) between 0.1 and 1.0, typically applied to allow for differences in a species' conservation status. The following sections explain the main components of the PBR formula. Section 6.2.2.4 then explains how to derive a threshold for a WEF informed by a PBR value and that takes account of non-project related human derived sources of mortality.

6.2.2.1 Minimum Population Size (Nmin)

Before the minimum population (Nmin) size can be estimated, the spatial extent of the population for each PCFM target species must be defined. Ideally, this 'spatial unit of analysis should be biogeographically determined, but this is often not possible, so the PCFM target species populations are more typically defined at other scales, such as nationally or with respect to a migratory flyway.

Assessing the population size for a selected spatial unit of analysis is also challenging because population size estimates are often approximate in emerging market countries and frequently derived from historical data. This is especially the case for bats, for which reliable estimates are often absent. Recognizing these constraints, a recommended approach is to use the information available and expert judgment through consultation and collaboration with relevant national and regional bird and bat experts to arrive at a reasoned qualitative population estimate. When population size information for the spatial unit of analysis is absent, one option is to use IUCN global or regional population estimates, calculate global PBR values, determine a global threshold, and use this as a guide to assess the threshold for the spatial unit of analysis. If IUCN population estimates are not available, another option is to derive PBR values using global IUCN Red List category population size criteria (IUCN 2012).⁵⁷

6.2.2.2 Maximum Annual Recruitment Rate (Rmax)

The maximum recruitment rate (Rmax) can be expressed as *the maximum annual population growth rate* – 1. With appropriate demographic information, matrix population models can be constructed to estimate the maximum annual population growth rate (Caswell 2001) and from this derive a maximum annual recruitment rate value for a species of interest. When there is insufficient information to calculate this value, it may be reasonable to use a surrogate Rmax value from a species or species group with similar demographic characteristics if these are available. An alternative approach is to use a formula developed by Niel and Lebreton (2005) that uses only adult survival rate and age at first breeding to derive a maximum recruitment rate. Users are referred to Neil and Lebreton (2005) and Dillingham and Fletcher (2008) for additional guidance for calculating PBR using these two parameters.

6.2.2.3 Recovery Factor (F)

The recovery factor (F) is a weighting between 0.1 and 1.0 that is typically used to account for variation in species conservation sensitivity. Following guidance in the published literature (Dillingham and Fletcher 2008; 2011; Niel and Lebreton 2005; Taylor et al. 2000; Wade 1998), it may be reasonable to set F values of 0.1 for Endangered or Vulnerable species, 0.3 for Near Threatened species, 0.5 for species of Least Concern that are declining or for which the population trend is uncertain; and 1.0 for species of Least Concern that are known to be increasing or stable. For Data Deficient species (which include nearly a fifth of all bat species) extinction risk is unknown, meaning that that an extinction risk category of Least Concern or Critically Endangered is equally plausible. For Data Deficient species it may therefore be reasonable to set a recovery factor of between 0.1 and 0.5 with the final value chosen based on the level of precaution warranted by stakeholder concerns and other evidence relevant to the conservation sensitivity of the population being evaluated. The recovery factor can be used as part of an adaptive management system and adjusted to reflect changing conservation status and increases in knowledge about a species during the lifecycle of a project.

6.2.2.4 From PBR Value to Threshold Through Expert Consultation

Establishing a PBR value for each PCFM target species population is the first step in the threshold-setting process, but it does not end there. Importantly, the PBR value provides a sustainable annual mortality limit for **all sources** of human derived mortality. This means that using the calculated PBR value to determine a threshold for an individual WEF would not be appropriate because it would result in a

⁹⁷ When using the IUCN Red List category and criteria document (IUCN 2012) be aware that a) different population size values are given in Criteria C and D for each category so the most appropriate value will need to be determined for the population being assessed, and b) the values given are for mature individuals of the species rather than all individuals. Because collision risk applies to all individuals, a correction would be needed to provide an estimate that includes immature individuals. One solution may be to use suitable surrogate species for which both mature and all individuals are reported in the IUCN Red List (www.redlist.org), determine the proportion attributed to non-mature individuals, and use this to calculate an estimate for all individuals of the species.

threshold that was too high. The population is likely to be exposed to non-project related sources of mortality such as, poisoning, hunting and mortality from other WEFs. To account for this, an additional step in the threshold-setting process is to estimate, for each PCFM species, how many of the annual mortalities calculated by the PBR were attributable to non-project related anthropogenic sources.

Given the likely lack of pre-existing quantitative data available, obtaining an estimate of annual mortality from non-project related anthropogenic sources will require consultation with recognized bird and bat experts from the region. It is recommended that this estimate, which may be qualitative, be derived systematically, such as through considering various categories of existing threats and the potential annual mortality that they cause. Once an estimate is agreed upon for each PCFM target species, experts should use this information along with the PBR value to derive a defensible threshold that will not compromise the long-term viability of the population within the spatial unit of analysis. The threshold value will always be lower than the PBR value, often substantially so.

Detailing the PBR-based threshold-setting approach is outside the scope of this Handbook, but the method is more fully outlined and implemented in the Tafila Region Wind Power Projects Cumulative Effects Assessment (IFC 2017), and the South Banat Wind Power Projects Rapid Cumulative Effects Assessment (IFC 2019). Readers are encouraged to review these and other sources.

When developing thresholds in this way, it is particularly important that data gaps and assumptions on which the thresholds are based are clearly documented so that they can be improved if better information becomes available.

Fatality threshold values will vary depending on the degree to which the assessed PCFM target species population can sustain fatalities and remain viable. It is likely that any additional humanderived mortality will compromise the long-term viability of threatened and restricted-range species. In this situation, a zero-fatality threshold may be appropriate, meaning that any fatality recorded will trigger an adaptive management response.⁵⁸

6.2.3 Other Risk Events

Adaptive management responses are not limited to cases in which numeric thresholds are exceeded. New information highlighting a potential change in the risk profile of the project or of a specific PCFM target species may also trigger management action. For example, development of an attractant near the WEF such as a slaughterhouse or a reservoir may increase risk for birds and bats that may warrant new forms of mitigation or other types of monitoring. These risk profile changes are context and location specific, and their relevance should be discussed as part of the adaptive management review process. They include the following:

- > A change in the IUCN global or national Red List category for a species may prompt a reevaluation of fatality thresholds or warrant a change to the PCFM design, with increased or decreased focus on the species.
- > Evidence of increased risk due to cumulative impacts may be caused by other WEFs sited in the vicinity and/or by other sources, such as increased hunting or poisoning. If unforeseen

⁵⁸ Note that chance finds of collision and electrocution fatalities outside the search plot (see Section 3.5.2) also should be considered when determining whether a threshold has been exceeded. This is especially relevant for PCFM target species with zero fatality thresholds because a single individual found as an off-plot discovery will exceed the threshold.

cumulative impacts arise during the operations phase, adaptive management responses may include re-evaluating fatality thresholds and collaborating with other WEFs to develop and participate in joint mitigation and inter-site adaptive management strategies.

- A change in the physical landscape near the WEF, for example a new landfill site, an increase in livestock, a change in livestock disposal practices, new water-related infrastructure, and new or modified drainage or irrigation systems may increase risk. These types of changes in the physical landscape could increase collision risk for specific or multiple PCFM target species and require an adaptive management response, such as implementing new or enhanced mitigation measures or engaging with external entities to develop strategies to modify human activities responsible for the increase in risk.
- > Near-miss incidents⁵⁹ are relevant when WEFs implement shutdown on-demand of turbines as a mitigation strategy (see Box 6.1). During shutdown on-demand, an example of a near-miss incident would be when a PCFM target species is observed flying through the rotor swept zone unharmed, and shutdown was initiated too late or not at all because of communication problems between bird observers and SCADA operators. Consistent, comprehensive recording of near-miss incidents is important for understanding underlying causes and informing an effective adaptive management response. In the example above, assessment of the near-miss incident records might lead to a change in bird observer or communication protocols as an adaptive management response. When using automated shutdown on-demand systems, false negatives should also be tracked to provide information on near-miss incidents.



Griffon Vulture (Gyps fulvus). Photo: Marian Ramos

⁵⁹ "Near miss" in this Handbook is similar to the concept as referred to in occupational health and safety practices but applied to birds and bats.

6.3 MANAGEMENT ACTIONS

The core of the adaptive management program may be defined as a series of actions that can be implemented to safeguard PCFM target species in response to an exceeded threshold or other type of risk event. Although site-level mitigation (e.g., curtailment) is the most obvious, other management responses may be warranted as a first step. These include investigation of underlying cause(s) of the fatalities, collaboration with external stakeholders (e.g., conservation organizations, government authorities, other developers), and development of joint management approaches over the landscape-scale.

Whatever adaptive management actions are chosen, they should be anticipated and defined, as much as possible, *in advance of operations*, preferably as part of the ESIA or management plans, rather than on an ad hoc basis during operations. Some WEF mitigation measures involve curtailment that may affect the project's energy output. Without advance consideration in the early stages of project development, this type of mitigation could take the developer by surprise. It is therefore recommended that, for all WEFs, *the project's energy yield assessment should precautionarily account for some level of curtailment for wildlife in case it is required during operations*. This practice is described in Hulka and Conzo (2021) and BID Invest and IFC (2019).⁶⁰ In some jurisdictions, turbine curtailment might be required; such is the case in certain districts in Germany (Fritze et al. 2019; KNE 2020).

Management actions should be site- and species-specific and informed by species ecology, spatial and temporal fatality, and activity patterns at the project site. Responses should also be proportional to exceeded thresholds and consider financial and potential regulatory implications. More than one mitigation option may be available to address the risk. As a general rule, the successive breaching of multiple thresholds should be matched by an increase in the measures to protect and promote the viability of PCFM target species.

When multiple WEFs are operating in the same geographical area and close to areas of high biodiversity value, developers are encouraged to collaborate to implement joint mitigation and monitoring measures. This will provide a wider understanding of impacts on PCFM target species and facilitate better informed adaptive management responses and may also result in cost savings.

Box 6.1 provides an illustrative list of mitigation measures to reduce collision risk associated with turbines and with collision and electrocution associated with WEF-associated OHLs. This is a topic extensively covered in other reference documents (e.g., Bennun et al. 2021; BID Invest and IFC 2019; Ledec, Rapp, and Aiello 2011; Rodrigues et al. 2015); readers are referred to these documents, which emphasize the importance of adhering to the mitigation hierarchy.

⁶⁰ See Box 6 (Cuadro 6) on page 62 of BID Invest and IFC (2019).

Box 6.1 Minimizing collision and electrocution risk through on-site mitigation



Turbine collision mitigation options — birds

- Shutdown on-demand: The most widely used method involves observers strategically located at vantage points around the WEF implementing shutdown of one or more turbines in response to birds approaching rotor blades. Turbines are restarted once observers determine that birds are no longer at risk (observer-led shutdown on-demand). Shutdowns of this type are typically short (<30 minutes). In some cases, radar is used to assist observers (radar-assisted shutdown on-demand). When WEFs are in areas of intense flight activity (e.g., bird migration corridors), shutdown protocols may allow for a larger proportion of the WEF turbines to be shut down for an extended period (several hours or more) when flight activity is observed or predicted to be particularly high. In addition to observer-initiated shutdown, some automated turbine shutdown systems focused on safeguarding larger bird species have been demonstrated to be effective and may be a good option in some circumstances. The most sophisticated of these systems combine imaging, artificial intelligence, and machine learning to detect target flying bird species and will automatically trigger a shutdown of turbines if a bird approaches within a threshold distance of turbine blades (see McClure et al. 2022).</p>
- > Livestock management measures are required when pastoralism is associated with the WEF site and when vultures or other large scavenging birds are associated with the area, because livestock have the potential to attract these birds, increasing collision risk. Management measures include collaborating with landowners, farmers, municipal authorities, and slaughterhouses to remove livestock carcasses from the WEF and its immediate surroundings; exclude livestock from certain areas during certain times of year; and ensure safe, rapid disposal of carcasses and other material that may attract birds.
- > **Turbine blade painting:** Increasing the visibility of rotor blades by painting one blade black may reduce turbine collisions for some bird species (Hodos 2003). Blade painting as an effective collision mitigation measure is currently based on a single field study (May et al. 2020), and further research is needed to evaluate its efficacy more fully. At the time of writing, this option is recommended only as a supplementary measure to use in addition to shutdown on-demand.

Turbine collision mitigation options — bats

- Blade feathering and curtailment is a type of collision minimization measure used mainly for bats that is achieved by slowing or stopping blade rotation (blade feathering) or raising the operational cut-in speed of the turbine (often referred to as curtailment). Curtailment prevents the turbine blades from rotating during periods of low wind speed and high bat activity. Curtailment may be applied to one or more turbines and may be constant for some fixed period of time (blanket curtailment) or vary according to environmental factors and bat activity (smart curtailment). Automated systems designed to curtail turbines in relation to environmental factors and bat activity that are linked to the WEF's (SCADA) systems have also been developed (Hayes et al. 2019).
- > Acoustic deterrents involve installation of devices on turbines to deter bats or birds from approaching by emitting noise (for bats, high-frequency sounds within the range of bat call frequencies).

Although acoustic deterrents may be effective for some bat species (e.g., Good et al. 2022; Weaver et al. 2020), evidence is limited, and deployment of this mitigation option is recommended only as a supplementary measure to feathering and curtailment.

OHL collision and electrocution mitigation options

- Bird-safe power poles can be used with medium-voltage distribution lines, which present a major electrocution hazard for birds, especially medium-sized and large birds, and large bats, such as fruit bats. When it is not possible to bury distribution lines because of geotechnical or other restrictions, bird-safe designs, such as ensuring adequate separation between energized components and between energized and grounded components such that a bird is not able to make simultaneous contact with these components, are essential. For more information, see APLIC (2012), Martín Martín and colleagues (2022), Prinsen and colleagues (2012), and RPS (2021).
- > **Bird flight diverters** are designed to increase power line visibility for birds, reducing collision risk. This risk is typically highest on transmission lines, although collisions also occur on distribution lines. A variety of suspended and fixed diverters are available, such as spheres, spirals, and flappers. A risk assessment that considers site conditions, species at risk, power line design, and long-term maintenance requirements, among other things, should be conducted to inform the type installed and their spacing and positioning along the lines. Some bird flight diverters (spirals) can entrap birds, which should also be considered in the selection. For more information, see APLIC (2012), Martín Martín and colleagues (2022), Prinsen and colleagues (2012), and RPS (2021).
- > Ultraviolet illumination of power lines involves projecting ultraviolet light along power lines to increase visibility and reduce collision risk. Although recent studies have demonstrated the effectiveness of



Multiple bird flight diverters installed on a transmission line. Algarve region, Portugal. Photo: Alvaro Camiña

this measure, especially for mitigating nocturnal collisions (see Baasch et al. 2022; Dwyer et al. 2019), further research is needed to confirm its effectiveness for a range of collision susceptible species and locations. At the time of writing, this option would be recommended as a supplementary measure to installing flight diverters.

6.4 ROLES, RESPONSIBILITIES, REPORTING, AND DISCLOSURE

The implementation of adaptive management should be understood at multiple levels of the wind energy company and is not limited to the LE and other teams managing environmental topics. The developer should establish who is responsible for communicating which types of information to whom—internally and externally. This section provides an overview of some of the main types of decisions that should be made at this stage of planning.

See Appendix B, "Implementing PCFM Fieldwork" for detailed information regarding responsibilities of the LE and search teams.

6.4.1 Roles and Responsibilities



A good adaptive management program should clearly state the responsibilities of the persons involved in, as well as those informing the adaptive management review process (see Section 6.1.1). Core roles and responsibilities involve the following persons:

- LE and PCFM search teams: Search teams should report PCFM data to the LE, who will check the quality of and collate this information weekly and monthly to inform routine and as-needed adaptive management review (see Section 6.1.2). In addition to monitoring fatalities, the LE and search teams should monitor for other types of risk events (see Section 6.2.3).
- > LE and environmental manager: PCFM is one component of the WEF's broader environmental and social management program. The LE should coordinate PCFM adaptive management activities with the WEF's environmental manager as part of the broader Environmental and Social Management System.
- Environmental manager and project manager: Because mitigation may impact energy yield, the environmental manager must keep the WEF project manager updated about developments so that they can keep the operations teams informed. The operations team may need to be involved in adaptive management review meetings so that any technical or financial implications of proposed measures are fully understood and evaluated. The environmental manager would also be responsible for communicating pertinent adaptive management measures to project lenders.
- Project manager and corporate board: It is critical that adaptive management measures are well-understood by the WEF project manager. For high-risk WEFs, the corporate board of directors should be aware of and in agreement with adaptive management measures, as informed by PCFM results.

6.4.2 Reporting and Disclosure

Reporting and disclosure are relevant for internal and external purposes. Section 5.1.5 discusses internal reporting with respect to the PCFM data analysis. This section builds on that reporting with respect to roles and responsibilities in the context of adaptive management and illustrates an approach that may be taken.

6.4.2.1 Internal Reporting



Other reports. The LE may need to provide reports in addition to 6-monthly reports to ensure that specific regulatory and lender requirements are satisfied. For example, it is recommended that the developer establish a reporting and rapid adaptive management response protocol for extreme events, such as when zero-fatality threshold species are recorded as fatalities or when multiple thresholds are exceeded.

6.4.2.2 External Reporting and Informing Strategic Planning

External reporting and disclosure of PCFM results will depend on the regulatory, lender, and stakeholder context. Developers are encouraged to make PCFM data available to relevant stakeholders, especially for projects in areas with high risks for bird or bat populations, where data-sharing can inform assessment of cumulative population impacts. A common data-sharing and reporting mechanism would facilitate this process, and developers are encouraged to partner with relevant stakeholders. Establishment of a biodiversity technical advisory committee could also inform the project's mitigation strategy through routine review of PCFM results and adaptive management decisions.

At the regional and national levels, a PCFM data-sharing mechanism provides the possibility of conducting meta-analyses to help understand cumulative and population-level impacts to inform future project siting and facilitate strategic planning for the wind energy sector. Wide use of PCFM data can play an important role in directing regional and national policy. Importantly, it can also inform the safeguarding of bird and bat populations given the likely expansion of wind energy in emerging market countries and the presence of a large number of globally threatened, collision susceptible species in these areas.



Wind Energy Facility. Valencia region, Spain. Photo: Alvaro Camiña

Réferences

REFERENCES

- Allison, T. D., J. E. Diffendorfer, E. F. Baerwald, J. A. Beston, D. Drake, A. M., Hale, C. D., Hein, M. M. Huso, S. R. Loss, J. E. Lovich, M. D. Strickland, K. A. Williams, and V. L. Winder. 2019. *Impacts to Wildlife of Wind Energy Siting and Operation in the United States*. Issues in Ecology. Washington, DC: The Ecological Society of America.
- APLIC (Avian Power Line Interaction Committee). 2012. "Reducing Avian Collisions with Power Lines: The State of the Art in 2012." Edison Electric Institute and APLIC, Washington, DC.
- Arnett, E. B., E. F. Baerwald, F. Mathews, L. Rodrigues, A. Rodríguez-Durán, J. Rydell, R. Villegas-Patraca, and C. C. Voigt. 2016. "Impacts of Wind Energy Development on Bats: A Global Perspective." *In Bats in the Anthropocene: Conservation of Bats in a Changing World*, edited by C. C. Voigt and T. Kingston, 295-323. Cham, Switzerland: Springer International Publishing.
- AWWI (American Wind Wildlife Institute). 2020a. "AWWI Technical Report, 2nd Edition: Summary of Bird Fatality Monitoring Data Contained in AWWIC." AWWI, Washington, DC.
- ———. 2020b. "AWWI Technical Report, 2nd Edition: Summary of Bat Fatality Monitoring Data Contained in AWWIC." www.awwi.org.
- Baasch, D. M., A. M. Hegg, J. F. Dwyer, A. J. Caven, W. E. Taddicken, C. A. Worley, A. H. Medaries, C. G. Wagner, P. G. Dunbar, and N. D. Mittman. 2022. "Mitigating Avian Collisions with Power Lines Through Illumination with Ultraviolet Light." Avian Conservation and Ecology 17.
- Bennett, E. 2015. "Observations from the Use of Dogs to Undertake Carcass Searches at Wind Facilities in Australia." In Wind and Wildlife, edited by C. Hull, E. Bennett, E. Stark, I. Smales, J. Lau, and M. Venosta, 113-123. Dordrecht: Springer. **https://doi.org/10.1007/978-94-017-9490-9_7**.
- Bennun, L., J. van Bochove, C. Ng, C. Fletcher, D. Wilson, N. Phair, and C. Carbone. 2021. *Mitigating Biodiversity Impacts Associated with Solar and Wind Energy Development*. Guidelines for Project Developers. Gland, Switzeraland: IUCN and Cambridge, UK: The Biodiversity Consultancy.
- Bernardino, J., K. Bevanger, R. Barrientos, J. F. Dwyer, A. T. Marques, R. C. Martins, J. M. Shaw, J. P. Silva, and F. Moreira. 2018. "Bird Collisions with Power Lines: State of the Art and Priority Areas for Research." *Biological Conservation* 222: 1-13. doi: 10.1016/j.biocon.2018.02.029.
- Bevanger, K. 1998. "Biological and Conservation Aspects of Bird Mortality Caused by Electricity Power Lines: A Review." Biological Conservation 86: 67-76.
- Biasotto, L. D., F. Moreira, G. A. Bencke, M. D'Amico, A. Kindel, and F. Ascensão. 2022. "Risk of Bird Electrocution in Power Lines: A Framework for Prioritizing Species and Areas for Conservation and Impact Mitigation." *Animal Conservation* 25 (2): 285-296. doi: 10.1111/acv.12736.
- BID Invest and IFC (International Finance Corporation). 2019. "Good Practice Guidelines for the Wind Energy Sector in Argentina: Management of Impacts on Birds and Bats [Guía de Buenas Prácticas para el Desarrollo Eólico en Argentina: Gestión de Impactos en Aves y Murciélagos]." Inter-American Investment Corporation and International Finance Corporation.
- BirdLife International. 2007. "Position Statement on Birds and Power Lines."
- -----. 2015. "Guidance on Appropriate Means of Impact Assessment of Electricity Power Grids on Migratory Soaring Birds in the Rift Valley/Red Sea Flyway. Migratory Soaring Birds Project, Regional Flyway Facility. Amman, Jordan."
- Brenninkmeijer, A., and E. Klop. 2017. "Bird Mortality in Two Dutch Wind Farms: Effects of Location, Spatial Design and Interaction with Powerlines." In *Wind Energy and Wildlife Interactions*, edited by J. Köppel, 99-116. Cham, Switzerland: Springer.
- Burnham, K. P., and D. R. Anderson. 2002. *Model Selection and Multimodel Inference: A Practical Information-Theoretic Approach, 2nd Edition.* New York: Springer.

- Cablk, M. E., and J. S. Heaton. 2006. "Accuracy and Reliability of Dogs in Surveying for Desert Tortoise (Gopherus agassizii)." Ecological Applications 16: 1926-1935.
- Caswell, H. 2001. Matrix Population Models: Construction, Analysis, and Interpretation. Sunderland, MA: Sinauer Associates.
- Choi, D. Y., T. W. Wittig, and B. M. Kluever. 2020. "An Evaluation of Bird and Bat Mortality at Wind Turbines in the Northeastern United States." PLoS One 15 (8): e0238034. doi: 10.1371/journal.pone.0238034.
- Chouhan, R., and S. Shrivastava. 2019. "Observation on Electrocution of Indian Flying Fox (Pteropus Giganteus) in Ramganjmandi, Kota, (Rajasthan) and Their Conservation Strategies." International Journal of Research in Engineering, Science and Management 2 (5): 2581-5792.
- Dalthorp, D., and M. Huso. 2015. "A Framework for Decision Points to Trigger Adaptive Management Actions in Long-Term Incidental Take Permits." U.S. Geological Survey Open-File Report 2015-1227.
- Dalthorp, D., M. M. P. Huso, and D. Dail. 2017. "Evidence of Absence (V2.0) Software User Guide." U.S. Geological Survey Data Series 1055. U.S. Geological Survey, Reston, VA. **https://pubs.usgs.gov/ds/1055/ds1055.pdf**.
- Dalthorp, D. H., L. Madsen, M. M. Huso, P. Rabie, R. Wolpert, J. Studyvin, J. Simonis, and J. M. Mintz. 2018. "GenEst Statistical Models—A Generalized Estimator of Mortality." U.S. Geological Survey Techniques and Methods, Volume 7, Chapter A2. https://pubs.usgs.gov/tm/7a2/tm7a2.pdf.
- Dalthorp, D., P. Rabie, M. Huso, and A. T. Tredennick. 2020. "Some Approaches to Accounting for Incidental Carcass Discoveries in Non-Monitored Years Using the Evidence of Absence Model." U.S. Geological Survey Open-File Report 2020-1027. https://pubs.er.usgs.gov/publication/ofr20201027.
- Dalthorp, D. H., M. M. Huso, M. Dalthorp, and J.M. Mintz. 2022. "Density-Weighted Proportion (DWP) Tool Suite: U.S. Geological Survey Software Release." https://doi.org/10.5066/P9QWO5lo.
- De La Zerda, S. and L. Rosselli. 2002. Mitigating Collision of Birds against Transmission Lines in Wetland Areas in Columbia, by Marking Ground Wire with Bird Flight Diverters (BFD). J. W. Goodrich-Mahoney, D. F. Mutrie, and C. A. Guild, eds. Pp. 395-402. In: The Seventh International Symposium on Environmental Concerns in Rightsof-Way Management, Calgary, Alberta, Canada. September 9-13, 2000. Elsevier, Oxford, United Kingdom.
- DeVault, T. L., T. W. Seamans, K. E. Linnell, D. W. Sparks, and J. C. Beasley. 2017. "Scavenger Removal of Bird Carcasses at Simulated Wind Turbines: Does Carcass Type Matter?" *Ecosphere* 8 (11): e01994.
- Dillingham, P.W., and D. Fletcher. 2008. "Estimating the Ability of Birds to Sustain Additional Human-Caused Mortalities Using a Simple Decision Rule and Allometric Relationships." *Biological Conservation* 141: 1783-1792.
- Dillingham, P.W., and D. Fletcher. 2011. "Potential Biological Removal of Albatrosses and Petrels with Minimal Demographic Information." *Biological Conservation* 144: 1885-1894.
- Dixon, A., M. Ming, A. Gunga, G. Purev-Ochir, and N. Batbayar. 2013. "The Problem of Raptor Electrocution in Asia: Case Studies from Mongolia and China." *Bird Conservation International* 23 (4): 520-529.
- Domínguez del Valle, J., F. Cervantes Peralta, and M. I. Jaquero Arjona. 2020. "Factors Affecting Carcass Detection at Wind Farms Using Dogs and Human Searchers." *Journal of Applied Ecology* 57: 1926-1935.
- Drewitt, A. L., and R. H. W. Langston. 2006. "Assessing the Impacts of Wind Farms on Birds" Ibis 148: 29-42.
- Dwyer, J. F., R. E. Harness, and K. Donohue. 2014. "Predictive Model of Avian Electrocution Risk on Overhead Power Lines." *Conservation Biology* 1228: 159–168.
- Dwyer, J. F., A. K. Pandey, L. A. McHale, and R. E. Harness. 2019. "Near-Ultraviolet Light Reduced Sandhill Crane Collisions with a Power Line by 98%." *Condor: Ornithological Applications* 121, 1-10.
- EBRD (European Bank for Reconstruction and Development). 2019. Performance Requirement 6: Biodiversity Conservation and Sustainable Management of Living Natural Resources. London: EBRD.

———. 2022 Guidance Note 6, Biodiversity Conservation and Sustainable Management of Living Natural Resources, EBRD.

- Ferrer, M., M. de Lucas, G. F. E. Janss, E. Casado, A. R. Muñoz, M. J. Bechard, and C. P. Calabuig. 2012. "Weak Relationship Between Risk Assessment Studies and Recorded Mortality in Wind Farms." *Journal of Applied Ecology* 49: 38-46.
- Frick, W. F., E. F. Baerwald, J. F. Pollock, R. M. R. Barclay, J. A. Szymanski, T. J. Weller, A. L. Russell, S. C. Loeb, R. A. Medellin, and L. P. McGuire. 2017. "Fatalities at Wind Turbines May Threaten Population Viability of a Migratory Bat." *Biological Conservation* 209: 172-177. doi: 10.1016/j.biocon.2017.02.023.
- Frick, W. F., Kingston, T., & Flanders, J. (2020). A review of the major threats and challenges to global bat conservation. Annals of the New York Academy of Sciences, 1469 (1), 5-25.
- Fritze, M., L. Lehnert, O. Heim, O. Lindecke, M. Roeleke, and C. Voigt. 2019. "Fledermausschutz im Schatten der Windenergie -Deutschlands Experten vermissen Transparenz und bundesweite Standards in den Genehmigungsverfahren." *Naturschutz und Landschaftsplanung* 51: 20-27.
- Gómez-Catasús, J., L. M. Carrascal, V. Moraleda, J. Colsa, F. Garcés, and C. Schuster. 2020. "Factors Affecting Differential Underestimates of Bird Collision Fatalities at Electric Lines: A Case Study in the Canary Islands." *Ardeola* 68 (1): 71-94.
- Good, R. E., G. Iskali, J. Lombardi, T. McDonald, K. Dubridge, M. Azeka, and A. Tredennick. 2022. "Curtailment and Acoustic Deterrents Reduce Bat Mortality at Wind Farms." *Journal of Wildlife Management* 86: e22244.
- Gove, B., R. H. W. Langston, A. McCluskie, J. D. Pullan and I. Scrase (2013). *Wind Farms and Birds: An Updated Analysis of the Effects of Wind Farms On Birds, and Best Practice Guidance on Integrated Planning And Impact Assessment*. Birdlife International on behalf of the Bern Convention.
- Hallingstad, E. C., P. A. Rabie, A. C. Telander, J. A. Roppe, and L. R. Nagy. 2018. "Developing an Efficient Protocol for Monitoring Eagle Fatalities at Wind Energy Facilities." *PLoS One* 13 (12): e0208700. doi: 10.1371/journal. pone.0208700.
- Hallingstad, E., D. Riser-Espinoza, S. Brown, P. Rabie, J. Haddock, and K. Kosciuch. 2023. "Game Bird Carcasses Are Less Persistent Than Raptor Carcasses, but Can Predict Raptor Persistence Dynamics." *PLoS One* 18 (1): e0279997. https://doi.org/10.1371/journal.pone.0279997.
- Hayes, M. A., L. A. Hooton, K. L. Gilland, C. Grandgent, R. L. Smith, S. R. Lindsay, J. D. Collins, S. M. Schumacher, P. A. Rabie, J. C. Gruver, and J. Goodrich-Mahoney. 2019. "A Smart Curtailment Approach for Reducing Bat Fatalities and Curtailment Time at Wind Energy Facilities." *Ecological Applications* 29: e01881.
- Hein, C. D., J. Gruver, and E. B. Arnett. 2013. *Relating Pre-Construction Bat Activity and Post-Construction Bat Fatality to Predict Risk at Wind Energy Facilities: A Synthesis. A Report Submitted to the National Renewable Energy Laboratory.* Austin, TX: Bat Conservation International.
- Hodos, W. 2003. Minimization of Motion Smear: Reducing Avian Collision with Wind Turbines; Period of Performance: July 12, 1999—August 31, 2002. Golden, CO: National Renewable Energy Laboratory.
- Hulka, S., and L. A. Conzo. 2021. "Recommendations for Integrating Biodiversity into the Wind Energy Sector in Emerging Market Countries." *Private Sector and Development* 35: 38-41.
- Hull, C. L., and S. Muir. 2010. "Search Areas for Monitoring Bird and Bat Carcasses at Wind Farms Using a Monte-Carlo Model." *Australasian Journal of Environmental Management* 17 (2): 77-87.
- Huso, M. M. P., and D. Dalthorp. 2014. "Accounting for Unsearched Areas in Estimating Wind Turbine-Caused Fatality." *Journal of Wildlife Management* 78 (2): 347-358. doi: 10.1002/jwmg.663.
- Huso, M. 2011. "An Estimator of Wildlife Fatality from Observed Carcasses." Environmetrics 22 (3): 318-329. doi: 10.1002/ env.1052.
- Huso, M., D. Dalthorp, and F. Korner-Nievergelt. 2017. Statistical Principles of Post-Construction Fatality Monitoring Design. Pp. 84-102. In: M. Perrow, ed. Wildlife and Wind Farms, Conflicts and Solutions. Vol. 2, Onshore: Monitoring and Mitigation. Pelagic Publishing, Exeter, United Kingdom
- Huso, M. M. P., D. Dalthorp, D. Dail, and L. Madsen. 2015. "Estimating Wind-Turbine-Caused Bird and Bat Fatality When Zero Carcasses Are Observed." *Ecological Applications* 25: 1213-1255. doi: 10.1890/14-0764.1.
- IFC (International Financial Corporation). 2012. Performance Standard 6. Biodiversity Conservation and Sustainable Management of Living Natural Resources. Washington, DC: IFC.
- ———. 2019. Guidance Note 6, Biodiversity Conservation and Sustainable Management of Living Natural Resources, Washington, DC, USA: IFC.
- ———. 2017. Tafila Region Wind Power Projects Cumulative Effects Assessment. International Finance Corporation. Washington, DC: IFC. <u>http://www.ifc.org/jordanwind</u>
- ———. 2019. South Banat Region Wind Power Projects. Rapid Cumulative Assessment. Washington, DC: IFC. See file "S Banat Region WEF RCIA Rev 6" under "Client Documentation" located in IFC's Disclosure Portal, <u>https://</u> disclosures.ifc.org/project-detail/ESRS/33839/dolovo-wind.
- IUCN (International Union for Conservation of Nature). 2012. IUCN Red List Categories and Criteria: Version 3.1. Gland, Switzerland: IUCN.
- James, G., D. Witten, T. Hastie, and R. Tibshirani. 2017. An Introduction to Statistical Learning: With Applications in R. New York: Springer.
- Jenkins, A. R., J. J. Smallie, and M. Diamond. 2010. "Avian Collisions with Power Lines: A Global Review of Causes and Mitigation with a South African Perspective." Bird Conservation International 20: 263-278.
- Katzner, T. E., Nelson, D. M., Diffendorfer, J. E., Duerr, A. E., Campbell, C. J., Leslie, D., Vander Zanden, H. B., Yee, J. L., Sur, M., Huso, M. M. P., Braham, M. A., Morrison, M. L., Loss, S. R., Poessel, S. A., Conkling, T. J., & Miller, T. A. (2019). Wind energy: An ecological challenge. Science, 366(6470), 1206-1207. <u>https://doi.org/10.1126/science.</u> aaz9989.
- KNE (Kompetenzzentrum Naturschutz und Energiewende). 2020. "Anfrage Nr. 279 zur Anzahl an Windenergieanlagen (Onshore) in Deutschland mit Abschaltungen zum Fledermausschutz."
- Korner-Nievergelt, F., I. Niermann, O. Behr, M. A. Etterson, R. Brinkmann, P. Korner, B. Hellriegel, T. Roth, M. M. P. Huso, and D. Dalthorp. 2015. "Mortality Estimation from Carcass Searches Using the R-Package Carcass—A Tutorial." *Wildlife Biology* 21 (1): 30-43.
- Lazo, A., A. Castelló, J.L. Conde, S. Braña, R. Otero, and G. Janss. 2016. *Metodología y Protocolos para la Recogida y Análisis de Datos de Siniestralidad de Aves por Colisión en Líneas de Transporte de Electricidad, Versión 2*. Sevilla, Spain: Red Eléctrica de España and Asistencias Técnicas Clave S.L.
- Ledec, G. C., K. W. Rapp, and R. G. Aiello. 2011. Greening the Wind: Environmental and Social Considerations for Wind Power Development. World Bank, Washington, DC.
- Lehman, R. N., P. L. Kennedy, and J. A. Savidge. 2007. "The State of the Art in Raptor Electrocution Research: A Global Review." *Biological Conservation* 136: 159–174.
- Loss, S. R., T. Will, and P. P. Marra. 2014. "Refining Estimates of Bird Collision and Electrocution Mortality at Power Lines in the United States." *PloS One* 9 (7): e101565.
- MacEwan, K., J. Aronson, E. Richardson, P. Taylor, B. Coverdale, D. Jacobs L. Leeuwner, W. Marais, and L. Richards, 2020. *South African Bat Fatality Threshold Guidelines*, 3rd Ed. South Africa: South African Bat Assessment Association.

- Marques, A. T., H. Batalha, S. Rodrigues, H. Costa, M. J. Ramos Pereira, C. Fonseca, M. Mascarenhas, and J. Bernardino. 2014. Understanding Bird Collisions at Wind Farms: An Updated Review on the Causes and Possible Mitigation Strategies. *Biological Conservation* 179: 40-52. doi: 10.1016/j.biocon.2014.08.017.
- Martín Martín, J., J. R. Garrido López, H. Clavero Sousa, and V. Barrios, editors. 2022. Wildlife and Power Lines. Guidelines for Preventing and Mitigating Wildlife Mortality Associated with Electricity Distribution Networks. Gland, Switzerland: IUCN.
- Mathews, F., M. Swindells, R. Goodhead, T. A. August, P. Hardman, D. M. Linton, and D. J. Hosken. 2013. "Effectiveness of Search Dogs Compared with Human Observers in Locating Bat Carcasses at Wind-Turbine Sites: A Blinded Randomized Trial." *Wildlife Society Bulletin* 37: 34-40.
- Maurer, J. D., M. Huso, D. Dalthorp, L. Madsen, and C. Fuentes. 2020. "Comparing Methods to Estimate the Proportion of Turbine-Induced Bird and Bat Mortality in the Search Area under a Road and Pad Search Protocol." *Environmental and Ecological Statistics* 27: 769-801. doi: 10.1007/s10651-020-00466-0.
- May, R., T. Nygård, U. Falkdalen, J. Åström, Ø. Hamre, and B. G. Stokke. 2020. "Paint It Black: Efficacy of Increased Wind Turbine Rotor Blade Visibility to Reduce Avian Fatalities." *Ecology and Evolution* 10: 8927-8935.
- McClure, C. J. W., B. W. Rolek, L. Dunn, J. D. McCabe, L. Martinson, and T. E. Katzner. 2022. "Confirmation That Eagle Fatalities Can Be Reduced by Automated Curtailment of Wind Turbines." *Ecological Solutions and Evidence* 3: e12173.
- Mogdans, J., J. Ostwald, and H.-U. Schnitzler. 1988. "The Role of Pinna Movement for the Localization of Vertical and Horizontal Wire Obstacles in the Greater Horseshoe Bat, (*Rhinolopus ferrumequinum*)." Journal of the Acoustical Society of America 84: 1676-1679.
- Murphy, R. K., et al. (2016). "Crippling and Nocturnal Biases in a Study of Sandhill Crane (*Grus canadensis*) Collisions with a Transmission Line." *Waterbirds* 39(3): 312-317, 316
- Niel, C., and J.-D. Lebreton. 2005. "Using Demographic Invariants to Detect Overharvested Bird Populations from Incomplete Data." *Conservation Biology* 19: 826-835.
- Paula, J., M. C. Leal, M. J. Silva, R. Mascarenhas, H. Costa, and M. Mascarenhas. 2011. "Dogs as a Tool to Improve Bird-Strike Mortality Estimates at Wind Farms." *Journal for Nature Conservation* 19: 202-208.
- Perrow M.R. (Ed.) 2017. Wildlife and Wind farms, Conflicts and Solutions. Volume 1 Onshore: Potential Effects. Pelagic Publishing, Exeter, UK
- Prinsen, H., G. Boere, N. Pires, and J. J. Smallie. 2011. Review of the Conflict Between Migratory Birds and Electricity Power Grids in the African-Eurasian Region. CMS/AEWA Technical Series. Bonn, Germany: United Nations Environment Program.
- Prinsen, H. A. M., J. J. Smallie, G. C. Boere, and N. Píres. (Compilers). 2012. "Guidelines on How to Avoid or Mitigate Impact of Electricity Power Grids on Migratory Birds in the African-Eurasian Region." AEWA Conservation Guidelines No. 14, CMS Technical Series No. 29, AEWA Technical Series No. 50, CMS Raptors MOU Technical Series No. 3. Bonn, Germany.
- R Development Core Team. 2018. "R: A Language and Environment for Statistical Computing." R Foundation for Statistical Computing, Vienna, Austria. <u>https://www.R-project.org/</u>.
- Rabie, P. A., D. Riser-Espinoza, J. Studyvin, D. Dalthorp, and M. Huso. 2021. "AWWI Technical Report: Performance of the GenEst Mortality Estimator Compared to the Huso and Shoenfeld Estimators". Renewable Energy Wildlife Institute, Washington, DC. <u>https://awwi.org/resources/genest/</u>.
- Rajeshkumar, S., C. Raghunathan, and K. Venkataraman. 2013. "Observation on Electrocution of Flying Fox (*Pteropus giganteus*) in Andaman Islands and Their Conservation Strategies." *Journal of the Andaman Science Association* 18 (2): 213-215.

- Rodrigues, L., L. Bach, M.-J. Dubourg-Savage, B. Karapandža, D. Kovač, T. Kervyn, J. Dekker, A. Kepel, P. Bach, J. Collins, C. Harbusch, K. Park, B. Micevski, and J. Minderman. 2015. "Guidelines for Consideration of Bats in Wind Farm Projects—Revision 2014". EUROBATS Publication Series No.6 (English version). UNEP/EUROBATS Secretariat, Bonn, Germany.
- Rioux, S., J.-P. L. Savard, and A. A. Gerick. 2013. "Avian Mortalities Due to Transmission Line Collisions: A Review of Current Estimates and Field Methods with an Emphasis on Applications to the Canadian Electric Network. *Avian Conservation and Ecology* 8: 7.
- RPS (Raptor Protection of Slovakia). 2021. "Electrocutions & Collisions of Birds in EU Countries: The Negative Impact & Best Practices for Mitigation." RPS, Bratislava, Slovakia.
- Rydell, J., R. Ottvall, S. Pattersson, and M. Green. 2017. "The Effects of Wind Power on Birds and Bats: An Updated Synthesis Report." Report No. 6791. Lund University, Lund, Sweden.
- Schmeller, D. S., M. Böhm, C. Arvanitidis, S. Barber-Meyer, N. Brummitt, M. Chandler, E. Chatzinikolaou, M. J.
 Costello, H. Ding, J. García-Moreno, M. Gill, P. Haase, M. Jones, R. Juillard, W. E. Magnusson, C. S. Martin, M. McGeoch, J.-B. Mihoub, N. Pettorelli, V. Proença, C. Peng, E. Regan, U. Schmiedel, J. P. Simaika, L.
 Weatherdon, C. Waterman, H. Xu, and J. Belnap. 2017. "Building Capacity in Biodiversity Monitoring at the Global Scale." *Biodiversity and Conservation* 26: 2765-2790.
- Schuster, E., L. Bulling, and J. Köppel. 2015. Consolidating the State of Knowledge: A Synoptical Review of Wind Energy's Wildlife Effects. *Environmental Management* 56(2): 300-331.
- Shaw, J. M., T. A. Reid, B. K. Gibbons, M. Pretorius, A. R. Jenkins, R. Visagie, M. D. Michael, and P. G. Ryan, 2021. "A Large-Scale Experiment Demonstrates That Line Marking Reduces Power Line Collision Mortality for Large Terrestrial Birds, but Not Bustards, in the Karoo, South Africa." *Ornithological Applications* 123: 1-10.
- Shoenfeld, P. 2004. "Suggestions Regarding Avian Mortality Extrapolation." Technical memo provided to FPL Energy. West Virginia Highlands Conservancy, Davis, WV.
- Simonis, J., D. Dalthorp, M. Huso, J. Mintz, L. Madsen, P. Rabie, and J. Studyvin. 2018. GenEst User Guide—Software for a Generalized Estimator of Mortality: U.S. Geological Survey Techniques and Methods, Book 7. U.S. Geological Survey. https://doi.org/10.3133/tm7C19.
- Skalski, J. R. and R. Townsend, 2017. 2014-2016 Evidence of Absence Model Parameter Research: 133-Turbine Wind Energy Facility, Illinois. Final Technical Report. Shoener Environmental, Pennsylvania. 45 pages
- Solick, D., Pham, D., Nasman, K., & Bay, K. (2020). Bat Activity Rates do not Predict Bat Fatality Rates at Wind Energy Facilities. Acta Chiropterologica, 22(1), 135-146.
- Studyvin, J., P. Rabie, D. Riser-Espinoza, and T. McDonald. 2020. "Windac: Area Correction Methods. R Package Version 1.2.4."
- Taylor, B. L., P. R. Wade, D. P. De Master, and J. Barlow. 2000. "Incorporating Uncertainty into Management Models for Marine Mammals." *Conservation Biology* 14: 1243-1252.
- Tella, J. L., D. Hernandez-Brito, G. Blanco, and F. Hiraldo. 2020. "Urban Sprawl, Food Subsidies and Power Lines: An Ecological Trap for Large Frugivorous Bats in Sri Lanka?" *Diversity* 12 (94). doi: 10.3390/d12030094.
- Thaxter, C. B., G. M. Buchanan, J. Car, S, H. M. Butchart, T. Newbold, R. E. Green, J. A. Tobias, W. B. Foden, S. O'Brien, and J. W. Pearce-Higgins. 2017. "Bird and Bat Species' Global Vulnerability to Collision Mortality at Wind Farms Revealed Through a Trait-Based Assessment." *Proceedings of the Royal Society B: Biological Sciences* 284: 20170829. <u>http://dx.doi.org/10.1098/rspb.2017.0829</u>.
- Travers, M. S., S. Driskill, A. Stemen, T. Geelhoed, D. Golden, S. Koike, A. A. Shipley, H. Moon, T. Anderson, M. Bache, and A. F. Raine. 2021. "Post-Collision Impacts, Crippling Bias, and Environmental Bias in a Study of Newell's Shearwater and Hawaiian Petrel Powerline Collisions." *Avian Conservation and Ecology* 16 (1): 15. <u>https://doi.org/10.5751/ACE-01841-16015</u>.

- Urquhart, B., S. Hulka, and K. Duffy. 2015. "Game Birds Do Not Surrogate for Raptors in Trials to Calibrate Observed Raptor Collision Fatalities." *Bird Study* 62 (4): 552-555. doi: 10.1080/00063657.2015.105375.
- Vanderelst, D., M. W. Holderied, and H. Peremans. 2015. "Sensorimotor Model of Obstacle Avoidance in Echolocating Bats." PLoS Computational Biology 11: e1004484.
- Voigt, C. C., A. G. Popa-Lisseanu, I. Niermann, and S. Kramer-Schadt. 2012. "The Catchment Area of Wind Farms for European Bats: A Plea for International Regulations." *Biological Conservation* 153: 80-86.
- Wade, P. R. 1998. "Calculating Limits to the Allowable Human-Caused Mortality of Cetaceans and Pinnipeds." *Marine Mammal Science* 14: 1-37.
- Wasser, S. K., B. J. Davenport, E. R. Ramage, K. E. Hunt, M. R. Parker, C. Clarke, and G. B. Stenhouse. 2004. "Scat Detection Dogs in Wildlife Research and Management: Application to Grizzly and Black Bears in the Yellowhead Ecosystem, Alberta, Canada." *Canadian Journal of Zoology* 82: 475-492.
- Weaver, S. P., C. D. Hein, T. R. Simpson, J. W. Evans, and I. Castro-Arellano. 2020. "Ultrasonic Acoustic Deterrents Significantly Reduce Bat Fatalities at Wind Turbines." *Global Ecology and Conservation* 24: e01099.
- Wilson, D., S. Hulka, and L. Bennun. 2022. "A Review of Raptor Carcass Persistence Trials and the Practical Implications for Fatality Estimation at Wind Farms." *PeerJ* 10: e14163.
- WII (Wildlife Institute of India). 2018. Power-Line Mitigation Measures. Uttarakhand, India: WII.
- World Bank. 2017. Environmental and Social Standard 6 on Biodiversity Conservation and Sustainable Management of Living Natural Resources. Washington, DC: World Bank.
- ———. 2018. Guidance Note 6, Biodiversity Conservation and Sustainable Management of Living Natural Resources, Washington, DC, World Bank
- WBG (World Bank Group). 2007. "Environmental, Health, and Safety Guidelines for Electric Power Transmission and Distribution" Washington, DC.
- ———. 2015 "Environmental, Health, and Safety Guidelines for Wind Energy." World Bank Group, Washington, DC

APPENDICES

APPENDIX A: PCFM DECISION SUPPORT TOOL

The PCFM Decision Support Tool is provided as a standalone Excel spreadsheet: **www.ifc.org/ windbirdbatmonitoring**

APPENDIX B: IMPLEMENTING PCFM FIELDWORK

Implementing an effective post-construction fatality monitoring (PCFM) fieldwork program at a wind energy facility (WEF) requires an appropriate study design and a well-planned search program managed by an experienced Lead Ecologist (LE) and conducted by an appropriately trained search team. Appendix B provides the LE with practical advice to help organize and manage PCFM fieldwork and collect the high-quality data necessary for generating reliable fatality rate estimates. Figure B.1 summarizes the LE's main tasks during a PCFM program.





Roles and responsibilities of the Lead Ecologist and search team

To ensure that the PCFM program runs smoothly, it is important that the LE and the search team understand their own and each other's responsibilities. Table B.1 provides a list of the typical responsibilities

 Table B.1
 Typical responsibilities of the Lead Ecologist and field team of searchers

POSITION	RESPONSIBILITIES
LEAD ECOLOGIST	 > Design and implement PCFM programs appropriate to the risk profile and stage of the project and informed by underlying PCFM principles (Section 3) and design guidance (see Section 4 and Decision Support Tool – Appendix A). > Communicate and coordinate PCFM program work with client and relevant WEF staff. > Train PCFM search team in field methods and data collection protocols. > Organize, manage, and coordinate fieldwork for PCFM searches, including: Preparing and distributing data collection forms, communication plans, health and safety protocols, etc. > Arranging appropriate, reliable transport to allow searchers to move between search plots. > Ensuring reliable communication during fieldwork between searchers and between searchers and the LE. > Managing and overseeing health and safety of field teams and conducting necessary risk assessments. > Arranging for storage of fatalities at the WEF, typically in a freezer located within the WEF operation and maintenance buildings. > Organize, manage, and coordinate fieldwork for bias correction trials, including: > Obtaining bias trial carcasses and arranging for these to be stored at the WEF. Obtaining decoys if these are being used for searcher efficiency trials. > Administering searcher efficiency and carcass persistence trials or training and managing a trial administrator to do so.^a > Manage data and conduct and report PCFM analysis, including: Conducting all fieldwork data. Conducting and reporting 6-monthly and annual data analysis, including calculating density weighted proportion (DWP) and generating fatality rate estimates using GenEst software. > Reporting findings and relevant concerns to client, lenders, and stakeholders
SEARCH TEAM	 Conduct fatality searches. Participate in searcher efficiency trials. Monitor carcass persistence trials and record and upload associated data. Understand and comply with health and safety plans and risk assessments, site rules, and personal protective equipment requirements. Record PCFM data (fatality observations, searcher efficacy, carcass persistence observations). Identify fatalities and provide informative high-quality photographs of fatality incidents that allow LE or other specialist to review species identification, condition, and general location of fatalities. Complete, submit, and store data collection forms as required.
STATISTICIAN, (IF AVAILABLE)	 Review sampling designs and provide advice and guidance on alternative approaches if necessary. Help LE conduct report and interpret seasonal and annual data analysis using GenEst.

^a In searcher efficiency bias trials, the trial administrator is an individual (often the LE) who places trial carcasses or decoys in the search plot before a search day begins, and checks whether they are still present if searchers do not find them. The person administrating trials must not be a searcher involved in the trials.

Tasks to be completed by the Lead Ecologist before the start of PCFM fieldwork

Effectively implementing a PCFM program safely and efficiently requires considerable pre-fieldwork planning, and it is important that the LE has sufficient time and budget to complete these tasks before the start of a PCFM program. The sections below provide details of the tasks that the LE will need to complete before the start of fieldwork.

WEF biodiversity document review

To ensure that the PCFM program is part of a structured framework and able to fulfil the WEFs monitoring objectives the LE should check that the WEF has developed an adequate Biodiversity Management Plan. This should include numerically derived annual fatality rate thresholds for priority biodiversity values and an appropriate adaptive management mechanism to evaluate and manage project impacts identified through the results and analysis of PCFM data (see Section 6 for more information). The LE should have a good understanding of all aspects of this plan and how it relates to the PCFM program.

Permits, permissions and legal requirements

The LE should ensure that any legal requirements are fulfilled and that any national or sub-national permits required are obtained before the monitoring program starts. These may include requirements related to reporting of fatalities of nationally protected species and treatment of injured species.

The LE should obtain site maps; information on landowners, land users and land ownership; project site access information, and health and safety protocols from relevant WEF staff. The LE should be familiar with this information and be aware of any access or other constraints relevant to the PCFM program at the WEF and, where relevant, WEF-associated OHL.

Determining the turbine sample and the search plot size

Once the number, hub height and blade length of turbines at the WEF is known, the LE can input this information into the DST [User Interface— Project setup tab] to determine the number of turbines to sample and the full search plot size information for the PCFM program. For First Year designs entering this information will provide the number of sample turbines and when relevant, search plot size for full plot searches. For Subsequent Year designs the LE should additionally answer questions 2 to 5 in the Subsequent Years tab of the DST to determine the search plot size. Once the number of turbines to sample and plot size for full plot searches are determined, the LE can randomly select which turbines will be searched and organize a site visit to the WEF to map the unsearched areas and visibility classes at each.





Site visit and mapping of unsearched areas and visibility classes



Additionally, the LE will use these visits to prepare maps of each sample turbine showing the extent of areas that will not be searched and the extent of different ground cover visibility classes.

Unsearched area mapping. To calculate unbiased fatality estimates requires the extent of the fatality fall zone that is unsearched to be quantified (see Section 3.3.7.1 for more information). There are two types of unsearched area: the area beyond the limit of the search plot out to the edge of the fall zone, and unsearchable areas within the search plot that cannot be searched because, for example, slopes are too steep, vegetation is too dense, or the ground surface is too rocky to walk over safely. To quantify the unsearchable areas within search plots the LE will need to visit each sample turbine and identify and map the extent of unsearched areas within the search plot. The LE should preview Appendix C to ensure they understand how these maps will be used. To accurately incorporate the unsearched area information into the fatality estimate, the LE will need to know the proportion of area not searched within each 5 m band around the turbines. Figure B.2 shows a sample turbine with unsearched areas mapped.





Note: In this example the DST recommended a search plot radius of 80m which is used as a minimum distance when defining the square plot.

Visibility class mapping. On the same site visit (and as part of the unsearched area mapping activity) the LE should also map the visibility classes at each sample turbine. In this Handbook, visibility classes are principally used to inform responses to questions required by the DST when generating a First Year or Subsequent Years design (i.e., User Interface—First Year tab Question 6 and User Interface—Subsequent Years tab, Questions 7 and 8). For Year 1 designs the visibility mapping is used to assess whether the vegetation height off the RAP at sample turbines is low enough to permit effective searching. In some habitats (e.g., deserts) the answer to this question may seem clear without visibility mapping, however, as good practice it is recommended that this exercise is conducted to provide evidence-based DST input values. This is particularly important when developing a Subsequent Years design as the input will influence the transect width output in the recommended design.

The Subsequent Years tab (see below) defines four visibility classes which should be used as a guide to mapping visibility at sample turbines and then used to inform input into the DST (for First Year and Subsequent Years).

- > Easy: Bare ground 90% or greater; all ground cover sparse and 15 cm or less in height (e.g., RAP area)
- > Moderate: Bare ground 25% or greater; all ground cover 15 cm or less in height and mostly sparse
- > Difficult: Bare ground 25% or less; 25% or less of ground cover over 30 cm in height
- > Very Difficult: Little or no bare ground; more than 25% of ground cover over 30 cm in height

Figure B.3 illustrates visibility mapping at a sample turbine. As is typical for many PCFM search plots not all visibility classes are present. In this case, no area in the search plot fulfils the 'Very Difficult' criterion.





When this mapping task is completed, the ground cover at sample turbines will be better understood and the LE will be able to answer all questions in the DST for either the First or Subsequent Years design and generate a full recommended PCFM design for the WEF.

Preparation for training the search team

The LE should make sure that fieldwork and safety protocols that will be required or relevant during PCFM monitoring are developed and ready for use before the search team training. These include the following:

- > PCFM fieldwork method statement and search schedule
- > Quality control protocol for daily checking of raw data for completeness and accuracy and specifying how errors will be corrected
- > Attendance protocol for tracking when search team members are present on site (i.e. sign-in/sign-out system)
- > Field-tested communication system and protocol, giving details of who is to be contacted for questions arising in the field and the method of communication (e.g., phone call, text, email)
- > Handling and care of injured animals protocol

The LE should also print out sufficient copies of all data recording sheets and obtain all equipment necessary to conduct the training of the search team effectively and efficiently.

Determining the number of searchers needed

To determine the number of searchers needed for a PCFM program, the LE should first calculate the total number of days it will take for a single search of all the sample turbines at a particular PCFM project. This will be based on:

- > Total transect length to be searched (across all search plots)
- > Time required to search transects
- > Additional time required to carry out the following activities associated with conducting searches:
 - Set up transects at each search plot
 - Travel between plots
 - Record fatalities
 - Check on-going carcass persistence trials

A hypothetical worked example provided in Tables B.2 to B.5 calculates the number of searchers required to conduct a single search of all sample turbines for a PCFM design with 20 sample turbines each with 160 m x 160 m search plots, a transect width of 10 m, and no unsearchable areas within the search plots. The worked example is provided to help the LE accurately calculate the number of searchers needed for their own PCFM program.







Calculating total length of transect to be searched.

To calculate the total length of the transect to be searched, the LE will need to obtain details of the **turbine sample**, **plot size and shape**, and **transect width** from the DST PCFM design output and the **size and shape of unsearched areas** within plots from maps of these areas created before the start of the PCFM program (see Determining the sample and the search plot size and Site visit and mapping of unsearched areas and visibility classes above for information on how the LE obtains these details).

Making a scale drawing of each sample search plot showing transects and unsearched areas may be helpful in calculating the total length of transect to be searched.

If the PCFM design has a sample search plot size and shape, transect width and unsearched areas that are the same for all sample plots:

> Follow the method in Table B.2 to calculate the total transect length to be searched for a single search of all sample turbines [final answer in Row 7 of Table B.2].

If sample search plot size and shape, transect width or unsearched areas differ between plots:

- Calculate sub-totals for each group of plots that have the same component values. (Use Table B2, rows 1-6 to calculate sub-totals for each group).
- > Sum the subtotals to obtain the total sample length for the PCFM study.

Table B.2 Calculating total transect length to be searched for a single search of all sample turbines

	COMPONENT FOR CALCULATING TOTAL TRANSECT LENGTH	COMPONENT VALUE	EXPLANATORY NOTES
1	Number of sample turbines	20	See text above; this will be the number of sample turbines with the same component values
2	Search plot (m x m)	160 x 160	Square search plot extending to minimum of 80 m from the turbine base
3	Transect width (m)	10	10 m wide spacing between transects (searchers will search 5 m either side of transect)
4	Transect length (m)	150	Search plot length minus 5 m at either end of transect (160 m – (5 + 5 m) = 150 m ^a
5	Number of transects per plot	16	Transects at 5, 15, 25, 35, 45, 55, 65, 75, 85, 95, 105, 115, 125, 135, 145, and 155 m = 16 transects per search plot (see Figure B.4)
6	Total transect length per search plot (m)	2,400	Transect length [Row 4] x number of transects [Row 5]
7	Total transect length per search visit (m)	48,000	Number of sample turbines [Row 1] x transect length per search plot [Row 6]

Note: Worked example—20 sample turbines.

^a. The worked example has no unsearched areas within any of the sample search plots, so none of the transects are shortened because of unsearched areas crossing the transect line. For search plots with unsearched areas, the transect length for each plot would need to be calculated separately.





Calculating time required to search transects

The time required to search transects will depend on ground conditions and height of vegetation and is best estimated by conducting a test transect search at the project or, if available, obtaining evidence from previous searches in similar conditions. This will provide the LE with a search rate that can be used to calculate the time required to search the total transect length for a single search of all sample turbines.

Table B.3 shows how to calculate the time required to search total transect length for a single search of all sample turbines by multiplying the total transect length calculated [Table B2, Row 7] by the search rate [Table B.3, Row 8].

Table B.3 Calculating time required to search total transect length for a single search of all sample turbines

ROW	COMPONENT FOR CALCULATING TIME REQUIRED TO SEARCH TOTAL TRANSECT LENGTH	COMPONENT VALUE	EXPLANATORY NOTES
8	Search rate along transect (pace at which searchers walk [m/s])	1.5	Based on field test or experience of previous bird fatality monitoring search in equivalent landscape
9	Total time to search total transect length per search visit (seconds)	72,000	[Table B.2, Row 7] x [Row 8]
10	Total time to search total transect length per search visit (hours)	20	Divide by 3,600 to convert seconds to hours

Note: Worked example—20 sample turbines.

Additional time required to carry out other activities associated with conducting searches

The LE should add on time needed for the following PCFM-related field activities to arrive at a realistic estimate of total searcher hours required.

- > Setting up transects within a plot: Likely to be a standard length of time per search plot.
- Recording fatalities: Should be based on time needed to record a single fatality and an estimate of the average number of fatalities likely to be found during a single search of all sample turbines. For example, allowing 15 minutes to record a fatality and an estimated average of five fatalities during a single search of all sample turbines will require at least 75 minutes to be added to the overall time required for each search visit.
- > Travel between search plots: Should be based on actual distance needed to travel between each turbine or power line search plot and take account of practical and landscape constraints and posted speed limits of public, WEF, or project-controlled roads.
- > Checking ongoing carcass persistence trial carcasses: Should be a standard amount of time per carcass and should include travel time between carcasses if checks cannot be integrated with turbine searches.
- > Work breaks: As prescribed in relevant protocols for searcher health and safety at the specific project site and to reduce searcher fatigue to maintain consistent detection rates.

Table B.4 shows how to calculate total time required for these additional activities for a single search of all sample turbines.

Table B.4	Calculating additional time required to carry out other activities associated with conducting searches for single search of
	all sample turbines

ROW	COMPONENT FOR CALCULATING ADDITIONAL TIME REQUIRED TO CONDUCT SEARCHES	COMPONENT VALUE	EXPLANATORY NOTES
11	Time to set up transects at search plots (minutes)	200	Based on 10 minutes per plot for 20 plots
12	Time to record fatalities (minutes)	75	Based on an average of 5 fatalities per search visit and 15 minutes per fatality to record
13	Time for searcher to move between plots (minutes)	285	Based on average of 15 minutes between finishing one plot and starting the next ^a
14	Time to check ongoing carcass trials (minutes)	100	Based on hypothetical estimate of 20 trials (per search visit) x 5 minutes per trial
15	Total additional time per search visit (minutes)	660	Sum of Rows 11 to 14
16	Total additional time per search visit (hours)	11	Divide by 60 to convert minutes to hours

Note: Worked example—20 sample turbines.

^{a.} For a project with 20 sample turbines the searcher will need to make 19 between-plot walks of 15 minutes each. Therefore the total time walking between plots is 19 X 15 = 285

Total time required for a single search visit to all sample turbines

As the final step, Table B.5 demonstrates how to calculate total time required for a single search visit to all sample turbines by summing total transect search time [Table B.3, Row 10] and additional time required [Table B.4, Row 16]

Table B.5 Calculating total time required for a single search visit to all sample turbines

ROW	COMPONENT FOR CALCULATING TOTAL TIME REQUIRED FOR SINGLE SEARCH VISIT TO ALL SAMPLE TURBINES	COMPONENT VALUE	EXPLANATORY NOTES
17	Total time required to conduct single search of all turbines (hours)	31	[Table B.3, Row 10] + [Table B.4, Row 16] = 20 + 11 =31
18	Total time required to conduct single search of all turbines (days rounded to nearest whole number)	5	Assuming 1 day = 6 hours of work time, allowing for breaks, divide Row 17 by $6 = 31/6 = 5.2$

Note: Worked example—20 sample turbines.

Once the LE has calculated the total number of days required to make a single search of all sample turbines, the number of searchers required will depend on the search interval that the PCFM design requires.

Number of searchers required for PCFM designs with different search intervals

Using the five working days required to conduct a single search of all sample turbines calculated in the worked example [Table B.5, Row 18], Table B.6 shows how to calculate number of searchers required for different PCFM design search intervals.

Table B.6 Calculation of number of searchers for PCFM design requiring twice-weekly, weekly, and every-other-week search intervals

TOTAL DAYS REQUIRED FOR SINGLE SEARCH VISIT TO ALL SAMPLE TURBINES	PCFM SEARCH INTERVAL, DAYS	NUMBER OF SEARCHERS REQUIRED
7 [5 working days + 2 weekend days]	3.5	7 / 3.5 = 2 searchers
7 [5 working days + 2 weekend days]	7	7/7 = 1 searcherª
7 [5 working days + 2 weekend days]	14	7/14 = 0.5 searchers (equivalent to 1 searcher working part time)

Note: Worked example—20 sample turbines.

^{a.} If a single searcher is conducting searches, the health and safety risks of working alone must be covered in the health and safety training and risk assessment. The WEF should establish a lone-worker protocol.

In addition to calculating how many searchers will be required to conduct the PCFM, the LE should make contingency plans to allow the program to continue when searchers are absent from work.

Recruiting and training the search team

The LE should train searchers in the necessary field and recording methods so that they can conduct PCFM but it is not essential that searchers are trained ecologists. The LE may contract searchers, or the WEF may employ them directly. To demonstrate independence, it is preferable for the LE to hire or contract the searchers directly but the most feasible option may be to make use of staff that the WEF already employs or arrange for the WEF to hire search team members from the local community. In both cases, the LE should be involved in selecting searchers and be provided with the resources to conduct adequate training. If WEF employees or local community members are employed as searchers, it is important that the LE provide the necessary support to the search team, including regular checks on implementation of field methods to ensure that searches are aligned with the PCFM design and are conducted to a high standard.

Training should be provided to all members of the field team of searchers in their principal working language and cover all relevant aspects of the PCFM program. Training should be conducted in a field situation, ideally at the WEF site, where the LE can demonstrate all aspects of PCFM to be practiced by search team members. Specifically, training should:

> Provide a summary of the field activities conducted during a PCFM program

This will include an overview of fatality searches and bias correction trials, with particular emphasis on roles and responsibilities of the fatality field team (see Table B.1 for roles and responsibilities).

> Explain all aspects of health and safety at the WEF

The LE should conduct a risk assessment with the searchers to highlight potential risks associated with their roles and responsibilities. Topics will include environmental risks, site-specific and activity related hazards, and where applicable working alone as a searcher. The LE will provide and explain the protocols and mitigation measures associated with these risks and ensure that appropriate good practice is adhered to throughout the program. As part of risk management, correct use of any personal protective equipment must be demonstrated to searchers. First aid training and fire training may be required at certain WEFs.

> Explain and demonstrate field skills required to conduct PCFM at turbines and along OHL

Training modules should explain and demonstrate good practice relevant to:

- > **Transect searching.** Training should emphasize consistency of effort and how to optimize reliability and efficiency (e.g., marking locations of fatalities found during a search, recording data once the search of the turbine is complete). Training should demonstrate appropriate use of a Global Positioning System (GPS) device for recording locations of fatalities; identifying search plot boundaries; and taking informative, good quality photographs of fatalities.
- > Handling and care of injured animals and fatalities. Training should follow protocols developed for the project, which should align with guidance provided in the *Handling and care of injured animals* and the *Field recording of fatalities* sections below.
- > Searcher efficiency trials. Training should explain that the principal purpose of searcher efficiency trials is to test for fatalities that searchers missed during fatality searches. It should emphasize that a trial administrator (usually the LE) will place trial carcasses (without the searchers' knowledge) at locations where fatalities could land and will not purposefully "hide" carcasses. The training will also explain that searcher

efficiency trials will take place periodically during the PCFM program. If decoys will be used in trials instead of actual carcasses, this should be explained during the training.

Carcass persistence trials. Training should explain the purpose of carcass persistence trials and provide clear protocols for confirming that a carcass has been removed and not just moved a short distance away (i.e., a careful search of a 10 m radius around the location where the carcass was deposited). There should also be standard criteria that searchers use to determine when a carcass is recorded as removed; two primary feathers connected, or 10 body feathers found within a 5 m x 5 m area (25 m²) is recommended as minimum evidence that a trial carcass is still present. The same level of evidence should be used to determine whether a carcass is recorded as present during fatality searches.

> Field data recording, including:

- **Filling in data collection forms**¹. The LE should discuss all input fields on every field data collection form (forms available in Appendices E-J) and set up mock searches and bias trials to give searchers practice filling out the various data collection forms under the LE's supervision. Common errors in data recording should be highlighted so that they can be avoided.
- **Photographing fatalities.** Emphasis should be on obtaining good quality, informative photographs adequate to identify a fatality. Carcasses should be photographed:
 - As found, taken from directly above
 - Turned over to show the ground-facing side
 - From different angles to show any features that may help with identification (e.g., head, wings, tail, feet, and additionally for bats, tragus (ear)
 - Showing any injuries
 - Showing the area immediately surrounding the fatality (e.g., relative to the nearest turbine, RAP, OHL)
 - All photographs should have a size reference in view (e.g., ruler).
- **Uploading and storing data.** The LE should demonstrate and give the searchers opportunity to practice (under supervision) how to store and upload data to a server or hard drive. Data should be stored and uploaded (and backed up) as frequently as possible (ideally once per day and no less than once per week).



Field and personal protective equipment

Having the correct equipment facilitates a high-quality PCFM study. A checklist for searchers of recommended essential and non-essential equipment is provided in Appendix K. Recommended personal protective equipment may include a hard hat, reflective vest, steel-toe capped boots, safety glasses, snake guards, full-length trousers, long-sleeved shirts, and sun hats, as directed by the developer and site-specific health and safety protocols. The LE should take direction from the WEF site manager on this topic.

¹ The LE should ensure that all data collection forms and instructions are available in a language that all members of the field team understand.

PCFM clearing searches

The first carcass search visit that the search team conducts (under the supervision of the LE) defines the beginning of the study period and is called the "clearing search," the purpose of which is to clear the search area of any bird or bat carcasses that might have been deposited before the study begins, so that a fatality estimate can be obtained for the study period only. Any new search plots used during the PCFM study (e.g., at the start of a new season, study year) should also be subject to a clearing search.

Bird and bat carcasses found during a clearing search should not be entered into the statistical analysis, but it is important that the search itself be documented and details on what was found recorded. The clearing search does not produce any data that are analyzed but rather serves to clear the search plots of old carcasses so that the fatality estimate is based on a period that starts immediately after the clearing search. These searches are also useful for detecting PCFM target species that may have been killed before the official start of the PCFM.

Any suitable fresh carcasses found during the clearing searches can be saved for searcher efficiency and carcass persistence trials if regulatory permits and project protocol allow (see Box 3.5 for carcass types that may and may not be used).

PRACTICAL ADVICE FOR CONDUCTING FATALITY SEARCHES

At the start of the PCFM program the LE will make sure that each search team member has all necessary data recording forms and equipment required to conduct PCFM monitoring. They will also provide a search schedule giving a daily plan of who will search each of the randomly selected turbines on each search visit. Throughout the PCFM program searchers should record the date when each turbine was searched and other details on the Search Summary Form (Appendix F).

During searches, all fatalities found during PCFM are recorded regardless of whether a First Year or a Subsequent Years design is being implemented and regardless of the PCFM target species. For example, a Subsequent Years design that is optimized for large birds with a relatively wide transect width and long search intervals will not be optimized for bats or smaller birds, but carcasses in these size classes will be found during searches and should always be recorded.

Maintaining data quality is critical for providing accurate input data for use in GenEst. To achieve this, the LE should conduct frequent checks of data completeness and accuracy. For example, after the first PCFM search, data collected should be reviewed to ensure that searchers understand the field methods and data collection forms. This should also provide an opportunity to address any questions about conducting searches and recording data that searchers may have. A minimum of one review each PCFM season is recommended.





Time of day to conduct PCFM searches

Searching at any time of day is generally acceptable, but when PCFM target species include bats, it is preferable for searches to begin early in the morning to reduce the probability of daytime scavengers removing carcasses before searchers detect them. From a searcher perspective, morning searches may also be preferable in summer because temperatures are cooler.

Changing the turbine search order so that the same turbines are not always searched at the same time of day is recommended.

How to conduct PCFM searches

Walked turbine transect searches.

Searchers should systematically walk regularly spaced transects according to the transect width specified in the PCFM design. A searcher will walk in a direction (e.g., north) along a transect until they reach the end, walk along the edge of a plot to the beginning of the next transect and walk the opposite direction (e.g., south). While walking, the searcher scans for fatalities in the area on both sides of the transect out to half the transect width. Searchers should walk the transect lines at a rate of approximately 45 to 60 m per minute (approximately 15 steps per 10 seconds). Once the searcher has walked all transects in this manner, they have searched the entire plot area. For RAP plots, the searcher will walk transects at the same width and rate, focused on the cleared dirt surfaces making up the RAP and not the surrounding land. Examples of walked transects across different plot types are shown (not to scale) in Figure B.5. If searchers have GPS devices, they should keep the track function on while operating the device so transect walk accuracy can be checked and corrective action taken if needed. An example of good tracks downloaded from a GPS device are provided in Figure B.6

It is important that the searcher is not focused on the GPS (so that they do not miss finding fatalities). Instead, the GPS should be switched on and in tracking mode and placed in their pocket or bag to be looked at only occasionally to make sure they are following the correct route.

Figure B.5 Different plot types and examples of ways to search them (Not to scale)



Note: Transect width depends on DST output. Irregular shapes (such as the example on the far right) can be searched in different ways depending on obstacles in the way as long as the transect width is maintained and there is no overlapping.



Walked OHL transect searches

The OHL sample search plots may require search teams to walk transects that extend over one or more kilometers and end a substantial distance away from their starting point.

The following transect walk strategies are recommended to minimize the distance that searchers need to walk when conducting PCFM searches along transects within an OHL search plot:

- > If **a single field team is using one vehicle** for dropping off and picking up searchers, it is preferable to use a PCFM design that has an even number of transects so that the searchers start and finish their search at the same location (see *Single search team* in Figure B.7).
- > If **two field teams are using one vehicle** for dropping off and picking up searchers, there can be an even or odd number of transects.
 - If **the PCFM design requires an even number of transect**s, the vehicle remains in the center of the search plot, and searchers walk transects away from the vehicle in both directions (see *Two search teams*; *even number of transects* in Figure B.7).
 - If **the PCFM design requires an odd number of transects**, the vehicle drops one search team at one end of the search plot and then drives the vehicle to the other end before conducting their transect search walk. (see *Two search teams; even number of transects* in Figure B.7).





Fatalities found along OHL search transects should be recorded, processed, and collected when found and not flagged and returned to later, which may be impractical.

Scanning searches

As described in Section 3.3.5 of the Handbook, a visual scanning search is an alternative method of searching for large bird fatalities over large areas. The scanning search is identical to the transect-based search component, with four important exceptions:

- > The method is only used for large birds, because small bird and bat carcasses are difficult to detect from a distance.
- > Searches are conducted by visually scanning the scan search area from specified observation points rather than by walking transects.
- > Unsearchable area bias correction should account for visually unscannable areas (areas that are visually obstructed).
- > Searcher efficiency trials require different methods.

Importantly, no turbine should have both a transect-based and a scanning search because the analysis becomes more complicated when both search types are used at one turbine, although it would be acceptable to have transect-based searches at some turbines and scanning searches at other turbines in the same WEF. Given these

additional complexities, only an advanced PCFM practitioner should use the scanning search method and with the assistance of a statistician.

Search Method. Scanning searches are conducted by positioning the searcher at a series of points (usually four — one in each cardinal direction) along the periphery of the turbine pad. From each point searchers use binoculars to scan a sector of the ground around the turbine out to a pre-defined distance, usually maximum blade tip height (Figure B.8 illustrates the scanning search method). Any carcasses observed during a scanning search are recorded and documented following the same procedures used in transect-based searches.

Figure B.8 Diagram of typical scanning search



Note: Image is not to scale.

Unscanned areas. Similar to transect-based search plots, scanning search plots must be delineated individually in the field to exclude areas that are unsearchable (unscannable) using this method, because vegetation or topographic relief blocks the view. This information is used to calculate the unsearchable (unscannable) area bias correction factor which is incorporated into the estimation of total large bird fatality.

Scanning search searcher efficiency trials. The design of searcher efficiency trials for a scanning search is slightly different from the design of searcher efficiency trials for transect-based searches because the carcasses are placed in a different spatial pattern. Unlike transect-based searches, in which trial carcasses are placed at random locations in the search plot, for a scanning search, trial carcasses are placed at varying distances from the turbine according to the expected collision fall distribution of large bird fatalities. Appendix D of this Handbook includes a

Searcher Efficiency Trial Placement Tool that guides placement of trial carcasses by generating random distances appropriate for this method.

For a given sample size of scanning searcher efficiency trials (e.g., 20 per season is an acceptable sample size; 15 is a minimum), the search plots where the trial carcasses should be placed should be randomly selected, and the Searcher Efficiency Trial Placement Tool (Appendix D) should be used to generate a random compass bearing (direction) from the turbine and a random distance for each carcass. If the random compass bearing and distance indicate a point that is unsearchable, the trial administrator should skip that point and choose another. The random distance can be as low as zero but should not be further than the maximum radius of the scanning search plot (to ensure that all trial carcasses are placed within the searched area). The LE should ensure that there are no more than two (ideally no more than one) searcher efficiency trial carcasses placed at a time at any one turbine for a scanning search. As with traditional searcher efficiency trials, the LE should track the number of trial carcasses found versus those available. The resulting data will be compatible with GenEst (see Section 5) for fatality estimation, and data can be formatted and processed as usual. Similar to ordinary search plots, an unsearched area correction is required (referred to as the density weighted proportion [DWP] of the area searched) to account for carcasses falling in the area beyond the search plot extent for the scanning searches (usually 100–120 m from the turbine base; see Section 3.3.7.1 for more information on the unsearched area, Box 3.4 for principles relating to DWP, and Appendix D for the DWP tool used to calculate DWP).

Handling and care of injured animals

Birds and bats found injured but still alive should be handled by experienced individuals who are trained in animal rehabilitation, vaccinated against rabies and tetanus, and authorized by regulatory permits or according to specific national legislation and project-specific protocols for handling injured animals. Once captured, the animal must be taken to the nearest animal rehabilitation or veterinary center. If none of the searchers are experienced handlers, the LE should assist (if they are experienced handlers and are nearby). If no one qualified or experienced in animal handling is available, or if care and rehabilitation is not feasible, leaving the injured animal undisturbed is recommended. All injured animals should be treated with care and compassion, and any action taken should be aligned with national and local animal welfare protocols and regulations and relevant cultural heritage values. Searchers should record as much information as possible about injured animals without putting themselves at risk of injury or stressing the animal further.

Field recording of fatalities found during searches

If a fatality is found, searchers should mark the location with a colored marker or flag (for ease of finding it later) and continue to complete the walked transects. Once the transects for that plot are complete, the searchers should return to each carcass found and record the details. Specifically, the searcher should:

- > Mark the fatality location using a GPS device.
- > **Take photographs** (see *Recruiting and training the search teams* above for details on how photographs should be taken).
- > Complete the Fatality Record form (Appendix E).

> If authorized, recover the fatality (Pick up the fatality, using leather or nitrile gloves and put it in a clear, sealable, labelled plastic bag. Fill out a Carcass Freezer Tag (Appendix I) and place the tag in the plastic bag with the fatality).

Labels for the outside of the plastic bag can be masking tape stuck on the outside of the bag and written with a pencil or permanent marker pen, or carcasses can be double bagged, using clear plastic bags, with an additional Carcass Freezer Tag put inside the outer bag. It is advisable that labels are clearly visible without having to take the fatality out of the bag. Keep all fatalities cool to prevent them from rotting. Smaller birds and bats can be kept in a cooler box until the end of the search day, when they should be deposited in a freezer designated for this purpose in the WEF complex. Keeping larger bird fatalities cool (e.g., vultures, cranes, eagles) is more challenging, and searchers may need to go to the freezer more than once on days when large birds are found. Using a Freezer Contents Log is advisable to keep track of fatalities placed in the freezer (Appendix J). Fatalities should be stored in a separate part of the freezer to surrogates used in bias correction trials, so all fatalities found during the PCFM study are easily located if further analysis is required. If searchers are not able to store the fatality, as much information as possible must be recorded at the turbine search plot. In additional photographs that may help identify the species. When PCFM target species include bird scavengers (e.g., vultures), carcasses should be removed from areas around turbines to avoid potentially attracting these species into high-risk areas.



PRACTICAL ADVICE FOR CONDUCTING SEARCHER EFFICIENCY BIAS CORRECTION TRIALS

To assist with the design of the PCFM program, the DST output recommends the number of searcher efficiency trial carcasses (per size class) that should be used (Appendix A). The following sections provide practical advice on conducting searcher efficiency trials recommended by the DST.

When and where to place searcher efficiency trial carcasses

Searcher efficiency trials should take place in each season and be conducted during scheduled fatality searches. Trial carcasses and decoys should be placed randomly² (using the Searcher Efficiency Trial Placement Tool in Appendix D), with no more than two trials conducted at any one turbine to reduce the probability of attracting scavengers to particular locations. If scavenging birds (e.g., vultures) are among PCFM target species at a WEF and carcasses rather than decoys are used in bias trials, random locations close to turbines should not be used because this might attract these species to the turbine collision risk zone. Using the Searcher Efficiency Trial Placement Tool will usually ensure that all visibility classes are properly represented.

² Randomly means that trial carcasses are placed i) at turbines selected randomly from those that are being searched and ii) at a random distance and compass direction (bearing) from the turbine base (but ensuring that each trial location is in a searchable portion of the plot). The Searcher Efficiency Placement Tool provides random distance and compass directions (bearings) for trial carcass placement.

Searcher efficiency trial carcasses, surrogates, and decoys

Depending on availability, searcher efficiency trials can use actual bird or bat carcasses found on site; surrogates such as ducks, geese, pigeons, and quail for birds; and dark-colored mice for bats. Artificial decoys can also be used, such as hunting bird decoys, bird toys, or rubber toy bats, but it is critical that these decoys and surrogates look reasonably similar to the carcasses expected on the site; brightly colored toys are not appropriate. Importantly, surrogates used in searcher efficiency trials are not always appropriate for use in carcass persistence trials (see *Practical advice for conducting bias correction carcass persistence trials — Carcass persistence trial carcasses and surrogates* below). If surrogates or decoys are not available, the LE could reach out to regulators or other stakeholders for advice regarding possible alternatives (see Section 3.3.7.2, Box 3.5 for more information on carcasses for searcher efficiency trials, and potential alternative approaches).

How to conduct searcher efficiency trials

The LE (or designated trial administrator not involved in the searches) should place trials earlier in the morning before normal searches commence. It is essential that searchers are unaware of the timing and location of searcher efficiency trials so that an estimate of typical searcher efficiency can be obtained.

Each trial carcass should be discreetly marked³ (so that it can be distinguished from actual fatalities) and dropped from waist height rather than being placed directly on the ground and its location recorded using a GPS device. The objective is not to hide carcasses.

Searchers will detect or fail to detect trial carcasses and report results to the trial administrator. At the end of the searches for that day, the trial administrator will check the locations of any trials that searchers did not find to determine whether they were removed (and thus not available to be found) before the search. Any undetected carcasses that the trial administrator cannot retrieve are considered "unavailable" and will not be counted in the data set. In most cases, searchers are allowed a single opportunity to detect searcher efficiency trial carcasses.⁴ Searcher efficiency is estimated based on the number of trial carcass/decoys found compared with the number that were 'available' to be found.

Field recording of searcher efficiency trial carcasses

Searchers should record found trial carcasses in the same way as for actual fatalities. The locations of all trial carcasses detected should be reported to the LE (or a designated trial administrator). All information related to the searcher efficiency trial should be recorded on a Searcher Efficiency Trial data collection form (Appendix G).

³ Markings should be discrete but unmistakable, such as with a small length of electrical tape around a foot or clipped flight feathers.

⁴ If the searcher efficiency bias trial design allows multiple opportunities to detect trial carcasses so that GenEst can estimate *k*, the trial administrator must check that the trials are still'available' to be found after each search attempt. See the GenEst user guide for more details.



PRACTICAL ADVICE FOR CONDUCTING CARCASS PERSISTENCE BIAS CORRECTION TRIALS

When and where to place carcass persistence trial carcasses

Scavengers may be more active during certain periods of the year or may respond to seasonal increases in the presence of carcasses in the landscape. Carcass persistence trials should therefore take place during each season throughout the PCFM program to ensure that seasonal differences in carcass persistence are accounted for in fatality estimates. Trial carcasses should be placed randomly in search plots or similar habitat at the WEF. Randomly placed trial carcasses should provide carcass persistence information that is broadly representative of the habitat within the WEF. No more than two carcasses should be placed near any one turbine to reduce the probability of the scavenging rate being biased high because carcasses in close proximity were more easily detected. When scavenging birds (especially scavenging raptors) are among the PCFM target species, trial carcasses should be placed from turbines to avoid attracting them into high-risk areas.

Carcass persistence trial carcasses and surrogates

Carcass persistence trials should use fresh carcasses or thawed carcasses that were frozen when fresh and stored frozen until use.⁵ Actual collision and electrocution fatalities should be used in carcass persistence trials where available and feasible. When actual fatalities are not available, suitable surrogates may be used.

Game birds and domestic poultry have the potential to be scavenged more quickly than wild birds and bats, resulting in fatality estimates that may be biased high. This is likely to be particularly pronounced for birds of prey (raptors) and for large species (e.g., cranes, storks) that are difficult for scavengers to remove. Only raptors and equivalent-sized large species surrogates should be used to test persistence for these groups. If raptor surrogates are not available, it may be possible to obtain an appropriate raptor persistence rate from published literature and use it in the fatality rate analysis until a sample of raptor carcasses can be sourced and a persistence trial conducted (see Section 3.3.7.2, Box 3.5 for more information on carcasses for searcher efficiency trials and potential alternative approaches).

How to conduct carcass persistence trials

To determine how long a bird (small, medium, large) or bat persists on the ground, fresh and intact carcasses (or carcasses that were frozen when fresh and stored in a freezer until needed⁶) are monitored periodically over the course of several weeks or until they are removed. Trial carcasses should be discreetly marked so that they can be distinguished from actual fatalities and dropped from waist height rather than being placed directly on the ground

⁵ Under no circumstances should carcass surrogates for any type of bias trials be acquired by killing individuals from wild populations of animals. If insufficient carcasses are available to conduct carcass persistence or searcher efficiency field trials the LE should follow advice provided in Section 3.3.7.2, Box 3.5.

⁶ Frozen carcasses should be allowed to defrost before being used in carcass persistence trials.

and their location recorded using a GPS device. Each trial should be checked frequently during the first week after placement and then less frequently. The LE should allow time in the searchers' daily routine to check carcass persistence trial carcasses according to the following schedule — Day 1, 2, 3, 4, 7, 14, 21, and 30 (and sometimes longer for raptors) after placing them.

If appropriate, unfound carcasses used for searcher efficiency trials can be left out as carcass persistence trials, but if artificial decoys were used or if most of the searcher efficiency carcasses were found, a stand-alone carcass persistence trial should be conducted.

Field recording of carcass persistence trial carcasses

The date and time the carcass was placed and the date and time it was checked should be recorded. All information regarding the carcass persistence trial should be recorded on a Carcass Persistence Trial Form (Appendix H). The visit on which a trial carcass was last recorded as present and the first visit it was recorded as absent are used to calculate the carcass persistence rate in GenEst. Recording the time of each trial visit as precisely as possible (date and time of day) will increase the precision of the carcass persistence estimates. In addition to contributing to the fatality estimate analysis, carcass persistence trials should be used to inform adjustments to the fatality search schedule in subsequent seasons.

LOGISTICS AND EQUIPMENT REQUIREMENTS

Vehicles

At least one reliable vehicle is needed for a successful PCFM study. (More may be needed depending on the number of searchers and size of the WEF.) The LE or developer may rent vehicles or reach an agreement with searchers to use their own. The LE should consider the need for high-clearance or four-wheel-drive vehicles and arrange for these to be available if needed and should ensure that there are enough licensed drivers to operate the number of vehicles required.

Site accommodation

Searchers may travel to the WEF daily if they live nearby, but if they live too far away to travel daily, closer accommodation should be arranged.

Freezer

Before fatality searches are initiated, a large freezer should be purchased and its location in the WEF operations and maintenance buildings should be agreed upon with WEF management. The freezer should be big enough to store trial carcasses and turbine- and WEF-associated OHL fatalities separately.

APPENDIX C: GENERALIZED ESTIMATOR OF MORTALITY — WORKED EXAMPLE

This appendix demonstrates the use of the Generalized Estimator of Mortality (GenEst) graphical user interface by analyzing a data set that is included with the GenEst software¹. Screen shots from the graphical user interface (Figures C.1 – C.24) are included throughout this worked example to help users follow along. Numbered dots are used to cross reference text to relevant places in the figures.

Step-by-step illustrated instructions are provided below that will help the user:

- > Install the program
- > Input a data set for each of the five data types
- > Run and understand the outputs from each analysis
- > Understand the final fatality estimate output

HOW TO INSTALL GenEst

GenEst is a free add-on package for the free statistical software R. The GenEst user manual (Simonis et al. 2018) describes how to install the GenEst analysis software. It is important to keep up to date with the latest versions of GenEst and R. (Screen captures are from GenEst Version 1.4.8.)

The following provides an example using a data set that is available in the GenEst application and package. To launch the GenEst software, the user must use a web browser and navigate to <u>https://connect.west-inc.com/</u> <u>GenEst/</u> or install the free statistical programming software R and the GenEst R package on their computer. The steps are as follows:

- > Install R following the instructions on the R Project website (https://www.r-project.org/).
- Install the GenEst R package by opening R and typing install.packages("GenEst", dependencies = TRUE) in the console and hitting enter.
- Once the package is installed, launch GenEst typing library(GenEst) and then runGenEst() on the line below.

Navigating GenEst

Figure C.1 illustrates the headings of various pages and tabs found in the GenEst software. The three main heading tabs — Data Input 1, Analysis 2, and Help 3 — will always be visible at the top of the page. These lead to all of the subheading tabs. The worked example will include numbered dots (as above) to cross-reference text to the figures.

¹ For users familiar with R, GenEst can also be used from the command line. For more information on implementing R using the command line, open R and type browseVignettes ("GenEst") into the R console.



GenEst Home Screen

When GenEst is first opened, the home screen (Figure C.2) will be displayed. It opens on the main Help tab 4 and Getting Started sub-tab 5. The text here provides a useful introduction.



GENEST v1.4.	.8 Data Input Analyses	s Help
For help, see: GenEst I	User Guide and GenEst Statisti	tical Models
5		
Getting Started	Example Data About	Disclaimers
GenEst is an R so power facilities. Mo 1. searcher eff 2. carcass per 3. schedule for 4. fraction of to <i>density-weig</i> 5. summary da carcass disc characterist	ftware package for estimating b ortality estimation requires five of iciency field trial results (SE), sistence field trial results (CP), r periodic carcass surveys (SS) otal carcasses falling in the sea ghted proportion (DWP), and ata from the carcass surveys (C covery along with values of othe ics	bird and bat fatalities at wind and solar data files:), arched area at each unit searched or CO), including unit and search date of each her, optional covariates or carcass
Data sets can be u guide. Example da	Iploaded under the Data Input ata sets are available for loading	t tab. Formats are explained in the user ig into GenEst under the Example Data tab.
Analysis involves s	several steps:	
 uploading data entering Get fitting search Efficiency estimating to (such as spottab) 	ataclick the Data Input tab, neral Input parametersclick her efficiency and carcass persi and Carcass Persistence tab otal mortality and splitting morta ecies or sector or season) as de	k the Analyses tab, sistence models click the Searcher bs, and tality estimate by various subcategories desiredclick the Mortality Estimation

PREPARING AND INPUTTING DATA INTO GENEST—THE FIVE DATA FILES

The first step in conducting a GenEst analysis is to load all the necessary data. The five input data files are those listed under the main Data Input tab () (Figure C.3).

Figure C.3 Display of data input main screen with the five tabs corresponding to the five data files needed to complete a fatality estimate)

Select Da	ata Files:	Searcher Efficiency	Carcass Persistence	Search Schedule	Density Weighted Proportion	Carcass Observations
Searcher El	fliciency (SE)					
Load file	No file selected					
Carcass Pe	rsistence (CP)					
Load file	No file selected					
Search Sch	edule (SS)					
Load file	No file selected					
Density We	ighted Proportion (DWP)					
Load file	No file selected					
Carcass Ob	servation (CO)					
Load file	No file selected					

The necessary data files are comma-separated value spreadsheets for searcher efficiency data, carcass persistence data, search schedule data, density weighted proportion (DWP) data, and carcass observation data. These files are described on the Help: Getting Started tab (Figure C.4), and examples of correctly formatted files can be downloaded from the Help: Example Data tab (S). Users who are not experienced in data analysis may be surprised to find that formatting data is often neither fast nor straightforward. Time and care are required to produce high-quality, correctly formatted data. Box C.1 provides more detail on formatting data.

Box C.1 Formatting input data



It is critical that the data are formatted properly for GenEst. In particular, users should take care to use consistent formatting for dates and consistent column names across data sets (including lower- and upper-case text). For example, GenEst might not recognize "6/6/21" and "6-6-21" as the same date, and will not recognize "Carcass Size" as the same column name as "Carcass size." The GenEst user manual provides data formatting rules that users should read carefully as they prepare their data.

All users should carefully examine the example data provided with GenEst before collecting any data to ensure that it will meet GenEst requirements and formatting. This worked example uses example data provided with GenEst, **but it is critical that users understand that correctly formatting data is essential and requires advance preparation.**

GENEST v1.4.8 Data Input Analyses Help
For help, see: GenEst User Guide and GenEst Statistical Models
Getting Started Example Data About Disclaimers
GenEst is an R software package for estimating bird and bat fatalities at wind and solar power facilities. Mortality estimation requires five data files:
 searcher efficiency field trial results (SE), carcass persistence field trial results (CP), schedule for periodic carcass surveys (SS), fraction of total carcasses falling in the searched area at each unit searched or <i>density-weighted</i> properties (DWP), and
 5. summary data from the carcass surveys (CO), including unit and search date of each carcass discovery along with values of other, optional covariates or carcass characteristics
Data sets can be uploaded under the Data Input tab. Formats are explained in the user guide. Example data sets are available for loading into GenEst under the Example Data tab.
Analysis involves several steps:
 uploading dataclick the Data Input tab, entering General Input parametersclick the Analyses tab, fitting searcher efficiency and carcass persistence models click the Searcher Efficiency and Carcass Persistence tabs, and estimating total mortality and splitting mortality estimate by various subcategories (such as species or sector or season) as desiredclick the Mortality Estimation tab

To load the example data, the user can navigate to the Help: Example Data tab 6 and select a data set to load (Figure C.5). Load the Wind—Cleared plots, bats + birds 7 data set by clicking Load Data 8 next to that data set. Users should explore the other example data sets, especially those similar to their PCFM design.





Note: The Wind—Cleared plots, bats + birds bats data sets were used in this example.

The program will automatically switch to the Data Input: Searcher Efficiency tab 9 and will display the example data (Figure C.6). Other sub-tabs (Carcass Persistence, 10 Search Schedule, 11 Density-Weighted Proportion, 12 and Carcass Observations 13) can be clicked to view those data. How the data should look when inputted, for each of these five variables, is shown and discussed below.

Figure C.6 Display of data input main screen with the five tabs corresponding to the five data files needed to complete a fatality estimate

	to Filmer	Searcher Efficiency	Carcass Persistence	Search Schedule	Density Weighted Proportion	Carcass Observations
Select Da	ita Files:					
Searcher Ef	ficiency (SE)					
Load file	No file selected					
Carcass Pe	rsistence (CP)					
Load file	No file selected					
Search Sch	edule (SS)					
Load file	No file selected					
Density Wei	ighted Proportion (DWP)					
Load file	No file selected					
Carcass Ob	servation (CO)					
Load file	No file selected					

Searcher Efficiency Input Data

If correctly formatted and inserted, the searcher efficiency data should display in the Data Input: Searcher Efficiency tab (Figure C.7). The columns labelled s1, s2, s3, etc., show the number of searches for which the trial carcass was left out for the searchers to find. A"o" means that the searchers did not find the trial carcass, "1" means that the searchers found the trial carcass on that search, and "N/A" means that the trial carcasses were no longer available for the searchers to find because they had already been found or had been scavenged. Note that each potential covariate (Size, Season, and Visibility) has its own column².

Select Da	ata Files:	Sea	rcher Efficien	9 Carcass	Persistence	Search Schedule	Density Weighted	Proportion	Carcass	Observation	ns
Searcher Ef	fficiency (SE)	Show	25 ~ entrie	es				Search	•		
Load file	No file selected	File: wi	nd_cleared\$SE	:							
Clear file			pkID	Size	Season	Visibility	s1 ;	s2	s3	s4	s5
		1	pk1	bat	spring	RP	0	NA	NA	NA	NA
arcass Pe	rsistence (CP)	2	pk2	bat	spring	RP	0	0	NA	NA	NA
Load file	No file selected	3	pk3	bat	spring	RP	1	NA	NA	NA	NA
Clear file		4	pk4	bat	spring	RP	1	NA	NA	NA	N
Search Sch	edule (SS)	5	pk5	bat	spring	RP	0	NA	NA	NA	NA
Load file	No file selected	6	pk6	bat	spring	RP	1	NA	NA	NA	NA
		7	pk7	bat	spring	RP	1	NA	NA	NA	NA
Clear file		8	pk8	bat	spring	RP	0	NA	NA	NA	N
Density We	ighted Proportion (DWP)	9	pk9	bat	spring	RP	0	NA	NA	NA	NA
Load file	No file selected	10	pk10	bat	spring	RP	0	1	NA	NA	NA
		11	pk11	bat	spring	RP	0	NA	NA	NA	NA
Clear file		12	pk12	bat	spring	RP	1	NA	NA	NA	NA
arcass Ob	eservation (CO)	13	pk13	bat	spring	RP	1	NA	NA	NA	NA
Load file	No file selected	14	pk14	bat	spring	RP	0	0	NA	NA	NA
Clear file		15	pk15	bat	spring	RP	0	NA	NA	NA	NA
		16	pk16	bat	spring	RP	1	NA	NA	NA	NA
		17	pk17	bat	spring	RP	1	NA	NA	NA	N/
Clear All		18	pk18	bat	enring	PP		NA	NA	NA	NU

Figure C.7 Display of data input: searcher efficiency screen with data correctly inputted

² Note that the Wind—Cleared plots, bats + birds' dataset includes information on visibility classes and is used in the worked example analysis as an optional predictor variable to demonstrate how a covariate like visibility may be used. For practical reasons, the Handbook does not specifically recommend that the practitioner collects data to test for effects of different visibility classes on fatality estimates (see Section 3.3.7.2). If a PCFM program has measured searcher efficiency and/or carcass persistence using an adequate sample for each visibility class present in the search plots, then the worked example will help the user handle this data in GenEst.
Carcass Persistence Input Data

Each carcass persistence trial carcass (cpID) is checked on days 1, 2, 3, 4, 7, 10, 14, and 21. If correctly formatted and inserted, the carcass persistence data should display in the Data Input: Carcass Persistence tab (Figure C.8). "Last Present" represents the number of full days (the whole number) and the fraction of the day from the last search (decimal fraction) when the carcass was last seen. "First Absent" represents the number of full days (the whole number) and the fraction of the day from the carcass was first discovered gone. Note that each potential covariate (Size, Season, and Visibility) has its own column.

	v1.4.8 Data Input Ana	alyses Hel	p		10					
Select Da	ata Files:	Sear	cher Efficiency	Carcass Persi	stence	Search Schedule	Density Weigh	ted Proportion	Carcas	s Observations
Searcher E	fficiency (SE)	Show	25 v entries					Search:		
Load file	No file selected	File: wir	nd_cleared\$CP							
Clear file			cpID	Size	Season	Visibility		LastPresent		FirstAbsent
		1	cp1	bat	spring	RP			0	0.93
Carcass Pe	rsistence (CP)	2	cp2	bat	spring	RP		0.	98	1.97
Load file	No file selected	3	срЗ	bat	spring	RP			0	1.01
Clear file		4	cp4	bat	spring	RP		13.	99	21.13
Search Sch	nedule (SS)	5	cp5	bat	spring	RP			0	1.17
Load file	No file selected	6	cp6	bat	spring	RP		20.	95	27.97
_		7	cp7	bat	spring	RP			0	0.92
Clear file		8	cp8	bat	spring	RP		0.	96	1.92
Density We	ighted Proportion (DWP)	9	ср9	bat	spring	RP			0	0.96
Load file	No file selected	10	cp10	bat	spring	RP		2.	05	2.91
Clear file		11	cp11	bat	spring	RP		0.	92	2.02
		12	cp12	bat	spring	RP			0	0.9
Carcass Ob	oservation (CO)	13	cp13	bat	spring	RP		0.	93	1.89
Load file	No file selected	14	cp14	bat	spring	RP		3.	94	6.97
Clear file		15	cp15	bat	spring	RP			0	0.96
		16	cp16	bat	spring	RP		1.	88	2.98
Close All		17	cp17	bat	spring	RP		1.	05	1.9
Clear All		18	cp18	bat	spring	RP		9.	96	13.88
		19	cp19	bat	spring	RP			0	0.98

Figure C.8 Display of data input: carcass persistence screen with data correctly inputted

Search Schedule Input Data

If correctly formatted and inserted, the search schedule data should display in the Data Input: Search Schedule tab (Figure C.9). The search schedule data contains a row for each date that a search occurred and columns for every turbine at the facility. The turbine columns are filled with 0s and 1s, indicating whether the turbine was searched on that date (1) or not (0). A column indicating the season associated with the search date may also be included. Note that the schedule includes a column for the time-based covariate "Season."





DWP Input Data

If correctly formatted and inserted, the DWP data should display in the Data Input: Density Weighted Proportion tab (Figure C.10). The DWP is the area correction applied to the estimate to account for carcasses that may have fallen outside the search area (outside the specified plot boundary) and in any unsearchable areas within the plot. It is called a DWP because it is a combination of the carcass fall density (where carcasses are likely to fall) and the proportion of area searched. It is an estimate of the proportion of carcasses that are likely to fall within an area that was searched. GenEst does not estimate the DWP; it is an input the user must provide. Appendix D provides a spreadsheet-based tool that will estimate the DWP. In this example, every turbine has its own DWP value for each carcass size. It is preferred that DWP be estimated for each turbine separately.

CENEST v1.4.8 Data Input A	nalyses Help				12		
Select Data Files:	Searcher	Efficiency Carca	ss Persistence	Search Schedule	Density Weighted Proportion	Carcass Obser	vations
Searcher Efficiency (SE)	Show 25	✓ entries			Search	:	
Load file No file selected	File: wind_c	leared\$DWP					
Clear file		Turbine		bat 🗧	sml	med :	Irg 🗧
	1	t8		0.688	0.944	0.612	0.555
Carcass Persistence (CP)	2	t9		0.688	0.944	0.612	0.555
Load file No file selected	3	t10		0.688	0.944	0.612	0.555
Clear file	4	t13		0.688	0.944	0.612	0.555
Search Schedule (SS)	5	116		0.688	0.944	0.612	0.555
Load file No file selected	6	t18		0.688	0.944	0.612	0.555
	7	127		0.688	0.944	0.612	0.555
Clear file	8	131		0.688	0.944	0.612	0.555
Density Weighted Proportion (DWP)	9	134		0.688	0.944	0.612	0.555
Load file No file selected	10	137		0.688	0.944	0.612	0.555
Clear file	11	142		0.688	0.944	0.612	0.555
	12	155		0.688	0.944	0.612	0.555
Carcass Observation (CO)	13	t61		0.688	0.944	0.612	0.555
Load file No file selected	14	167		0.688	0.944	0.612	0.555
Clear file	15	t70		0.688	0.944	0.612	0.555
	16	t72		0.688	0.944	0.612	0.555
	17	178		0.688	0.944	0.612	0.555
Clear All	18	t80		0.688	0.944	0.612	0.555

Figure C.10 Display of data input: density-weighted proportion screen with data correctly inputted

Carcass Observations Input Data

If correctly formatted and inserted, the carcass observation data should display in the Data Input: Carcass Observation tab (Figure C.11). In the columns shown on this screen, carcasses will be listed in numerical order according to their carcass identification numbers. Turbine number, Date Found, and Species are essential input variables here, although other variables such as Turbine Type, Visibility, Species Group, Size, and Distance can be included if they are important to the study. **All variables that are important to the study must be included in the searcher efficiency data, carcass persistence data, search schedule (if the variable is time bound), and carcass observations data, and the columns must be named and formatted identically in all data files.**

Select Da	ata Files:	Se	archer Effici	ency Car	cass Persistence	Search Schedu	le Density	Weighted Pro	oportion Carcass	Observatio	ns
Searcher Ef	ficiency (SE)	Show	25 v e	ntries					Search:		
Load file	No file selected	File: v	vind_cleared	d\$CO							
Clear file			carcID	Turbine	Turbine Type	DateFound	Visibility	Species	SpeciesGroup	Size	Distance
Caroace Bo	reistance (CP)	1	x25	t78	z	1955-04-27	RP	LA	brd2	Irg	9.
Logd file	No file selected	2	x39	196	Z	1955-05-06	D	LC	brd3	Irg	54.
Load nie	No file selected	3	x117	t13	x	1955-05-12	D	SC	brd1	sml	52.
Clear file		4	x174	19	x	1955-05-12	D	SC	brd1	sml	43.
Search Sch	edule (SS)	5	x175	137	Y	1955-05-12	RP	BB	bat1	bat	9.
Load file	No file selected	6	x181	t91	Z	1955-05-12	RP	BA	bat1	bat	2.
		7	x183	t96	Z	1955-05-12	D	BA	bat1	bat	31.
Clear file		8	x48	134	Y	1955-05-15	D	SA	brd1	sml	5
Density Wei	ighted Proportion (DWP)	9	x239	t87	z	1955-05-15	D	SC	brd1	sml	5
Load file	No file selected	10	x133	127	х	1955-05-18	м	BB	bat1	bat	19.
Classifie		11	x292	t67	z	1955-05-18	М	LA	brd2	Irg	41.
Clear ne		12	x244	t72	Z	1955-05-21	D	BA	bat1	bat	2
Carcass Ob	servation (CO)	13	x273	t70	Z	1955-05-21	D	SG	brd1	sml	4
Load file	No file selected	14	x327	t96	z	1955-05-21	D	BA	bat1	bat	24.
Clear file		15	x330	t81	z	1955-05-21	D	BA	bat1	bat	54.
		16	x334	t80	z	1955-05-21	D	MC	brd1	med	38.
		17	x357	t16	x	1955-05-21	D	SG	brd1	sml	46.
Clear All		18	x111	t10	x	1955-05-27	D	BA	bat1	bat	50.
		19	x287	110	x	1955-05-27	D	BB	bat1	bat	20

Figure C.n Display of data input: carcass observations screen with data correctly inputted

ANALYZING INPUT DATA

The various sub-heading tabs displayed in Figure C.12 are found under the Analyses main tab **(2)**.





General Analysis Inputs

A GenEst analysis requires three general inputs (Figure C.13). First, the user specifies the number of iterations 14 to be used for bootstrapping (or simulating from) the model. The bootstrap simulations are used to construct confidence intervals (CIs) on the estimates. A larger number of simulations will improve the accuracy of the CIs but computations will take longer. A generally accepted value of 1,000 iterations was used in this analysis, but users are encouraged to use 5,000 iterations if the computations are not unacceptably slow. Second, the user specifies the confidence level 15 for the analysis. Typical choices are 0.90 (90% CI) and 0.95 (95% CI). The default value of 0.90 was used in this analysis, which is a good choice for good international industry practice. Third, if applicable, the user specifies the carcass class. If the study focuses on one size class (e.g., bats only), the user does not need to select the carcass class. More commonly, the carcass class column allows the user to produce fatality estimates according to type of carcass (e.g., size class). For this example, Size 17 was chosen because the example data include carcass information for bats and small, medium, and large birds. Visibility was another option by which to stratify, but it was not chosen for this example. Any selected carcass class column must also be present in the searcher efficiency, carcass persistence, and carcass observations data. After all the above is completed, the user can move to the Analyses: Searcher Efficiency sub-tab 18.

Figure C.13 Screenshot of landing page for "Analyses" main tab of GenEst software

GENEST V	1.4.8 Data Input	Analyses Help		
General Inputs	Searcher Efficiency	18 Carcass Persistence	Mortality Estimation	Detection Probability
Number of Ite	erations: 14			
1000				
Confidence L	evel: 15			
0.9				
Carcass Clas	s Column (optional):	16		
Size 17		•		
Visibility				
Size				

Note: These are global settings needed to run the GenEst analysis.

Searcher Efficiency

The Analyses: Searcher Efficiency tab 18 is empty when first loaded (Figure C.14). The user must specify the model inputs, which are the observations 19, the predictor variables (e.g., visibility class or season) 20, and the value for k 21. The searcher efficiency schedule contains a row for each trial carcass (or surrogate) placed and available to be found. The "observations" are a column of 0s and 1s indicating whether the available carcass was found (1) or not found (0). In the example data, the column s1 is the column of 0s and 1s for the first attempt to find each carcass. The data also includes s2, s3, and s4 for data on subsequent searches, but only data from the first search has been used in this example. Searches s2, s3, and s4 could be used to estimate the parameter k, but most practitioners will not have such data so s1 was selected as the observations field.

The user can also specify predictor variables (also known as covariates), such as season or plot type, that might influence searcher efficiency. For this example, season and visibility were chosen. Lastly, because data from a single searcher efficiency trial are being used, k = 0.67 was set for all size classes. The parameter, k has rarely been estimated, but tends to be near 0.67 in those cases (mostly for bats) where it has been estimated (e.g. Huso et al. 2017, Skalski and Townsend 2017. Once these settings are specified, the Selected Data sub-tab **22** will populate with the chosen data for review, and the user can select Run Model **23** to fit searcher efficiency models.

ENEST ef Mortality Estimator	v1.4.8 Data Inpr	Analyses Help					
neral Input	Searcher Efficiency	Carcass Persistenc	e Mortality Estimation	Detection Probability			
Model In	iputs:	Selected	22 Model Compariso	on Figures Estimates	Model Selection		
Observatio	ons: 19	File: wind_	cleared\$SE				
s1		Show 25	 ✓ entries 			Search:	
Predictor	Variables (optional): 2	0	Size	Season	Visibility		s
Season	Visibility	1	bat	spring	RP		
Fixed	k (optional)	2	bat	spring	RP		
bat:	0.67	3	bat	spring	RP		
Irg:	0.67	4	bat	spring	RP		
med	0.67	5	bat	spring	RP		
sml:	0.67	6	bat	spring	RP		
Run Mor		7	bat	spring	RP		
. tart mot		8	bat	spring	RP		
		9	bat	spring	RP		
		10	bat	spring	RP		

Figure C.14 Screenshot of Analyses: Search Efficiency: selected data tab in the GenEst software

Note: The si column was used for the response (trial carcass found or not) and Season and Size as potential predictor variables (covariates).

GenEst fits all possible combinations of models, based on the Predictor Variables that the user selects. After the Run Model 23 button is pressed, GenEst automatically displays the Analyses: Searcher Efficiency: Model Comparison tab 24 (Figure C.15), which shows the models fitted and their corrected Akaike Information Criterion (AICc) scores 23 and rankings 26 (see Box C.2 for more information on AICc scores and Box C.3 for more detail on the p-Formula).

Searcher Efficient	y Carcas	24	27		
<u>Model Inputs:</u>		Selected Data Model Comparison	Figures Estimates Model Selection		
s1		Carcass class: bat			
Predictor Variables (optional):		Show 25 V entries	h Francis	Search: 25	
Season Visibility		p Formula	K Formula	AICC	DAICC
Fixed k (optional)		3 p~visibility	k fixed at 0.67	238.64	0
bat: 0.67		2 p ~ Season + Visibility	k fixed at 0.67	241.32	2.68
Irg: 0.67		5 p~constant	k fixed at 0.67	247.79	9.15
med 0.67		1 p~Season Visibility	k fixed at 0.67	249.20	10.02
sml: 0.67		4 p - Season	K IIXEU at 0.07	200.46	11.04
Run Model		Showing 1 to 5 of 5 entries		Previous	I Next
Clear Model		Z Download			
able & Figure Selection:					
arcass Class:					
bat	-				
Model:					
p ~ Visibility	-				
Model:					
k fixed at 0.67	-				

Figure C.15 Screenshot of Analyses: Searcher Efficiency: Model Comparison tab of the GenEst Software

Note: This tab shows the ranking of competing models, scored according to corrected Akaike Information Criterion (AICc). The Figures tab can be viewed to compare the model estimates graphically.

Box C.2 Understanding Model Output: Corrected Akaike Information Criterion (AICc) Scores



GenEst uses AICc (Akaike 1973; Burnham and Anderson 2002) to rank models. AICc is a standard model score that evaluates how well a model fits the data. Models with lower AICc scores are better than models with higher AICc scores (but AICc can only be compared between models using exactly the same data set). GenEst automatically ranks the models by ΔAICc, which is a model's AICc value minus the minimum AICc in the model set (ΔAICc = 0 is always the model with the lowest AICc). Generally, models within 2 AICc points can be viewed as receiving similar support from the data (Burnham and Anderson 2002). AICc rank is a good place to start for model selection but a good practitioner will consider all models within 2 AICc points of the top-ranked model and choose the most reasonable of them (see Box C.4). When multiple models fit the data well and have AICc scores within 2 points of one another, the model with fewer parameters is usually preferred because having fewer parameters results in greater precision.

Box C.3 Understanding the p-Formula

The "p Formula" column in Figure C.15 indicates the model, where "p ~" reads: "p as a function of..." For example, in the output in Figure C.15, the top two models are: "p ~ Visibility" and "p ~ Season + Visibility." The "p ~ constant" model (ranked third) has no covariate effects; the "p ~ Visibility" model incorporates the effect of visibility on searcher efficiency (p as a function of visibility).

A searcher efficiency model is automatically fitted for each size class separately based on the size class column indicated in Figure C.11. Figure C.15 shows the results for bats, but the user must inspect models for each size class, which is achieved by choosing a different size class to display in the menu below Model Inputs.

Based on AICc rank, number of parameters, and whether the model is reasonable, the user chooses a model for further evaluation (Box C.4). In the case of bats, the model with visibility as a covariate has the lowest AICc by more than 2 points (the next-ranked model has ΔAICc = 2.68). Therefore, the model with visibility as a covariate, "p ~ Visibility," was chosen, which means that predictions of search efficiency for bats from the model will be unique for each visibility class in the data, and the estimates will not vary according to season. The Figures sub-tab ²⁰ shows differences among the competing model's predictions. Users should examine this graph to ensure that the selected model is making reasonable predictions and choose a different model if there is a problem with the selected model. The GenEst user guide provides information on interpreting these graphs.

Box C.4 Model Selection

GenEst fits several models to the carcass persistence and searcher efficiency data and leaves it to the user to select the best among them. There are three basic criteria that a user can apply to this task:

- 1. AICc (Box C.2) gives an objective measure of model fit.
- 2. For models within 2 AICc points of one another, the model with fewer parameters is usually preferred because having fewer parameters results in greater precision.
- 3. "Reasonableness" of a model refers to how well the model fits the data and whether it makes sense. It can take some practice to determine whether a model is reasonable. A good way to determine whether a model is reasonable is to look at the 'Figures' sub-tab 27 in Figure C.16 (searcher efficiency) or Figure C.17 (carcass persistence). "Reasonableness" should be assessed by examining the figures and considering whether the model makes sense. For example, if a searcher efficiency model predicted that carcasses were more detectable in tall grass than on a gravel pad, that might be an unreasonable model, and the user should consider whether there is a good explanation for the model. The model may be missing an important covariate (e.g., season, if grass height depends on season, or searcher, if searchers have different skill levels). Alternatively, small sample size can sometimes lead to unreasonable models by chance. In the latter case, it may be preferable simply to remove an unreasonable covariate from the model.

The Estimates sub-tab 28 in Figure C.16 shows the estimates that the model chosen for evaluation in Figure C.15 produced. Four columns are displayed. Cell 29 refers to the group for which the estimate applies (visibility class in this example); n 30 is the sample size within each stratum (or cell); p 31 in this case is the estimated search efficiency and associated 90% CI; and k 32 shows the detection reduction factor, which is constant in this case. Because the model has the lowest AICc (lower AICc is better) and appears to produce a reasonable output, move to the Model Selection sub-tab 33 and select the "p ~ Visibility" model from the drop-down menu. A model must be selected for each size class after inspection of the competing models for each size class. GenEst does not automatically select the top-ranked model. The user must select the model manually.



neral Input	ts Searcher Eff	iciency Car	cass Persistence Mortality Es	timation Dete	ction Probability		
					27 2	.8 33	
Model In	nputs:		Selected Data Model	Comparison F	Figures Estimates	Model Selection	
Observati	ions:		Carcass class: bat				
s1			Table shows median estimate	es and 90% confid	ence intervals		
Predictor	Variables (optiona	il):	Show 25 v entries			Search:	
Season	Visibility		Coll 29	30		× 37	
Fixed	k (optional)		1	60	P 0.25 (0.17, 0.352)	0.67.10.67.1	0.671
bat:	0.67		2 M	60	0.467 (0.364, 0.573)	0.67 [0.67, 0	0.671
Irg:	0.67		3 RP	60	0.567 [0.46 0.667]	0.67 [0.67, 0	0.671
med	0.67		Showing 1 to 3 of 3 entries		0.007 [0.40, 0.007]	0.07 [0.07, 0	Previoue 1 N
sml:	0.67		Showing 1 to 5 of 5 entities				
Dup Mar	del		📥 Download				
Run Mod							
Clear Model							
Table &	Figure Selectio	<u>n:</u>					
Carcass C	Class:						
bat		•					
Model:							
p ~ Visib	pility	•					
Model:							

Note: For searcher efficiency, this tab displays the estimated probability of detection and associated confidence interval (CI); k is fixed, so the point estimate and CI bounds are all the same.

In summary, GenEst takes field data on searcher efficiency trials and fits a set of models. The models differ in terms of predictor variables (covariates). GenEst produces searcher efficiency estimates from each model and ranks the models using AICc. The user then selects the model to use for mortality estimation based on AICc ranks, number of model parameters, and an evaluation of the 'reasonableness' of model predictions.

Carcass Persistence

The next sub-tab under Analyses 14 is Carcass Persistence 34 (Figure C.17). GenEst estimates average carcass persistence time (in days) and computes the probability of a carcass persisting through a specified search interval. The data set includes a row for each trial carcass placed on the landscape. Because a carcass is rarely, if ever, observed being removed from the landscape, the observations recorded are the last date and time the carcass was checked and found and the first date and time that the carcass was checked and not found. For example, if a carcass was placed on day 0, a typical carcass check schedule might be day 1, 2, 3, 7, 14, 21, and 30. If the carcass disappeared between searches on days 3 and 7, the last day present would be 3, and the first day absent would be 7. GenEst can produce more precise estimates if users record the date and time when carcasses are placed and checked rather than just the date.

ENEST v1.4.8 Data	Input A	nalyses	Help			
neral Inputs Searcher Ef	ficiency	Carcass Pe	ersistence	Mortality Es	timation Detection Pro	bability
Model Inputs:	Sel	ected Data	Model Co	38 omparison	Figures Estimates	Model Selection
Last Time Present:	File: v	vind_cleared	\$CP			
LastPresent -	Show	25 v en	tries		Search:	
First Time Absent:		Size	Season	Visibility	LastPresent	FirstAbsent
FirstAbsent -	1	bat	spring	RP	(0 0.93
Predictor Variables (optional):	2	bat	spring	RP	0.98	3 1.97
Season Visibility 3	5 3	bat	spring	RP	(0 1.01
Distributions to Include	4	bat	spring	RP	13.99	9 21.13
exponential 36	5	bat	spring	RP	(0 1.17
 Weibull Iognormal 	6	bat	spring	RP	20.9	5 27.97
	7	bat	spring	RP	(0.92
Run Model	8	bat	spring	RP	0.96	5 1.92
	0	bot	opring	DD		0.006

Figure C.17 Screenshot of Analyses: Carcass Persistence: Selected Data tab in the GenEst software

Note: The user selects the appropriate columns for the response data (last date and time present and first time absent). Season and visibility were also selected as potential covariates. By default, GenEst fits four distributions. It is suggested that this default setting is used.

Like the searcher efficiency analysis, the user can specify which predictor variables (covariates) **35** to consider in the set of competing models and which distributions to fit (buttons for Distributions to include **36** in left window of Figure C.17Q). The default is to fit all distributions, and this is suggested as standard practice. Model selection criteria and visual inspections of the fitted models will guide model selection. Clicking the Run Model **37** button results in all models being fitted, and the Model Comparison **38** sub-tab will automatically display.

The Model Comparison tab 38 displays all the fitted models and their properties (Figure C.18): the fitted distribution (Distribution) 39, the formula for the location parameter (Location Formula) 40, the formula for the scale

parameter (Scale Formula) (4), and the AICc scores (AICc and ΔAICc; Box C.2) (42). Each of these properties will be described in turn. All the carcass persistence distributions fitted by GenEst are parameterized (defined) according to a location parameter and a scale parameter. The location parameter defines the central tendency of the distribution (e.g., the mean of a normal distribution), and the scale parameter defines the spread around the central tendency (e.g., the standard deviation of a normal distribution). The exponential distribution is defined only according to the location parameter. GenEst allows both the location and the scale parameter to vary according to covariates, a feature that sets it apart from other estimators. The AICc column reports the AICc value, and ΔAICc is the difference from the top-ranked model. Like the Searcher Efficiency analysis, the user can compare models by checking the Figures sub-tab (43) and evaluate the estimates of any selected model (chosen via the window on the left of the screen). Users should select a model based on AICc score (lower is better) and check the figures to ensure that the model predictions make sense (Box C.4). Also, like the Searcher Efficiency analysis, the results are displayed with only one size class at a time; the user must inspect models for each size class independently.



eneral Inputs Searcher Efficiency	Carcass Persistence	Mortality Estimation Dete	ection Probability		
,		38 4	3		
Model Inputs:	Selected Data	Model Comparison Figures	Estimates Model Selec	tion	
Last Time Present:	Carcass class: bat				
LastPresent	Show 25 v entr		h:		
First Time Absent:	Distribution	39 40	41 Scale Formula	AICo	AAICo
FirstAbsent -	22 weibull	Le Segeon + Visibility		750 7	Anot
Predictor Variables (optional):	22 weibull		s ~ Season	751.61	0.0
Season Visibility	20 weibull	I ~ Visibility		751.01	1.5
Distributions to Include	27 weibull	I~ Season + Visionity	s ~ Constant	752.21	1.0
exponential weibull	12 weibull	I ~ Visibility	s ~ Season + Visibility	752.01	1.0
	12 weibull	1~ Vieibility	s ~ Visibility	754.86	2.1
Run Model	17 weibull	I ~ Visibility	s ~ Visibility	755.01	4.1
Clear Model	13 weibull		s ~ Season + Visibility	755.8	4.0
	52 leggermal	I - Seecon + Visibility		756.04	5.
Table & Figure Selection:	52 lognormal	I~ Season + Visibility	s ~ constant	750.24	5.5
Carcass Class:	47 lognormal	I ~ Season + Visibility	s ~ Season	757	6.
bat 👻	21 weibull	I ~ Season * Visibility	s ~ Season	757.76	7.0
Distribution:	77 loglogistic	I ~ Season + Visibility	s ~ constant	757.86	7.1
weibull	26 weibull	I ~ Season * Visibility	s ~ constant	758.41	7.7
Location:	53 lognormal	I ~ Visibility	s ~ constant	758.88	8.1
I ~ Season + Visibility -	72 loglogistic	I ~ Season + Visibility	s ~ Season	758.91	8.2
Scale:	42 lognormal	I ~ Season + Visibility	s ~ Visibility	760.57	9.8
s ~ Season 🗸	78 loglogistic	I ~ Visibility	s ~ constant	760.98	10.2
	11 weibull	I ~ Season * Visibility	s ~ Season + Visibility	761.01	10.3
	48 lognormal	I ~ Visibility	s ~ Season	761.31	10.6

Note: The model formulas are displayed, and the models are ranked according to corrected Akaike Information Criterion (AICc). The user can further evaluate the differences between the models graphically by viewing the Figures tab.

In this example, there are four models with ΔAICc less than 2 (Figure C.19). Because models within 2 AICc points of one another can be considered equally supported by the data, the second ranked model was chosen because it is simpler; the location parameter is a function of visibility, and the scale is constant. Again, Figure C.19 shows the results for bats only; users must inspect models for each size class independently by selecting each size class from the 'Carcass Class' dropdown box.

GenEst produces, for the selected model, the parameters of the persistence distribution (location and scale parameters), the estimated carcass persistence time (medianCP) 44, and the probability of persistence for several typical search intervals (π , r_3 , r_7 , π_4 , and r_28 , with the numbers indicating number of days in the search interval) 45. Ninety percent CIs for all estimates (as selected in General Analysis Inputs 15) are reported in brackets following the point estimate. For final mortality estimation, GenEst computes a probability of persistence for each carcass based on the search interval for that search location. The user selects a final model for each size class from the Model Selection sub-tab 46. As with the searcher efficiency models, GenEst does not automatically select the top-ranked model. The user must select the models manually.

GENEST V1.4.8 Data Input Analyses Help 24 General Inputs Searcher Efficiency Carcass Persistence Mortality Estimation 46 Selected Data Model Comparison Figures Model Inputs: Carcass class: bat Last Time Present: LastPresent Table shows median estimates and 90% confidence intervals for median persistence and r statistics First Time Absent: Show 25 v entries Search: FirstAbsent 45 44 r3 17 r14 r28 Predictor Variables (optional): D 60 7.918 [5.498, 11.28] 0.901 [0.863, 0.93] 0.807 [0.752, 0.853] 0.691 [0.624, 0.751] 0.564 [0.489, 0.636] 0.418 [0.342, 0.493] Season Visibility 2.819 [1.977, 4.009] 0.814 [0.76, 0.862] 0.66 [0.591, 0.726] 0.494 [0.421, 0.568] 0.345 [0.279, 0.417] 0.211 [0.161, 0.271] 60 **Distributions to Include** 60 1.775 [1.23, 2.551] 0.758 [0.694, 0.816] 0.573 [0.499, 0.647] 0.394 [0.324, 0.471] 0.253 [0.195, 0.319] 0.142 [0.105, 0.191] RP exponential 🖾 weibull 🖾 lognormal Ioglogistic Showing 1 to 3 of 3 entries Previous 1 Next Run Model La Download Clear Model Table & Figure Selection: Carcass Class: bat -Distribution: weibull Location: I ~ Visibility Scale: s ~ constant

Figure C.19 Screenshot of Analyses: Carcass Persistence: Estimates tab in the GenEst software

Note: This tab displays the estimates from the selected model, as described in the main text.

In summary, GenEst fits a set of models to the carcass persistence trial data, with the predictor variables of interest selected by the user. All possible models are fit and ranked according to AICc. The user then selects the model to use for mortality estimation based on AICc rank, number of parameters, graphical evaluations of model fits, and reasonableness of the model given the setting and study objectives.

Mortality Estimation

GenEst incorporates the information described above to produce fatality estimates, called Mortality Estimation in GenEst. Clicking the Analyses: Mortality Estimation 24 tab loads a screen with a window for Model Inputs 45 on the left (Figure C.20). The user must specify the column in the carcass observation data set (Figure C.11) that identifies unique carcasses, which is carcID 47 in this example, and specify the Fraction of the Facility Surveyed 48, which is the number of turbines searched divided by the number of turbines at the facility (or sampled length of transmission line divided by total length of transmission line). For this example, a sampling fraction of 1 will be assumed, meaning that all turbines were searched. Last, the user must specify the Date Found 49 column from Figure C.11. The date found should be a column in the carcass observation data that identifies the date on which the recorded carcass was observed. The user should ensure that all dates on which a carcass was found are marked as searched (1) in the search schedule data set, or GenEst will not complete the estimate.

Figure C.20 Screenshot of Analyses: Mortality Estimation tab in the GenEst software



Note: The user must provide the Model Inputs, which are described in the main text.

Once the user inputs are defined, the user can click the Estimate button 50 below the inputs. The Estimate button will not appear if searcher efficiency or carcass persistence models have not been selected. The screen will then display the distribution of total estimated mortality (across all size classes) at the facility during the survey period as a graph (Figure C.21). The high point in the figure is usually near the median estimate (point estimate) 51, and the shaded regions delimit the CIs. The bars show the histogram of the estimate distributions, and the line is a smoothed density curve of that histogram. The median and CI bounds are also displayed on the top right-hand side of the figure.

Figure C.21 Screenshot of Analyses: Mortality Estimation: Figures tab in the GenEst software



Note: This tab displays the distribution of the fatality estimate after the user clicks the Estimate button in the window on the left of the screen.

More information is contained under the Summary sub-tab **52** (Figure C.22). Specifically, GenEst reports quantiles (5%, 25%, 50%, 75%, 95% quantiles) **53** of the fatality estimate distribution and the total number of fatalities (carcasses) observed (the X column) **54**. For example, the 5% column and the 95% column bound the 90% CI, yielding an estimate of 416.61 fatalities per year with 90% CI = 330.02–527.6 (same as reported on the Figures tab, although the figure reports estimates rounded to 1 decimal place). The estimate can be scaled to per-turbine units by dividing the estimate by the number of turbines at the facility or to per MW by dividing by the number of MW at the facility.



neral Inputs Searcher Efficiency	Carcass Persistence	Mortality Estima	tion Detection	Probability		
		52				
Model Inputs:	Figures Summar	y				
Carcass ID Column (CO)						
carcID 🗸	Show 25 v entries			Search:		
Fraction of Facility Surveyed:	x : 54	5%	25%	50%	75%	95%
1	87	330.02	382.53	416.61	456.94	527.6
Date Found:	Showing 1 to 1 of 1 entri	es			Previous	1 Next
Estimate Clear Estimate						
Estimate Clear Estimate Splitting Mortality:						
Estimate Clear Estimate Splitting Mortality: Max. two total splits, max. one schedule-based split						
Estimate Clear Estimate Splitting Mortality: Max. two total splits, max. one schedule-based split Search Schedule (SS) Variable:						
Estimate Clear Estimate Splitting Mortality: Max. two total splits, max. one schedule-based split Search Schedule (SS) Variable:						
Estimate Clear Estimate Splitting Mortality: Max. two total splits, max. one schedule-based split Search Schedule (SS) Variable: Carcass Observation (CO) Variable:						

Note: This tab displays quantiles of the distribution of the estimated fatality rate.

The mortality estimate can be split into groups that the user defines. For example, GenEst can split mortality estimates according to size class in the example, yielding size-specific estimates of mortality (see Splitting Mortality in Figure C.23) **55**. Practitioners will typically be interested in fatality estimates per size class. Differences among the splits will depend on the number of observed carcasses associated with each grouping and differences among the bias trial models for those groups. Estimates can also be split according to search date, species, and visibility, among other possibilities. GenEst does not provide warnings for ill-advised splits. In particular, splits resulting in groups with very few (fewer than five) carcasses may produce misleading estimates. For example, it is possible to produce an estimate for every turbine for every search date, but in many cases, this will result in a very small count of carcasses informing each estimate and unreliable estimates.

Figure C.23 Screenshot of Analyses: Mortality Estimation: Figures tab in the GenEst software after user selects to Split Mortality According to Carcass Size



In summary, GenEst uses the bias trial models and area correction values to appropriately scale observed carcass finds according to search effort (Figure C.9). GenEst returns a distribution for the estimate, yielding a point estimate and Cls. Importantly, GenEst can split the fatality estimate into relevant groups defined by the user. The Cls for fatality estimates might not be reliable if there are fewer than five carcasses and users are urged not to interpret Cls when there are fewer than five carcasses.

Detection Probability

In some cases, the user may want to report the overall Detection Probability **56** for the study, which is the probability of finding a carcass in the entire study area over a specified period of time. Essentially, the overall detection probability is the denominator of Figure 5.2 without the sampling fraction component: p × r × a. To compute overall detection probability, the user must enter search interval, total span of monitoring in days, and carcass size class (Figure C.24). The value inputted for the search interval depends on the needs of the user. Most commonly, the user will enter the average search interval over the monitoring period. The detection probability that GenEst returns can be multiplied by the sampling fraction to obtain a value that is conceptually similar to the detection probability index (DPI) discussed in Sections 3.2.4 and 4.1.3.2. The DPI is an index rather than a value calculated with data. Although conceptually similar, the DPI will not necessarily be quantitatively similar to a GenEst detection probability.

Figure C.24 Screenshot of Analyses: Detection Probability tab in the GenEst software

neral Inputs Searcher Efficiency	Carcas	s Persistence	Mortality E	Stimation	Detection Probability	50		
Model Inputs:	Figu	summary						
Search Interval (days):	Carcas	ss class: bat S	Search Sch	nedule: I = 7, sp	an = 200			
7 Total Span of Monitoring (days):	Show	25 → entries				Search:		
200		Visibility		5%	25%	50%	75%	95%
	1	D		0.18	0.225	0.258	0.296	0.35
Estimate	2	М		0.199	0.242	0.269	0.298	0.344
Clear Estimate	3	RP		0.182	0.218	0.244	0.272	0.311
Table & Figure Selection:	Showi	ng 1 to 3 of 3 entrie	s				Previous	1 Next
Carcass Class:	* D	ownload						
bat 👻								

Note: The tab reports quantiles of the distribution of the estimate of detection probability for the specific carcass class.

APPENDIX D: DENSITY WEIGHTED PROPORTION (DWP) CALCULATION TOOL AND SEARCHER EFFICIENCY TRIAL PLACEMENT TOOL

The DWP Calculation Tool and Searcher Efficiency Trial Placement Tool is provided as a standalone MS Excel spreadsheet: **www.ifc.org/windbirdbatmonitoring**

APPENDIX E: PCFM FATALITY RECORD FORM¹

	Project Nam	e							Search	ner N	lame	e					
	Carcass ID C	ode ²							Turbin	ie or	оні	3					
	Search Year								Search	1 Vis	it Nu	umbe	2 1 74				
	Date of Sear (yyyy-mm-dd)	ch							Time ((hh:mm	of Fir	nd						
	Carcass four	nd wit	hin th	e searcl	ו plot?	Yes			Carca: No	ss fo	ound	duri	ng sch	eduled	l searc	h? Yes	
	Was this fat (e.g., flock of bi	ality p rds collic	art of ding wit	a multi h OHL or	ple fa t multiple	tality e bats f	found bel	ow	/es 🗌 one turbi	Nc ne)	b]					
	Coordinates ⁵ Latitude (dec.deg.)							Longitude (dec.deg.)									
	Nearest turbine / OHL segment ⁶				Dista fatali turbii cente	nce o ty fro ne / C er line	of om OHL e (m)				B fa tu	earir atalit urbin	ig of th y from e ⁷	ie I			
	Carcass Size Class ⁸			bat sm			small bi	small bird medium bird			n bird		large l	oird 🗌			
	Carcass Com	pleter	ness ⁹	complete 🗌			partial			fea	ather	spot [
	Carcass Con	dition	0	intact 🗌			partially scavenged decom			nposed							
	Species							s	pecies	grou	ו ף יי						
	Age	adult [subadult	: 🗌 j	uvenil	e 🗌	unknown									
	Sex	male		female [(unknov	wn 🗌										
	Eyes	round [fluid-fille	ed 🗌 o	dehydr	ated 🗌		sunk	en 🗌		absen	it 🗌	not ap	plicable		
	Cause of Dea	ath ¹²	unknov	wn 🗌 d	ollision		electroc	utic	on 🗌	otl (pr	her [roject	relate	ed)	other (non-	project i	elated)	
	Days since De	eath ¹³	unknov	wn 🗌 d	D-1		2-3		4-7			8–14 [15-30		>30]
	Visibility cla	ss whe	re car	cass fou	nd ¹⁴	easy [moo	derate 🗌] dif	fficult		very d	ifficult [
P	hotograph D	etails															
Ρ	hoto number	-	Date/ (yyyy-m	Time Im-dd. hh	:mm)	Dese	cription	1						Filenar	ne		

Additional Notes

This data collection form is a fieldwork resource (Appendix E) provided in IFC 2023. Post-construction Bird and Bat Fatality Monitoring for Onshore Wind Energy Facilities in Emerging Market Countries. Good Practice Handbook and Decision Support Tool.

Instructions for filling in the Fatality Record Form (Numbers on this page relate to numbers on the form)

- ¹ Each fatality found requires a separate PCFM Fatality Record Form.
- ² Carcass ID Code should be a unique identifier for each carcass across the whole monitoring program. The Lead Ecologist (LE) should determine the code format. The code should provide as much information about the carcass as possible without being too long. For example, "20220128_T23_HOBA_1" = date found (2022-01-28)_turbine number (T23)_species name (Hoary Bat or HOBA for short)_order found at that turbine during the search (1); if only one carcass was found, this last number will always be"1".
- ³ State whether the search is for a PCFM program at turbines or for an overhead power line (OHL)?
- ⁴ Give the search visit number. A search visit is a single search of all sample turbines or OHL segments. For example, if the study design is to search 20 turbines every seven days, the first search of all 20 turbines in the first seven days will be search visit 1, the second search of the same 20 turbines in the next seven days will be search visit 2. Similarly, at a WEF-associated OHL, if the study design is to search sample search plots of 10 x 5-km segments per month, the first search of plots in these 10 segments during the first month will be search visit 1, the second search of the same plots in these 10 segments in the second month will be search visit 2.
- ⁵ The fatality location coordinates (latitude and longitude [in decimal degrees]) should be determined using a Global Positioning System (GPS) device and recorded on the form.
- ⁶ Give the turbine number/ OHL segment nearest to the fatality. Each turbine should have a unique number. The LE should provide searchers with turbine numbers using those given in the project layout. OHLs should be divided into segments of equal length with each segment assigned a unique number. Segment numbers should be provided to searchers by the LE.
- ⁷ Use a compass or GPS device to obtain a descriptive bearing (north, north-east, east, south-east, south, south-west, west, north-west). This should be the bearing from the base of the turbine to the carcass not the bearing from the carcass back to the turbine. For fatalities found along OHLs give a descriptive bearing to indicate which side of the OHL the fatality was located.
- ⁸ Tick the relevant box to indicate the size class of this fatality. Size guide: small birds (<30 cm body length), medium birds (30-55 cm body length). Large fruit bats should be entered in the same size class as medium birds.
- ⁹ Tick the relevant box next to the term that best describes the completeness of the fatality. Complete = all parts of the carcass are present. Partial = only part of the carcass remains. Feather spot = feathers are the only evidence of a carcass. The minimum criteria for a feather spot is more than two connected primary wing feathers or more than 10 body feathers in proximity (within a 5 m x 5 m area [25 m²]). Details of feathers found that do not meet this criterion can be noted in the Additional Notes section.
- Tick the relevant box to indicate the term that best describes the overall condition of the fatality. Intact = no evidence of scavenging or decomposition. Partly scavenged = carcass has clear signs of predation but some of the carcass remains (e.g., wings) but it is likely that the carcass can be identified to a species. Decomposed = carcass has degraded beyond the point where it can be identified to a species. Note that a carcass categorized as decomposed may also have been subject to partial scavenging.
- ¹¹ If the species cannot be determined, identifying the fatality to a species group (e.g., raptor or wader) may be relevant to a WEFs monitoring objectives and can be used as a covariate in the analysis. State as unknown if the species group cannot be determined.
- ¹² Cause of death can be difficult to determine, but if there are clues to what caused the animal to die, they should be noted. **It is better to score the cause of death as unknown than to guess.**
- ¹³ Tick the relevant box to give an estimate of time of death of the fatality. Estimating time of death is difficult. Scavenging (animal or insect), environmental conditions (e.g., temperature, humidity, precipitation), size and body composition of the specimen, and other factors affect carcass condition and the speed with which carcasses decompose. The LE will review evidence (carcass condition, photographs, completed PCFM Fatality Record forms) from fatalities found during the study to improve time of death estimates. Carcass persistence trial decomposition rates should also be used to increase the accuracy of time of death estimates.
- ¹⁴ Tick the relevant box to indicate the ground surface visibility class where the fatality was found. The LE should ensure that definitions for each visibility class used on the site are provided to searchers.
- ¹⁵ At least five photographs should be taken of the fatality. Carcasses (including feather spots) should be photographed: i) as found, taken from directly above, ii) turned over to show the ground-facing side, iii) from different angles to show any features that may help with identification (e.g., head, wings, tail, feet), iv) showing any injuries, v) showing the area immediately surrounding the fatality (e.g., relative to the nearest turbine / OHL). All photographs should have a size reference (e.g., ruler) and the carcass ID number either written on a white board or sheet of paper. When taking digital photographs, the photo number, date, and time can be made visible in the camera viewfinder. Record this information to assist in associating the carcass with the photograph. Photographs should be saved and stored using the Carcass ID Code (see 2 above) as the file name. Using the example in 2 above, photographs would have file names 20220128_T23_HOBA_1.1, 20220128_T23_HOBA_1.2.

This data collection form is a fieldwork resource (Appendix E) provided in IFC 2023. Post-construction Bird and Bat Fatality Monitoring for Onshore Wind Energy Facilities in Emerging Market Countries. Good Practice Handbook and Decision Support Tool.

ШШ
щ
>
Ē
Ĕ
E
5
S
_
2
đ
e,
ш.
×
5
Ξ.
.
a
2
•

APPENDIX F: SEARCH SUMMARY FORM¹

		Additional notes (e.g., reasons for no search, weather)										
me	Number ³	Carcass ID code(s) ⁷										
Searcher Na	Search Visit	Number of searcher efficiency trials found (enter 0 if none found)										
		Number of fatalities found (enter 0 if nonefound)										
		Not searched ⁶										
	OHL	Searched										
	Turbine	cheduled earch date ⁵ yyy:mm:dd)										
Project Name	Search visit type²	Turbine number / So OHL segment ⁴ So OHL segment ⁴										

This data collection form is a fieldwork resource (Appendix F) provided in IFC 2023. Post-construction Bird and Bat Fatality Monitoring for Onshore Wind Energy Facilities in Emerging Market Countries. Good Practice Handbook and Decision Support Tool.

(Instructions for filling in the Search Summary Form (Numbers on this page relate to numbers on the form)

- The Lead Ecologist (LE) should provide each searcher with a Search Summary form at the beginning of each search visit. A separate Search Summary form should be used for turbine search visits and for overhead power line (OHL) segment search visits. See 3 below for search visit definition
- Tick the box to indicate whether the Search Summary form relates to a turbine search visit or an OHL search visit.
- OHL, if the study design is to search sample search plots of 10 x 5-km segments per month, the first search of plots in these 10 segments during the first month will be search visit 1, the search of all 20 turbines in the first seven days will be search visit 1, the second search of the same 20 turbines in the next seven days will be search visit 2. Similarly, at a WEF-associated Give the search visit number. A search visit is a single search of all sample turbines or OHL segments. For example, if the study design is to search zo turbines every seven days, the first second search of the same plots in these to segments in the second month will be search visit 2.
- Give the turbine number/ OHL segment searched. Each turbine should have a unique number. The LE should provide searchers with turbine numbers using those given in the project layout. OHLs should be divided into segments of equal length with each segment assigned a unique number. Segment numbers should be provided to searchers by the LE.
- ⁵ Give the scheduled search date for the turbine/OHL segment.
- Tick the "Not Searched" box if it was not possible to search the plot on the scheduled searched date. Give reasons in the "Additional notes" column. ی
- 7 Give all Carcass ID codes of fatalities found.

This data collection form is a fieldwork resource (Appendix F) provided in IFC 2023. Post-construction Bird and Bat Fatality Monitoring for Onshore Wind Energy

Facilities in Emerging Market Countries. Good Practice Handbook and Decision Support Tool

APPENDIX G: SEARCHER EFFICIENCY (SE) TRIAL FORM¹

Project N	lame						•	Trial A	dminis	trator I	Name ²			
SE Trial C	arcass ID	Code ³						Carca	ss ID C	:ode⁴				
Season⁵								PCFM	Type ⁵	Turbir	ie 🗌		OHL [
Date of S (yyyy-mm-	Gearch dd)							Time (hh:mm	of Find າ)	l				
Start Tria Period (yyyy-mm-	al- dd)		End Ti Period (yyyy-m	r ial- 1 nm-dd)				Date Placeo (yyyy-m	1 ım-dd)			Time Place (hh:mr	n)	
Nearest OHL seg	turbine / ment ⁷			Dista fatali turbir cente	nce o ty fro ne / C er line	of om OHL e (m)				Bearin the tr turbin	ng of ial froi ne®	m		
Coordina	ites⁵	Latitu	de (dec.o	deg.)					Longit	:ude (de	c.deg.)			
Trial Size	e Class ¹⁰		bat 🗌			small bird]	med	ium bird		large	bird 🗌	
Trial Typ	e"		WEF Fat	ality 🗌		surrogate]	decc	у 🗌				
Trial Det	ails12	Species	/group ¹¹						Deco	oy descri	otion			
Visibility	class at p	laceme	ent site	з (easy [m	ode	erate	diffic	cult 🗌	very	difficult		
Trial Mar	'king¹⁴													
Trial Resul	ts ¹⁵													
Attempt	Search D	ate	Foi	und ?						Avail	able?			
1			Yes	5	No [Yes	N	o 🗌		
2			Yes	5	No					Yes	N	o 🗌		
3			Yes	<u> </u>	No					Yes	N	o 🗌		
4			Yes	5	No					Yes	N	o 🗌		
SE trial u persister	ised for ca nce (CP) ti	rcass rial?16		Yes		Νο		CP Ca	rcass	Trial ID	Code	7		
Photograp	h Details ¹⁸	3												
Photo num	nber	Date/1 (yyyy-mi	Time m-dd. hh	:mm)	Desc	cription						Filena	me	
Additional	Notes													

This data collection form is a fieldwork resource (Appendix G) provided in IFC 2023. Post-construction Bird and Bat Fatality Monitoring for Onshore Wind Energy Facilities in Emerging Market Countries. Good Practice Handbook and Decision Support Tool. Cross-references on the instruction sheet are related to sections and appendices in this document.

Instructions for filling in Searcher Efficiency (SE) Trial Form (Numbers on this page relate to numbers on the form)

- ¹ Each trial carcass, surrogate, or decoy requires a separate Searcher Efficiency Trial Form.
- ² The trial administrator, usually the Lead Ecologist (LE) places searcher efficiency trial carcasses in the field before searchers arrive for a search and confirms trial carcass availability after the search if surveyors do not find them. **The trial administrator must not be a searcher because searchers must not be aware of the timing and location of searcher efficiency trials.**
- ³ Each SE trial carcass/decoy should be assigned its own unique identification (ID) code. For example, "SE_20220215_BAT_1" = Searcher Efficiency Trial (SE)_Date Placed (2022-02-15)_Size Class (BAT)_Size Class Trial Carcass Number (1).
- ⁴ If the carcass was originally a fatality found at the site, give the Carcass ID Code previously assigned on the Fatality Recording Form (e.g., 20220128_T23_HOBA_01), otherwise leave this box blank.
- ⁵ "Season" is defined based on climate, (e.g., summer, autumn, winter, spring; rainy (monsoon), dry) or according to ecological or biological factors (e.g., migration periods, breeding, or non-breeding season). Before trials, these seasons must be defined with start and end dates.
- ⁶ Tick the relevant box to indicate whether the trial is for a PCFM program at turbines or for an overhead power line (OHL).
- ⁷ Give the turbine number/ OHL segment nearest to the trial. Each turbine will have a unique number. The LE should provide the trial administrator with turbine numbers using those given in the project layout. OHLs should be divided into segments of equal length with each segment assigned a unique number. Segment numbers should be provided to the trial administrator by the LE.
- ⁸ Use a compass or Global Positioning System (GPS) device to obtain a descriptive bearing i.e. (north, north-east, east, south-east, south, south, south-west, west, north-west). This should be the bearing from the base of the turbine to the carcass not the bearing from the carcass back to the turbine. For OHL SE trials, give a descriptive bearing to indicate which side of the OHL the trial was placed.
- 9 Trial locations must be determined using the Searcher Efficiency Trial Placement Tool (see Appendix D). The trial administrator will navigate to each pre-determined location and place the trial. Once the trial is placed, the actual location coordinates (latitude and longitude [in decimal degrees]) should be determined using a GPS device and recorded on this form.
- ¹⁰ Tick the relevant box to indicate the size class of this trial. Size guide: small birds (<30 cm body length), medium birds (30-55 cm body length). If large fruit bats are used, these should be entered in the same size class as medium birds.</p>
- ¹¹ Tick the relevant box to indicate the type of trial carcass used. Actual collision fatalities should be used in searcher efficiency trials when available and feasible. (See Box 3.5 for details on suitable surrogates and decoys).
- ¹² State the species or if not known a more general grouping, e.g., vulture, raptor. For decoys briefly describe e.g., rubber bat, duck decoy.
- ¹³ Tick the relevant box to indicate the ground surface visibility class where the trial was placed. This should be consistent with the defined visibility classes that the Lead Ecologist has mapped.
- ¹⁴ Briefly describe the trial marking. Trial carcasses should be discreetly marked so that they can be distinguished from actual fatalities. Markings should be discrete but unmistakable, such as with a small length of electrical tape around a foot or clipped flight feathers (see Appendix B).
- ¹⁵ Fill in details of each search attempt in the table. If the SE trial design only allows one attempt to find the trial leave the other rows blank. After the first search, the trial administrator must check whether the carcass, surrogate, or decoy is still present and therefore was 'available' to be found by the searcher. If the carcass, surrogate, or decoy is no longer visible at the placement location, the trial will not be included in the searcher efficiency trial. If the searcher efficiency bias trial design allows multiple opportunities to search for a carcass, surrogate, or decoy, the trial administrator must check that they are still'available' to be found after each search event. Decomposed carcasses should not be used for subsequent search attempts.
- ¹⁶ Intact carcasses used for SE trials may be left to test carcass persistence. State if the SE trial was subsequently used for this purpose.
- ¹⁷ If the SE trial is used as a carcass persistence trial, it should have a dedicated Carcass Persistence Form with a CP Carcass ID code assigned. Provide the CP Carcass ID code on this form as a cross reference.
- ¹⁸ For record keeping, at least two photographs of the trial carcass/decoy at the time of placement showing its location should be taken. When taking digital photographs, the photo number, date and time can be made visible in the camera viewfinder. Record this information to assist in associating the trial with the photograph. Photographs should be saved and stored using
- ¹⁹ the SE Trial ID Code as the file name (see 4 above). Using the example in 4 above, two photographs would have file names. SE_20220215_BAT_1.1, SE_20220215_BAT_1.2.

This data collection form is a fieldwork resource (Appendix G) provided in IFC 2023. Post-construction Bird and Bat Fatality Monitoring for Onshore Wind Energy Facilities in Emerging Market Countries. Good Practice Handbook and Decision Support Tool. Cross-references on the instruction sheet above are related to sections and appendices in this document.

APPENDIX H: CARCASS PERSISTENCE (CP) TRIAL FORM¹

	Project Name	1							Trial A	dminis	trator N	lame ²		
	CP Trial Carca	ss ID	Code ³						Carca	ss ID C	ode⁴			
	Season⁵								PCFM	Type⁵	Turbin	e 🗌		OHL
	Start Trial- Period (yyyy-mm-dd)			End Tr Perioc (yyyy-m	rial- 1 1m-dd)				Date Placed (hh:mm	1 1)			Time Place (hh:mn	n)
	Nearest turbi OHL segment	ne / 7			Dista from OHL (m)	ance o turbi cente	f trial ne / r line				Bearin the tri turbin	ig of al fror e ⁸	n	
	Coordinates ⁵		Latitı	Ide (dec.c	leg.)					Longit	u de (dec	.deg.)		
	Trial Size Clas	S ¹⁰		bat 🗌			small b	oird [med	ium bird		large l	bird
	Trial Type"			WEF Fata	ality 🗌]	surrog	ate [
	Trial Details ¹²		Species	s/group"										
	Visibility class	s at pl	acem	ent site ¹	3	easy 🗌		mod	lerate] diffic	ult	veryo	difficult [
	Trial Marking ¹	4												
	Condition of carcass at time of placement ¹⁵				fresh thav			/ed						
	Check Day¹⁵	Cheo Dato (mm:	ck e ¹⁷ :dd) (Check Fime hh:mm)	Caro Com	ass plete	ness ¹⁸	Car Cor	cass Idition ¹⁹	Pho Nun	to nbers²º	Cheo by	cked	Notes
-														
-														

Additional Notes

This data collection form is a fieldwork resource (Appendix H) provided in IFC 2023. Post-construction Bird and Bat Fatality Monitoring for Onshore Wind Energy Facilities in Emerging Market Countries. Good Practice Handbook and Decision Support Tool. Cross-references on the instruction sheet are related to sections and appendices in this document..

Instructions for filling in Carcass Persistence (CP) Trial Form (Numbers on this page relate to numbers on the form)

- ¹ Each trial carcass and surrogate requires a separate Carcass Persistence Trial Form
- ² The trial administrator, usually the Lead Ecologist (LE) is responsible for placing trials and ensuring that they are checked at the prescribed intervals (see 16 below).
- ³ Each CP trial carcass should have its own unique identification (ID) code. For example, "CP_20220215_BAT_1", = Carcass Persistence Trial (SE)_Date Placed (2022-02-15)_Size Class (BAT)_Size Class Trial Carcass Number (1)
- ⁴ If the carcass was originally a fatality found at the site, give the Carcass ID Code previously assigned on the Fatality Recording Form (e.g., 20220128_T23_HOBA_01), otherwise leave this box blank.
- ⁵ "Season" is defined based on climate, (e.g., summer, autumn, winter, spring; rainy (monsoon), dry) or according to ecological or biological factors (e.g., migration periods, breeding, or non-breeding season). Before trials, these seasons must be defined with start and end dates.
- ⁶ Tick the relevant box to indicate whether the trial is for a PCFM program at turbines or for a PCFM at an overhead power line (OHL).
- ⁷ Give the turbine number/ OHL segment nearest to the trial. Each turbine will have a unique number. The LE should provide the trial administrator with turbine numbers using those given in the project layout. OHLs should be divided into segments of equal length with each segment assigned a unique number. Segment numbers should be provided to the trial administrator by the LE.
- ⁸ Use a compass or Global Positioning System (GPS) device to obtain a descriptive bearing i.e. (north, north-east, east, south-east, south, south, south-west, west, north-west. This should be the bearing from the base of the turbine to the carcass not the bearing from the carcass back to the turbine. For OHL CP trials, give a descriptive bearing to indicate which side of the OHL the trial was placed.
- ⁹ Record trial location coordinates (latitude and longitude [in decimal degrees]) from a GPS device.
- ¹⁰ Tick the relevant box to indicate the size class that is relevant to this trial. Size guide: small birds (<30 cm body length), medium birds (30-55 cm body length), large birds (>55 cm body length).
- ¹¹ Tick the relevant box to indicate the type of trial carcass used. Actual collision fatalities should be used in carcass persistence trials when available, fresh and feasible. (See Box 3.5 for details on suitable surrogates).
- ¹² State the species or if not known a more general relevant grouping, e.g., vulture, raptor.
- ¹³ Tick the relevant box to indicate the ground surface visibility class where the trial was placed. This should be consistent with the defined visibility classes that the LE has mapped.
- ¹⁴ Trial carcasses should be discreetly marked so that they can be distinguished from actual fatalities (see Appendix B).
- ¹⁵ Tick the relevant box to indicate the condition of the carcass at time of placement. Carcass persistence trials should use fresh carcasses or thawed carcasses that were frozen when fresh and stored frozen until use.
- ¹⁶ Enter the scheduled check day. At a minimum CP trial should be checked 1, 2, 3, 4, 7, 14, 21, and 30 days after they are placed (see details in Appendix B).
- ¹⁷ The "Check Date" column should always give the date that the carcass was actually checked. For example, if it was not possible to check a carcass on day 30 but it was checked a day later, the date given in the "Check Date" column should be the date on Day 31, and not the scheduled date i.e., Day 30.
- ¹⁸ Assign one of the following carcass completeness categories. Complete = all parts of the carcass are present. Partial = only part of the carcass remains. Feather spot = feathers are the only evidence of a carcass. The minimum criteria to be defined as a feather spot are more than two connected primary wing feathers or more than 10 body feathers found in proximity (within a 5 m x 5 m area [25 m²]).
- ¹⁹ Assign one of the following carcass condition categories. Intact = no evidence of scavenging or decomposition. Partly scavenged = carcass has clear signs of predation but some of the carcass remains (e.g., wings). Decomposed = carcass has degraded beyond the point where it can be identified to a species. Note that a carcass categorized as decomposed may also have been subject to partial scavenging.
- ²⁰ For record keeping, at least two photographs of the carcass and its location should be taken. When taking digital photographs, the photo number can be made visible in the camera viewfinder. Record this information to assist in associating the trial with the photograph. Photographs should be saved or stored using the Carcass Trial ID Code (see 4 above) as the file name. Using the example in 4, above, photographs would have file names CP_20220128_BAT_1.1, CP_20220128_BAT_1.2.

This data collection form is a fieldwork resource (Appendix H) provided in IFC 2023. Post-construction Bird and Bat Fatality Monitoring for Onshore Wind Energy Facilities in Emerging Market Countries. Good Practice Handbook and Decision Support Tool. Cross-references on the instruction sheet above are related to sections and appendices in this document.

APPENDIX I: CARCASS FREEZER TAGS¹

Project Name	Carcass ID Code ²		Searcher ³	
Carcass Source	systematic search	chance find	trial surrogate	
Carcass Size Class⁴	bat small bird	medium bird	large bird	
Date Stored	Stored by⁵		Turbine/OHL Segm	ent ⁶

Project Name	Carcass ID Code ²	Searcher ³
Carcass Source	systematic search chance find	trial surrogate
Carcass Size Class⁴	batsmall birdmedium bird	large bird
Date Stored	Stored by ⁵	Turbine/OHL Segment ⁶

Project Name		Carcass ID Code ²		Searcher ³	
Carcass Source	systema	atic search 🗌	chance find	trial surrogate	
Carcass Size Clas	s ⁴ bat	small bird	medium bird	large bird	
Date Stored		Stored by₅		Turbine/OHL Seg	jment ⁶

Project Name	Carcas Code²	s ID	Searcher ³	
Carcass Source	systematic search	chance find	trial surrogate	
Carcass Size Class⁴	bat small bird	d 🔄 medium bird 🗌	large bird	
Date Stored	Stored	by₅	Turbine/OHL Segm	ent ⁶

Instructions for filling in Carcass Freezer Tags (Numbers on this page relate to numbers on the form)

- ¹ Each carcass requires its own tag. This form contains five tags that can be cut along the dotted lines. Before storing carcasses in the freezer, double bag each one, with the carcass tag placed in between the two bags. It is preferable that the outer bag is clear and the tag be placed so that it is readable without having to open any bags
- ² If the carcass was a fatality found during PCFM searches at the project, provide the Carcass ID Code assigned to this fatality. If the carcass is a trial surrogate leave blank.
- ³ If the carcass was a fatality found during PCFM searches at the project, give the name of the searcher who recorded the fatality. Otherwise leave blank.
- ⁴ Tick the relevant box to indicate the size class of the carcass. Size guide: small birds (<30 cm body length), medium birds (30-55 cm body length), large birds (>55 cm body length). If large fruit bat carcasses are found, they should be entered in the same size class as medium birds.
- ⁵ Give the name of the person who stored the carcass in the freezer.
- ⁶ For fatalities found during systematic searches or as chance finds within search plots give the turbine number or OHL segment where the fatality was found.

This data collection form is a fieldwork resource (Appendix I) provided in IFC 2023. Post-construction Bird and Bat Fatality Monitoring for Onshore Wind Energy Facilities in Emerging Market Countries. Good Practice Handbook and Decision Support Tool.

()
ŏ
S L
Ż
ш
\leq
$\overline{\bigcirc}$
\bigcup
പ
Ш
<u>···</u>
\geq
\square
\geq
Ш

Project Name								
Date Stored	Carcass ID Code ¹	Project Fatality	Bias Trial Surrogate	Surrogate Source ²	Carcass Size Class³	Species/ Group ⁴	Carcass Completeness ⁵	Carcass Condition ⁶
setructions for fill	ina in Ereezer Contents	t od (Nimbors	or open sidt no	rotania ot otel	e on the form)			

s on the jorm) טוו נוווא אמשב ובומנב CONTENTS LOG (NUMPERS Instructions for filling in Freezer

For fatalities found during PCFM searches at the project give the Carcass ID code from the Fatality Record Form (Appendix E). For trial surrogates leave blank.

² For trial surrogates state the source of the carcass e.g. road kill. For other carcasses leave blank.

Give the size class of the carcass. Size guide: small birds (<30 cm body length), medium birds (30-55 cm body length), large birds (>55 cm body length). Large fruit bats should be entered in the same size class as medium birds.

⁴ State the species or if not known a more general relevant grouping, e.g., vulture, raptor.

For project fatalities assign the carcass completeness category given on the Fatality Record form (Appendix E). For surrogates leave blank.

For project fatalities assign the carcass condition category given on the Fatality Record form (Appendix E). For surrogates leave blank.

This data collection form is a fieldwork resource (Appendix I) provided in IFC 2023. Post-construction Bird and Bat Fatality Monitoring for Onshore Wind Energy Facilities in Emerging Market Countries. Good Practice Handbook and Decision Support Tool. Cross-references on the instruction sheet above are related to sections and appendices in this document.

APPENDIX K: ESSENTIAL FIELDWORK EQUIPMENT

Field binder file containing:
Health and safety plan and risk assessment and protocol, including details on an emergency
response plan; next-of-kin contact for each team member; and name, address, and phone number of nearest hospital
Injured animal handling protocol, including name, address, and phone number of nearest wildlife rehabilitation center or veterinary clinic (if such a facility exists nearby)
Relevant training certificates or copies of the same
Multiple copies of data collection forms, including:
Fatality Recording Form (Appendix E)
Search Summary Form (Appendix F)
Searcher Efficiency Trial Form (Appendix G)
Carcass Persistence Trial Form (Appendix H)
Carcass Freezer Tags (Appendix I)
Freezer Contents Log (Appendix J)
Guide for codes used in data entry
Species list (highlighting PCFM target species) with descriptions and photographs or illustrations
Copies of any required permits (e.g., collection and handling of dead specimens, transport of carcasses)
Personal Protective Equipment (PPE) prescribed by wind energy facility or Lead Ecologist
First aid kit
Vehicle emergency kit
Communication device (cell phone, two-way radio or satellite phone)
Drinking water (at least 5 liters)
Food
Sunscreen
Camera
Whiteboard and wipeable marker to display Carcass ID Code in fatality photos
Global Positioning System (GPS) device
Compass
Ruler
Clipboard (for use with data collection forms)
Pens or pencils for completing data collection forms
Permanent or freezer marker pens
Gloves for handling carcasses (leather and nitrile)
Scissors or nail clippers and tape for marking trial carcasses
Sealable clear plastic bags of various sizes
Marking tape or labels for freezer bags
Other Fieldwork Equipment that may be needed
Electronic tablet for recording data electronically
Rangefinder
Calipers if more accurate forearm length measurements of bats are required
Containers for fatalities collected as voucher specimens and for DNA samples (e.g., Ziploc bags, vials, ethanol)
Motion sensor camera for carcass persistence monitoring

Insect repellent

This checklist is a fieldwork resource (Appendix K) provided in IFC 2023. Post-construction Bird and Bat Fatality Monitoring for Onshore Wind Energy Facilities in Emerging Market Countries. Good Practice Handbook and Decision Support Tool.

© 2023 International Finance Corporation. All rights reserved.

2121 Pennsylvania Avenue, NW Washington, DC 20433 USA Internet: www.ifc.org

